

Single-Phase Grid-Connected Battery-Supercapacitor Hybrid Energy Storage System

Author:

Wickramasinghe Abeywardana, Damith

Publication Date: 2016

DOI: https://doi.org/10.26190/unsworks/19297

License:

https://creativecommons.org/licenses/by-nc-nd/3.0/au/ Link to license to see what you are allowed to do with this resource.

Downloaded from http://hdl.handle.net/1959.4/57053 in https:// unsworks.unsw.edu.au on 2024-05-05

Single-Phase Grid-Connected Battery-Supercapacitor Hybrid Energy Storage System

by Damith B. Wickramasinghe Abeywardana



A thesis submitted in fulfilment of the requirements for the degree of

Doctor of Philosophy

School of Electrical Engineering and Telecommunications

Faculty of Engineering

The University of New South Wales

© Damith B. Wickramasinghe Abeywardana December 2016

Sydney, NSW, Australia

ORIGINALITY STATEMENT

'I hereby declare that this submission is my own work and to the best of my knowledge it contains no materials previously published or written by another person, or substantial proportions of material which have been accepted for the award of any other degree or diploma at UNSW or any other educational institution, except where due acknowledgement is made in the thesis. Any contribution made to the research by others, with whom I have worked at UNSW or elsewhere, is explicitly acknowledged in the thesis. I also declare that the intellectual content of this thesis is the product of my own work, except to the extent that assistance from others in the project's design and conception or in style, presentation and linguistic expression is acknowledged.'

COPYRIGHT STATEMENT

'I hereby grant the University of New South Wales or its agents the right to archive and to make available my thesis or dissertation in whole or part in the University libraries in all forms of media, now or here after known, subject to the provisions of the Copyright Act 1968. I retain all proprietary rights, such as patent rights. I also retain the right to use in future works (such as articles or books) all or part of this thesis or dissertation.

I also authorise University Microfilms to use the 350 word abstract of my thesis in Dissertation Abstract International (this is applicable to doctoral theses only).

I have either used no substantial portions of copyright material in my thesis or I have obtained permission to use copyright material; where permission has not been granted I have applied/will apply for a partial restriction of the digital copy of my thesis or dissertation.'

Signed

Date

AUTHENTICITY STATEMENT

'I certify that the Library deposit digital copy is a direct equivalent of the final officially approved version of my thesis. No emendation of content has occurred and if there are any minor variations in formatting, they are the result of the conversion to digital format.'

Date

THE UNIVERSITY OF NEW SOUTH WALES Thesis/Dissertation Sheet

Surname or Family name: Wickramasinghe Abeywardana

First name: Damith

Abbreviation for degree as given in the University calendar: PhD

School: Electrical Engineering and Telecommunications

Title: Single-Phase Grid-Connected Battery-Supercapacitor Hybrid Energy Storage System

Abstract 350 words maximum:

Battery technology is popular in distributed energy storage systems (ESSs) due to its ease of implementation. However, batteries have limited power capabilities, and the lifetime of batteries deteriorates due to high and fluctuating battery currents.

Battery-supercapacitor hybrid energy storage systems (HESSs) become a promising way of increasing the battery lifetime and system power capability. The HESSs can be divided into two groups, DC link based and direct AC line integrated HESSs. The direct AC line integrated HESSs have been proposed to eliminate multiple power processing stages associated with the DC link based HESSs.

The objective of this research is to develop a direct grid-connected battery-supercapacitor HESS able to allocate fast power fluctuations to the supercapacitor while maintaining the state of charge of the ESSs within a safe operating region. Since the lifetime of ESSs deteriorates with ripple current components, ways of reducing the current ripple components of the ESSs are studied.

In this thesis, a boost inverter based battery-supercapacitor HESS is proposed. A supercapacitor voltage controller and a filter based method is used to allocate the fast power fluctuations to the supercapacitor. Then, a supercapacitor energy controller (SCEC) based power allocation method facilitating the HESS dynamic analysis and precise supercapacitor sizing is proposed. Later, a sliding mode controlled HESS with a SCEC based power allocation method is proposed to achieve better output voltage reference following performance.

To mitigate the second-order harmonic current components in the boost inverter based HESSs, a rule-based control method and a novel current feedback method are proposed. Both methods achieve a significant ripple current reduction without being affected by the output capacitor tolerances while the latter method mitigates the current ripple even during the output power transients.

The proposed HESS is the first experimentally verified single-phase direct grid-connected HESS able to reduce the switching frequency and the second-order harmonic current ripples.

Declaration relating to disposition of project thesis/dissertation

I hereby grant to the University of New South Wales or its agents the right to archive and to make available my thesis or dissertation in whole or in part in the University libraries in all forms of media, now or here after known, subject to the provisions of the Copyright Act 1968. I retain all property rights, such as patent rights. I also retain the right to use in future works (such as articles or books) all or part of this thesis or dissertation.

I also authorise University Microfilms to use the 350 word abstract of my thesis in Dissertation Abstracts International (this is applicable to doctoral theses only).

The University recognises that there may be exceptional circumstances requiring restrictions on copying or conditions on use. Requests for restriction for a period of up to 2 years must be made in writing. Requests for a longer period of restriction may be considered in exceptional circumstances and require the approval of the Dean of Graduate Research.

FOR OFFICE USE ONLY

Date of completion of requirements for Award:

Other name/s: Buddika

Faculty: Faculty of Engineering

Date

Abstract

Battery based ESSs are popular among distributed network ESSs due to ease of implementation and geographical independence as compared to other energy storage technologies. However, due to its chemical properties batteries have limited power capabilities compared to other energy storage technologies such as supercapacitors. Hence, a large number of batteries may need to be connected in parallel in order to meet the required power capabilities. Furthermore, the lifetime of the battery deteriorates due to high battery currents and battery current fluctuations as explained in Chapter 1.

Battery-supercapacitor hybrid energy storage systems (HESS) become a promising way of battery lifetime extension since they can effectively minimize the battery current variations. HESSs combine individual advantages of both the battery and the supercapacitor and create a single ESS with both higher power and energy capabilities. The HESSs can be divided into two basic groups, namely, DC link based and direct AC line integrated HESSs. The direct AC line integrated ESSs have been proposed in literature as a way of eliminating the multiple power processing stages associated with the DC link based ESSs.

The objective of this research is to develop a single phase direct grid integrated battery-supercapacitor HESS which is able to allocate the fast power fluctuations to the supercapacitor. Furthermore, the battery and the supercapacitor state of charge (SOC) have to be maintained within a safe operating region. Since the lifetime of the DC ESSs deteriorates due to the ripple current components as explained in Chapter 1, it is vital to identify ways of reducing the switching frequency ripple components and the second-order harmonic components of both the battery and the supercapacitor currents.

A boost inverter based battery-supercapacitor HESS is proposed in this thesis. The boost inverter provides both boosting and inversion functions in a single power processing stage allowing the direct grid integration of low voltage DC sources. A supercapacitor voltage controller (SCVC) and a filter based power allocation method is used to allocate the fast power fluctuations and the second-order harmonic ripple component to the supercapacitor. An interleaved operation for the boost inverter is proposed to reduce the switching frequency ripple components of the DC source currents.

Then, a supercapacitor energy controller (SCEC) and a filter based power allocation method is proposed to circumvent the issues associated with the SCVC and filter based power allocation method. The proposed method enables the dynamic analysis of the HESS, precise sizing of the DC sources and selection of the power allocation parameters.

A novel fixed frequency sliding mode (SM) controller is proposed for the boost inverter based HESS. The SM controller is able to achieve better output capacitor voltage reference tracking and reduce the implementation complexity of the HESS controller. A SCEC based power allocation method, a power allocation parameter selection method, and an energy storage element sizing method are proposed for the SM controlled HESS.

Then, ways of mitigating the second-order harmonic ripple component in the HESS are studied. With the conventional HESS operation, the ripple current component is allocated to the supercapacitor which deteriorates its lifetime. A rule-based control method is proposed to overcome the limitations associated with the existing waveform control ripple reduction method. The proposed method is able to achieve a significant ripple current reduction without being affected by the output capacitor tolerances and inductor ESR values. However, the method requires a specific time period to reduce the ripple component once a change in the active or reactive power reference occurs. Then, a novel current feedback ripple reduction method is proposed, and the method achieves superior performance compared to the existing ripple reduction methods. The current feedback method is able to reduce the second-order harmonic ripple current even during the output power transient conditions and hence it is suitable for the HESS operation.

All the proposed systems are verified experimentally to illustrate the applicability of the proposed HESS in grid ESS applications. The proposed HESS is the first experimentally verified single phase direct grid connected HESS which is able to reduce both the switching frequency ripple components and also the second-order harmonic ripple components in both the battery and the supercapacitor currents. Furthermore, the proposed SCEC based power allocation strategy enables the accurate HESS power allocation parameter selection and precise energy storage element sizing.

Acknowledgements

I would like to express my sincere appreciation and gratitude to my supervisor Dr. Branislav Hredzak, for his guidance, support, and encouragement throughout my Ph.D. research. His valuable insights and inputs enabled me to achieve success in my Ph.D. research. I am thankful for his guidance and encouragement which influenced my research activities as well as my future career.

I would also like to express my deepest gratitude to my co-supervisor, Professor Vassilios Agelidis for his encouragement and valuable inputs from the beginning to the final stage of my Ph.D. research.

I would like to thank Dr. Baburaj Karanayil for his technical knowledge and support provided me throughout the laboratory activities at the Australian Energy Research Institute, University of New South Wales.

I would like to thank Dr. Georgios Konstantinou, Dr. Mihai Ciobotaru, Professor Josep Pou, Dr. Amer Ghias, Dr. Shamim Reza, Dr. Ricardo Aguilera, and Dr. Pablo Acuna for the productive discussions.

I am also grateful to fellow students, Ghias, Vita, Ali, Thomas, Milad, Harith, Roky and Leonardo for their support during my Ph.D. studies at the University of New South Wales.

I would like to thank my friends and colleagues in Sydney who provided me the support and encouragement.

I would like to express my deepest gratitude to my parents and my sister and brother for their unconditional support and love which enabled me to follow my dreams.

I am grateful to my wife for her understanding, encouragement, and endless support.

Finally, I would like to thank the University of New South Wales for awarding me the University International Postgraduate Award, which provided me the required financial support throughout the program.

Contents

1	Intr	oducti	on	1
	1.1	Backg	round and Motivation	1
		1.1.1	Power Converter Configurations for HESSs	2
		1.1.2	HESS Power Allocation Methods	6
	1.2	Object	tives and Contributions	7
		1.2.1	Objectives of the Research $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	7
		1.2.2	Contributions of the Research	9
	1.3	List of	Publications	11
		1.3.1	Journal Publications	11
		1.3.2	Conference Publications	11
	1.4	Thesis	Outline	12
2	Ope	eration	of the Boost Inverter	15
	2.1	Introd	uction	15
	2.2	Opera	tion of the Single Phase Grid Connected Boost Inverter \ldots .	17
	2.3	Switch	ing Frequency Ripple Component of the Boost Inverter DC Side	
		Curren	nt	19
		2.3.1	In-phase PWM Operation	20
		2.3.2	Phase Shifted Interleaved Operation	22
		2.3.3	Boost Inverter Component Selection with the Interleaved Operation	26
		2.3.4	Comparison Between the In-Phase PWM Operation and the	
			Interleaved Operation $\ldots \ldots \ldots$	27
	2.4	Conclu	isions	27
3	Inte	erleave	d Boost Inverter Based Battery-Supercapacitor Hybrid	
	Ene	ergy St	orage System	30
	3.1	Introd	uction	31
	3.2	Boost	Inverter Based Battery- Supercapacitor Hybrid Energy Storage System	31

	3.3	Contr	ol System Description	33
		3.3.1	Power Converter Control Strategy	33
		3.3.2	Hybrid Energy Storage System Grid Integration	37
	3.4	Batter	ry State of Charge Estimation and Control	39
		3.4.1	Battery Model	39
		3.4.2	Battery Model Parameter Estimation	40
		3.4.3	Extended Kalman Filter Based Battery SOC Estimation	42
		3.4.4	Battery SOC Controller	44
	3.5	Exper	imental Results	45
	3.6	Concl	usions	50
4	Sup	ercapa	acitor Energy Controlled Filter Based Battery-Supercapacitor	
	Hyl	orid E	nergy Storage System	53
	4.1	Introd	luction	53
	4.2	HESS	Operation with SCVC	55
	4.3	Super	capacitor Energy Controller	57
		4.3.1	Effects of the Supercapacitor Energy Controller on the HESS Power	
			Allocation Strategy	58
		4.3.2	Calculating the Capacitance of the Supercapacitor ${\cal C}_{sc}$ for a Set of	
			HESS Power Allocation Parameters and for a Given HESS Power	
			Profile	60
		4.3.3	Calculating the Battery Specifications for a Set of HESS Power	
			Allocation Parameters and for a Given HESS Output Power Profile	62
	4.4	HESS	Power Allocation Parameter Selection and Supercapacitor Sizing for	
		a PV	Power Smoothing Application	62
	4.5	Exper	imental Results	67
	4.6	Concl	usions	70
5	A F	ixed F	requency Sliding Mode Controller for the Boost Inverter Based	
	Hyl	orid E	nergy Storage System	71
	5.1	Introd	luction	72
	5.2	Effect	of the Inductor ESR Values on Output Capacitor Reference Voltage	
		Track	ing with DL Control Method	73
	5.3	Slidin	g Mode Controller Analysis	75
		5.3.1	Left Hand Side Supercapacitor Connected Boost Converter Leg	75
		5.3.2	Left Hand Side Battery Connected Boost Converter Leg	79
		5.3.3	Sliding Coefficients Selection for the HESS	82

		5.3.4	HESS Power Allocation	82
		5.3.5	Supercapacitor Selection for the SM Controlled HESS	85
	5.4	Exper	imental Results	86
	5.5	Conclu	usions	90
6	A F	Rule-Ba	ased Controller to Mitigate DC-Side Second-Order Harmoni	с
	Cur	rent in	n a Single Phase Boost Inverter	91
	6.1	Introd	$\operatorname{luction}$. 92
	6.2	Second	d-Order Harmonic Ripple Current Analysis and Reduction Using the	
		Wavef	Form Control Method	93
		6.2.1	Second-Order Harmonic Input Current Component in the Boost	
			Inverter	93
		6.2.2	Second-Order Harmonic Input Current Ripple Reduction Using the	
			Waveform Control Method	. 95
		6.2.3	Effect of the Waveform Control Method on the Inductor Current	
			Harmonic Content and Efficiency	. 97
		6.2.4	Selection of the Output Capacitor Capacitance C	. 99
		6.2.5	Effect of the Capacitor Tolerances and the Boost Inductor ESR	
			Values on the Waveform Control Method Performance	101
	6.3	The P	roposed Rule Based Controller for Ripple Current Mitigation	102
	6.4	Boost	Inverter Control System	107
	6.5	Exper	imental Results	110
	6.6	Conclu	usions	114
7	An	Input	Current Feedback Method to Mitigate the DC-Side Low	v
	Fre	quency	Ripple Current in a Single-Phase Boost Inverter	116
	7.1	Introd	$\operatorname{luction}$	117
	7.2	Curren	nt Feedback Ripple Current Reduction Method	118
	7.3	Boost	Inverter Control System with the Current Feedback Method \ldots .	123
	7.4	Exper	imental Results	125
	7.5	Batter	ry-Supercapacitor HESS with Reduced Low Frequency Input Current	
		Ripple	e	129
		7.5.1	Second-order Harmonic Ripple Component in the Supercapacitor	
			Current	130
		7.5.2	Current Feedback Method for the HESS	130
		7.5.3	Experimental Results	133
	7.6	Conclu	usions	136

8	Con	clusions and Future Work 13	39
	8.1	Conclusions	39
	8.2	Future Work	42
	A.1	Appendix 1	43
		A.1.1 Derivation of equation (4.8)	43
		A.1.2 Derivation of equation (5.15)	43
		A.1.3 Derivation of equation (5.26)	44
		A.1.4 Derivation of equation (5.39)	45
		A.1.5 Derivation of equation (6.40)	45
		A.1.6 Derivation of equation (7.4)	46
		A.1.7 Derivation of equation (7.10) and (7.11)	46
		A.1.8 Derivation of equation (7.24)	47
		A.1.9 Derivation of equation (7.29)	47
		A.1.10 Derivation of equation (7.33) , (7.34) , and (7.35)	48
	A.2	Appendix 2	50
		A.2.1 Simulink Models used for DSpace controller implementation 1	50

List of Figures

1.1	Types of hybrid energy storage systems	3
1.2	Types of hybrid energy storage system based on the power converter	
	configuration	3
2.1	Single phase grid connected boost inverter	17
2.2	Average model of the left hand side boost converter leg	18
2.3	Equivalent circuit of the grid connected boost inverter $\ldots \ldots \ldots \ldots$	19
2.4	Boost inverter operating waveforms	20
2.5	Ripple current waveforms when the PWM signals for two converter legs are	
	in-phase	21
2.6	Ripple current waveforms for interleaved operation when $d_1 + d_2 \ge 1$	24
2.7	Ripple current waveforms for interleaved operation when $d_1 + d_2 < 1$	25
2.8	Switching frequency ripple current amplitude comparison with the	
	interleaved and in-phase PWM operation. (a) Output capacitor voltage	
	waveforms, (b) DC source current waveform with in-phase PWM operation,	
	(c) DC source current with interleaved operation, (d) DC source current	
	switching frequency ripple amplitude and (e) ripple reduction factor	28
3.1	Single phase grid connected battery-supercapacitor hybrid energy storage	
	system	32
3.2	Overall block diagram of the proposed controller	34
3.3	Block diagram of the left hand side outer voltage control loop and inner	
	current control loops	36
3.4	Bode diagrams of the HESS controller. (a) Open loop Bode plot of the	
	left hand side voltage control loop, (b) open loop Bode plot of the current	
	control loop for the left hand side battery connected boost converter leg,	
	(c) open loop Bode plot of the current control loop for the left hand side	
	supercapacitor connected boost converter leg and (d) open loop Bode plot	
	of the SCVC.	38

Active and reactive power controller (PQ controller), and battery SOC	
controller	39
Second order RC battery model	39
Relationship between the battery OCV and SOC	41
Battery current and voltage profile for the battery RC model parameter	
estimation	42
Estimated battery SOC using the extended Kalman filter $(SOC_{batt,K})$ and	
the current integration method $(SOC_{batt,i})$	45
HESS experimental setup	46
Current waveform of the grid interfacing inductor, i_g (CH2), HESS output voltage, v_o (CH3), and grid voltage, v_a (CH4), when the HESS with	
interleaved operation delivers 30W active power to the grid	46
Inductor current waveforms i_{Lsc1} (CH1) and i_{Lbatt1} (CH2) when the HESS	
with interleaved operation delivers 30W active power to the grid	46
The total battery current, i_{batt} (CH2) and the total supercapacitor current,	
$i_{sc}~({\rm CH1})$ when the HESS with interleaved operation delivers 30W active	
power to the grid.	47
The total battery current, i_{batt} (CH2) and the total supercapacitor current,	
i_{sc} (CH1) when the HESS with in-phase PWM operation delivers 30W	
active power to the grid	47
Operation of the HESS with step changing output power reference signals	49
Example application of a direct AC line integrated battery-supercapacitor	
HESS with a renewable power generation system	49
Operation of the HESS when the power reference signals are generated to	
emulate PV power generation situation	50
Operation of the HESS when the battery SOC limits are reached	51
Outer voltage control loop for the left-hand side boost converter legs with	
SCVC	55
Simplified block diagram of the HESS power allocation with the SCVC	56
Outer voltage control loop for the left-hand side boost converter legs with	00
SCEC	57
Simplified block diagram of the HESS power allocation with the SCEC.	58
Asymptotic Bode plot of the transfer function H_{Dheff} .	59
Step response of (a) H_{Phatt} and (b) H_{dEsc} for various <i>n</i> values for a sample	
gain cross-over frequency $\omega_{crea} = 0.05 \text{ rads}^{-1}$.	61
Block diagram of the HESS in a PV power smoothing application	63
	Active and reactive power controller (PQ controller), and battery SOC controller

4.8	Power profiles considered for the HESS parameter selection. (a) PV power,	
	power injected to the grid and the HESS power and (b) the total power	
	profile of the battery and the supercapacitor	64
4.9	Histogram of the non-zero gradient values of the P_{tot} power profile \ldots	64
4.10	Variation of (a) $max(\nabla P_{batt})$ and (b) C_{sc} with n and ω_c .	65
4.11	Operation of the HESS when $\rho = 75$. (a) P_{tot} and P_{batt} power profiles, (b)	
	gradient of the battery power profile, (c) supercapacitor energy variation	
	and (d) voltage variation of the supercapacitor	66
4.12	HESS experimental setup	68
4.13	Operation of the HESS with SCEC (a) power reference of the HESS,	
	(b) variation of P_{tot} , P_{batt} and P_{sc} , (c) gradient of the P_{tot} profile, (d)	
	gradient of the battery power profile, (e) supercapacitor energy variation,	
	(f) supercapacitor voltage variation, (g) battery current variation and (h)	
	supercapacitor current variation	69
51	Comparison of the effect of the boost inductor FSP on the output connector	
5.1	voltage reference tracking with the double loop control method and with the	
	sliding mode control method (a) DC shift of the output capacitor voltage	
	and (b) DC shift error of the output capacitor voltage	74
5.2	Single-phase grid-connected battery-supercapacitor hybrid energy storage	11
0.2	system	76
5.3	Equivalent circuit of the HESS in the viewpoint of the left hand side	
	supercapacitor connected boost converter leg	77
5.4	Equivalent circuit of the HESS in the viewpoint of the left hand side battery	
	connected boost converter leg	80
5.5	Block diagram of the proposed sliding mode controller.	81
5.6	Equivalent block diagram of the HESS power allocation	83
5.7	Asymptotic Bode plot of H_{Pbatt} with real system poles	84
5.8	Effect of $K_{p,Esc}$ and $K_{i,Esc}$ on H_{Pbatt} and H_{dEsc} . (a) pole zero plot of	
	H_{Pbatt} , (b) step response of H_{Pbatt} and (c) step response of H_{dEsc}	85
5.9	The output capacitor voltage waveforms v_{o1} and v_{o2} , and the HESS output	
	voltage v_o when the HESS delivers 50W active power and 25VAr reactive	
	power to the grid \ldots	86
5.10	The grid voltage v_g , the HESS output voltage v_o and the HESS output	
	current i_g when the HESS delivers 50W active power and 25VAr reactive	
	power to the grid \ldots	87

5.11	The inductor current waveforms i_{Lbatt1} and i_{Lsc1} when the HESS delivers
	50W active power and 25VAr reactive power to the grid
5.12	Experimental waveforms illustrating the operation of the HESS:
	Supercapacitor responds to sudden power changes
5.13	Experimental waveforms illustrating the operation of the HESS in a PV
	power smoothing application
6.1	Single phase grid connected boost inverter
6.2	AC components of the inductor current i_{L1} , without and with the waveform
	control method when delivering $10W$ active power and 15 VAr reactive
	power to the grid
6.3	Relationships between B_{max} , $v_{o1,max}$ and $V_{DC,min}$ versus the output
	capacitor C values during operation with the waveform control 100
6.4	Ripple reduction factor for various values of the inductor ESR values R_L and
	capacitance C of the output capacitors when the waveform control method
	is used (inverter supplies 10W active power and 15VAr reactive power). $\ . \ . \ 103$
6.5	Second-order harmonic ripple current amplitude as a function of B and φ
	when (a) inverter supplies 15W active power and 10VAr reactive power,(b)
	inverter supplies 15W active power and absorbs 10VAr reactive power, (c)
	inverter absorbs 15W active power and supplies 10VAr reactive power and
	(d) inverter absorbs 15W active power and 10VAr reactive power 104
6.6	Flow diagram of the proposed rule-based controller for the second-order
	harmonic ripple current reduction
6.7	Second order harmonic ripple current amplitude measurement technique 106
6.8	Overall block diagram of the proposed control system $\ldots \ldots \ldots$
6.9	Outer voltage control loop and inner current control loop for the left hand
	side boost converter leg

6.10	Experimental results illustrating operation of the battery storage system	
	when delivering 15VAr reactive power and 10W active power. (a) Output	
	capacitor voltage waveforms and inverter output voltage without the	
	rule-based controller, (b) Output capacitor voltage waveforms and inverter	
	output voltage with the rule-based controller, (c) Inverter output voltage,	
	grid voltage and inverter output current without the rule-based controller,	
	(d) Inverter output voltage, grid voltage and inverter output current with	
	the rule-based controller, (e) Inductor currents and total input current of	
	the boost inverter without the rule-based controller, (f) Inductor currents	
	and total input current of the boost inverter with the rule-based controller,	
	(g) Frequency characteristics of the input current without the rule-based	
	controller and (h) Frequency characteristics of the input current with the	
	rule-based controller	. 111
6.11	Experimental results illustrating operation of the proposed rule-based	
	controller when the battery storage system supplies 10W active power and	
	15 VAr reactive power	. 112
6.12	Operation of the rule-based controller with step-changing inverter output	
	power	. 112
6.13	Experimental results illustrating the effect of capacitor tolerances on the	
	ripple current reduction when the waveform control method is used (a)	
	Inductor currents and total input current of the boost inverter without	
	any ripple current reduction method, (b) Inductor currents and total input	
	current of the boost inverter with the waveform control (WFC) method and	
	$C{=}60\mu\mathrm{F},$ (c) Inductor currents and total input current of the boost inverter	
	with the waveform control method and $C{=}54\mu\mathrm{F},$ (d) Inductor currents and	
	total input current of the boost inverter with the waveform control method	
	and $C{=}66\mu$ F, (e) Inductor currents and total input current of the boost	
	inverter with the rule-based controller (RBC), (f) Comparison of the ripple	
	amplitude and ripple reduction factor for cases (b) to (e). \ldots	. 113
6.14	Comparison of the ripple current reduction using the waveform control	
	method (WFC) and the rule-based controller (RBC)	. 114
7.1	Single phase grid connected boost inverter	. 118
7.2	Numerical solution of the variation of the input current with K , (a) the	
	second and the fourth-order harmonic current ripple amplitude, (b) change	
	in the second-order harmonic ripple amplitude for a unit change in K .	. 120

7.3	Comparison of the numerical and approximated analytical solution, (a)	
	the second-order harmonic current ripple amplitude, (b) error between the	
	numerical and analytical solution	. 122
7.4	Illustration of the effect of (a) the output capacitor tolerance and (b) the	
	input voltage variation on the ripple reduction method performance when	
	P=15W and $Q=10VAr$. 123
7.5	Overall block diagram of the control system	. 124
7.6	Illustration of the required feedback gains K to satisfy (7.12) for each output	
	power condition.	. 125
7.7	Error between the numerical solution and the analytical solution for all	
	output power conditions and $K=100$. 125
7.8	Experimental results illustrating operation of the battery storage system	
	when delivering 15W active power and 10VAr reactive power. (a) Output	
	capacitor voltage waveforms and inverter output voltage without the	
	proposed controller, (b) Output capacitor voltage waveforms and inverter	
	output voltage with the proposed controller, (c) Inverter output voltage,	
	grid voltage and inverter output current without the proposed controller,	
	(d) Inverter output voltage, grid voltage and inverter output current with	
	the proposed controller, (e) Inductor currents and total input current of	
	the boost inverter without the proposed controller, (f) Inductor currents	
	and total input current of the boost inverter with the proposed controller,	
	(g) Frequency characteristics of the input current without the proposed	
	controller and (h) Frequency characteristics of the input current with the	
	proposed controller.	. 127
7.9	Experimental results illustrating transient operation of the proposed control	
	system in response to 15W and 10VAr power step change, (a) Inverter	
	output active power, (b) inverter output reactive power (c) second-order	
	harmonic ripple amplitude with and without the proposed method. \ldots	. 128
7.10	Comparison of the ripple current reduction using the waveform control	
	method, the rule-based control method and the proposed current feedback	
	method (K =100)	. 129
7.11	Experimental results illustrating change in the second-order harmonic ripple	
	current amplitude $\Delta m_{2\omega}$ for a unit change in K for all output power	
	conditions.	. 129
7.12	Single-phase grid-connected battery-supercapacitor hybrid energy storage	
	system	. 131

7.13 Variation of the supercapacitor current numerical solution with K , (a) the
second and the fourth-order harmonic current ripple amplitudes, (b) change
in the second-order harmonic ripple amplitude for a unit change in K 133
7.14 Overall block diagram of the control system
7.15 Illustration of the required feedback gains K to satisfy (7.36) for each output
power condition. $\ldots \ldots 135$
7.16 Experimental waveforms of the HESS when delivering 15W active power
and 10VAr reactive power. (a) Output capacitor voltage and inverter
output voltage without the ripple reduction method, (b) output capacitor
voltage and inverter output voltage with the ripple reduction method, (c)
battery and supercapacitor currents without the ripple reduction method,
(d) battery and supercapacitor currents with the ripple reduction method,
(e) frequency characteristics of the supercapacitor current without the ripple
reduction method and (f) frequency characteristics of the supercapacitor
current with the ripple reduction method
7.17 Operation of the HESS with the current feedback ripple reduction method
for a step changing output power profile
A.1.1Simplified block diagram of the HESS power allocation with the SCEC 143
A.1.2Simplified block diagram of the HESS power allocation
A.2.1Simulink blocks: Duty ratio signal output to DSpace system
A.2.2Simulink blocks: Voltage and current inputs through DSpace analog to
digital converter
A.2.3Simulink blocks: Active and reactive power controller and double loop
control system $\ldots \ldots \ldots$

List of Tables

1.1	The publications relevant to this thesis
2.1	Boost inverter parameters
3.1	Estimated battery RC model parameters
3.2	Hybrid energy storage system parameters
4.1	HESS power allocation parameters and supercapacitor values for various
	$\nabla P_{batt, limit}$ values with and without SCEC
4.2	Hybrid energy storage system parameters
5.1	Hybrid energy storage system parameters
6.1	Experimental prototype parameters
6.2	Second-order harmonic ripple current mitigation for various output power
	conditions
7.1	Experimental prototype parameters

Nomenclature

AC	Alternating current
DC	Direct current
DL	Double loop
ESR	Equivalent series resistance
ESS	Energy storage system
FFT	Fast Fourier transform
HESS	Hybrid energy storage system
$LiFePO_4$	Lithium iron Phosphate
OCV	Open circuit voltage
P&O	Perturb and observe
PCC	Point of common coupling
PI	Proportional integral
PLL	Phase lock loop
PR	Proportional resonant
PV	Photovoltaic
PWM	Pulse width modulation
QSG	Quadrature signal generator
SCEC	Supercapacitor energy controller
SCVC	Supercapacitor voltage controller
SM	Sliding mode

SOC	State of charge
SOGI	Second-order generalized integrator

List of Variables and List of Greek Symbols

ω	Frequency of the inverter output voltage in rads ⁻¹
δ	Phase angle of the inverter output voltage in rad
η	Efficiency of the HESS
ω_{HPF}	Cut-off frequency of the high-pass filter used for HESS power allocation in rads $^{-1}$
$\nabla P_{batt,limit}$	Predefined maximum allowable gradient value for the battery power profile in W/s
C_1	Left hand side output capacitor in μF
C_2	Right hand side output capacitor in $\mu {\rm F}$
C_{batt}	Energy capacity of the battery in Ah
C_{sc}	Capacitance of the supercapacitor in $\mu {\rm F}$
d_1	Duty cycle time average value of the switch $Sw_{1,B}$
d_{batt1}	Duty cycle time average value of the switch $Sw_{batt1,B}$
d_{sc1}	Duty cycle time average value of the switch $Sw_{sc1,B}$
dE_{batt}	Change in the stored energy in the battery in J
dE_{sc}	Change in the stored energy in the supercapacitor in J
E_o	Open circuit voltage of the battery in V
E_{sc}	Energy stored in the supercapacitor in J
$E_{sc,init}$	Initial stored energy of the supercapacitor in J

$E_{sc,max}$	Maximum stored energy of the supercapacitor in J
$E_{sc,min}$	Minimum stored energy of the supercapacitor in J
$E_{sc,ref}$	Reference energy level for the supercapacitorin J
$E_{sc,u}$	Utilized energy of the supercapacitor in J
f	Grid voltage frequency in Hz
f_{sw}	Switching frequency in kHz
i_{batt}	Battery current in A
i_{C1}	Current through capacitor C_1 in A
i_{L1}	Current through inductor L_1 in A
i_{Lbatt1}	Current through the inductor L_{batt1} in A
i_{Lsc1}	Current through the inductor L_{sc1} in A
i_{o1}	Output current from the left hand side boost converter leg(s) in A
i_{sc}	Supercapacitor current in A
K	Feedback gain for the current feedback ripple reduction method
k_{batt1}	SM controller gain for the battery connected converter legs
$k_{batt1,m}$	SM controller gain for the battery connected converter legs
$K_i(PI_{batt})$	Integral gain of the inner i_{Lbatt} current controller
$K_i(PI_P)$	Integral gain of the active power controller
$K_i(PI_Q)$	Integral gain of the reactive power controller
$K_i(PI_{sc})$	Integral gain of the inner i_{Lsc} current controller
$K_i(PI_{scv})$	Integral gain of the supercapacitor voltage controller
$K_i(PI_{SOC})$	Integral gain of the battery SOC controller
$K_i(PR)$	Integral gain of the PR controller
$K_{i,Esc}$	Integral gain of the supercapacitor energy controller
K_{Kalman}	Kalman filter gain for the battery SOC estimator

$K_p(PI_{batt})$	Proportional gain of the inner i_{Lbatt} current controller
$K_p(PI_P)$	Proportional gain of the active power controller
$K_p(PI_Q)$	Proportional gain of the reactive power controller
$K_p(PI_{sc})$	Proportional gain of the inner i_{Lsc} current controller
$K_p(PI_{scv})$	Proportional gain of the supercapacitor voltage controller
$K_p(PI_{SOC})$	Proportional gain of the battery SOC controller
$K_p(PR)$	Proportional gain of the PR controller
$K_{p,Esc}$	Proportional gain of the supercapacitor energy controller
$K_{p,vsc}$	Supercapacitor voltage controller gain
$k_{sc1,a}$	SM controller gain for the supercapacitor connected converter legs
$k_{sc1,b}$	SM controller gain for the supercapacitor connected converter legs
$k_{sc1,c}$	SM controller gain for the supercapacitor connected converter legs
$k_{sc1,m}$	SM controller gain for the supercapacitor connected converter legs
L_1	Left hand side boost inductor in μH
L_2	Right hand side boost inductor in μH
L_{batt1}	Left hand side battery connected boost inductor in $\mu {\rm H}$
L_g	Grid interfacing inductor in μH
L_{sc1}	Left hand side supercapacitor connected boost inductor in $\mu {\rm H}$
N_{arphi}	Gain for the rule based controller
N_B	Gain for the rule based controller
Р	Active power supplied by the inverter in W
P_{batt}	Power supplied by the battery in W
P_{HESS}	Active power supplied by the HESS in W
P_{Kalman}	Kalman filter error covariance matrix for the battery SOC estimator
P_{sc}	Power supplied by the supercapacitor in W

P_{tot}	Total input power supplied by both the battery and the supercapacitor in W
P_{tot1}	Total instantaneous power requirement from the left hand side boost converter legs in W
Q	Reactive power supplied by the inverter in VAr
Q_{HESS}	Reactive power supplied by the HESS in VAr
Q_{Kalman}	Kalman filter process covariance matrix for the battery SOC estimator
R_{C1}	ESR value of the output capacitor C_1 in Ω
R_{Kalman}	Kalman filter measurement covariance matrix for the battery SOC estimator
R_{L1}	ESR value of the inductor L_1 in Ω
R_{Lbatt1}	ESR value of the inductor L_{batt1} in Ω
R_{Lsc1}	ESR value of the inductor L_{sc1} in Ω
R_{RBC}	Second-order harmonic input current ripple amplitude with the rule based control method in A
R_{WFC}	Second-order harmonic input current ripple amplitude with the waveform control method in A
SOC_{batt}	SOC of the battery
T_D	Time delay between the perturbations and measurements for the rule based controller in s
T_s	Switching time period in s
v_{batt}	Battery terminal voltage in V
$V_{DC,ref}$	Reference DC shift of the output capacitor voltage in V
V_{DC1}	DC shift of the voltage across the output capacitor C_1 in V
V_{DC2}	DC shift of the voltage across the output capacitor C_2 in V
V_g	Grid voltage amplitude in V
v_{in}	Input voltage to the boost inverter in V

v_{Lbatt1}	Voltage across the inductor L_{batt1} in V
v_{Lsc1}	Voltage across the inductor L_{sc1} in V
v_o	Inverter output voltage in V
V_o	Amplitude of the inverter output voltage in V
v_{o1}	Voltage across output capacitor C_1 in V
v_{o2}	Voltage across output capacitor C_2 in V
v_{sc}	Supercapacitor terminal voltage in V
$v_{sc,init}$	Initial terminal voltage of the supercapacitor in V
$v_{sc,max}$	Maximum allowed supercapacitor voltage in V
$v_{sc,min}$	Minimum allowed supercapacitor voltage in V
$v_{sc,ref}$	Reference level for the supercapacitor voltage in V

Chapter 1

Introduction

1.1 Background and Motivation

Grid connected energy storage systems (ESSs) have many applications in modern electric grid such as renewable energy time shifting, intermittency handling, and power quality improvement [1–3]. Many energy storage technologies are available for grid connected applications depending on the required power density and energy density [4]. Battery ESSs are popular in residential and distributed level power networks due to its ease of implementation and geographical independence as compared to other energy storage technologies.

Battery ESSs for renewable energy applications should be designed with a higher power density to withstand the high and rapid power variations associated with the intermittent renewable power generation [5]. However, batteries have limited power capability and hence, a large number of batteries may need to be connected in parallel to obtain the required power capability. Additionally, battery ESSs should have a sufficient energy capacity to support the renewable energy time shifting.

However, the battery ESSs have a limited cycle life and the lifetime depends on the operating conditions of the battery. Numerous battery lifetime degradation factors can be identified as follows.

- 1. *High current amplitude*: High current amplitudes are identified as a major battery lifetime degradation factor [6, 7]. In renewable energy applications, battery ESSs have to handle high power variations and hence battery has to sustain high current rates even with a large number of parallel battery modules.
- 2. Battery current fluctuations and micro-cycles: Battery current fluctuations are another factor affecting the battery lifetime [8–10]. Furthermore, such current fluctuations can increase the number of battery charge discharge micro-cycles (short

charge-discharge cycles [11]) which reduces the effective cycle life of the battery [11, 12]. Due to the intermittency of renewable power generation, battery current fluctuations and micro-cycles are imminent in such battery ESSs.

3. *Ripple current*: Continuous ripple components in the battery current can increase the internal heating of the battery which causes a degradation in battery lifetime [13,14]. In single phase grid connected applications, a second-order harmonic ripple component appears in the battery current. Second-order harmonic component in the battery current increases as the inverter output power increases. Additionally, a switching frequency ripple current is also present in the battery current waveform due to the operation of the power converters. Both the second-order harmonic ripple current as well as the switching frequency current component adversely affect the lifetime of the battery [13–15].

Recently the supercapacitor technology has become popular in ESS applications due to its higher power capability and superior cycle life compared to the battery technology [16]. Typically, supercapacitors have 1 to 10 Wh/kg specific energy and 1000 to 5000 W/kg specific power. The charge/discharge efficiency of supercapacitors is very high, ranging from 85% to 98% [17]. In contrast, Lithium batteries usually have 50 to 500W/kg specific energy and 10 to 500W/kg specific power. Additionally, its charge/discharge efficiency is typically between 75% to 90%. [17]. Battery-supercapacitor hybrid energy storage systems (HESSs) have been proposed in the literature as a way of increasing the power capability of the energy storage systems and extending the lifetime of the battery [6,10,12,16–53]. HESSs combine individual advantages of both the battery and the supercapacitor and create a single ESS with increased power and energy capabilities. These HESSs are classified based on the power converter configuration and power allocation methods in Section 11.1 and Section 11.2.

Battery-supercapacitor HESSs have been proposed for various applications such as electric vehicles and grid energy storage systems. In both electric vehicle and grid energy storage system applications, the supercapacitors increase the energy storage system power capability and the battery lifetime. Numerous battery-supercapacitor HESSs were studied in the literature, and they can be classified based on the power converter configuration as shown in Figs. 1.1 and 1.2.

1.1.1 Power Converter Configurations for HESSs

HESS can be categorized into two main groups, namely passive HESSs and active HESSs [18]. In passive HESSs, the battery and the supercapacitor terminals are



Fig. 1.1: Types of hybrid energy storage systems



Fig. 1.2: Types of hybrid energy storage system based on the power converter configuration

interconnected without using an interfacing power converter as illustrated in Fig. 1.2(a) [18, 51–53]. In this configuration, power allocation between the battery and the supercapacitor is uncontrollable and depends on the impedance of the battery and the supercapacitor. The battery voltage has to be controlled in a range suitable for the DC/AC converter operation. Hence, the complete operating range of the battery may not be utilized. The supercapacitor voltage is controlled by the battery voltage,

and hence, the supercapacitor cannot charge or discharge independently. Because of that energy stored inside the supercapacitor cannot be utilized properly in passive HESSs [17]. To alleviate the inherent disadvantages of the passive HESS, active HESSs have been proposed. In the active HESS, power converters are used to interface the battery and/or the supercapacitor to a DC link or to an AC line. DC link connected battery-supercapacitor HESSs can be categorized into four groups as illustrated in Figs. 1.1 and 1.2.

- 1. Only the battery is connected through a DC/ DC converter: In this case, the battery is connected through a DC/DC converter and the supercapacitor is directly connected to the DC/AC converter as shown in Fig. 1.2(b) [17–22]. The battery interfacing DC/DC converter enables the maximum utilization of the battery operating region and usage of low voltage battery modules. However, voltage variation of the supercapacitor has to be controlled to maintain the required DC link voltage. Hence, the energy stored in the supercapacitor cannot be utilized properly. Additionally, a large number of supercapacitor cells may need to be connected in series to support the required DC link voltage.
- 2. Only the supercapacitor is connected through a DC/DC converter: The battery is directly connected to the DC link while the supercapacitor is connected through a DC/DC converter as depicted in Fig. 1.2(c) [12, 23–26]. A high voltage battery system may be required due to the direct DC link connection of the battery. Furthermore, the operating range of the battery may have to be limited to maintain the DC link voltage. However, a low voltage supercapacitor module can be employed as a result of the interfacing DC/DC converter and also this enables a broader operating voltage range to the supercapacitor which helps to increase its energy utilization.
- 3. Cascade connection using DC/DC converters: A cascaded battery-supercapacitor HESS is illustrated in Fig. 1.2(d) and an alternative cascaded system can be obtained by interchanging the battery and the supercapacitor positions [18,30,42]. Both the battery and the supercapacitor modules can have lower voltages compared to the DC link voltage. Moreover, the DC/DC converters enable the full utilization of the operating regions of the battery as well as the supercapacitor. However, the introduction of the additional DC/DC converter can reduce the efficiency of the system compared to the previous solutions and the DC/DC converter connected to the DC link should be designed such that it can handle the full amount of HESS

power.

4. Parallel connection using DC/DC converters: Parallel connection is an alternative way to the cascade connection and in this configuration, two DC/DC converters are used to interface the battery and the supercapacitor to a DC link as shown in Fig. 1.2(e) [6, 10, 16, 18, 27–43]. The battery and the supercapacitor can have lower voltages compared to the DC link voltage and the full operating regions of the DC sources can be utilized. Since only a fraction of the total HESS power is supplied by the battery and the supercapacitor, the interfacing DC/DC converters can be designed with a lower power rating than the DC/AC converter power rating. The battery and the supercapacitor currents can be controlled individually based on the power requirement and the configuration has a degree of fault tolerance since the system can operate even when a failure occurs in the battery or the supercapacitor HESSs proposed in the literature due to the above mentioned advantages.

Foregoing active HESS structures in the single phase grid connected applications contain at least a DC/DC converter and a DC/AC converter. Hence, a DC to DC power conversion stage followed by a DC to AC power conversion stage can be identified and this leads to a reduction in the conversion efficiency. Direct AC line connected ESSs were proposed in the literature as a way of increasing the system power conversion efficiency by minimizing the number of power processing stages [44,49,54–56]. Authors in [57] reported approximately 5% greater power conversion efficiency for a single stage ESS compared to a two stage ESS. Various types of direct AC line connected battery-supercapacitor HESSs were proposed in the literature using numerous power converter topologies [44–50].

Two parallel inverter based battery-supercapacitor HESS topology is shown in Fig. 1.2(f) [47–50]. In this topology, the battery and the supercapacitor operating voltage ranges have to be selected based on the DC/AC power converter configuration. For example, if the H-bridge inverters are used as the DC/AC converters, the battery and the supercapacitor must have higher voltage values compared to the peak output voltage. Alternatively, DC/AC converters with boosting capability allow using low voltage battery and supercapacitor modules. The parallel inverter configuration provides redundancy and independent control of the battery and the supercapacitor currents.

Few direct AC line connected battery-supercapacitor HESSs were proposed using three level neutral point calmed inverter [44, 45] and matrix converter [56] for three phase applications. However, most of the battery-supercapacitor HESSs proposed in the literature are based on the DC link connected structure and consequently, many design issues related to the implementation of the direct grid integrated battery-supercapacitor HESSs such as second-order harmonic ripple current reduction and controller implementation have not been addressed.

1.1.2 HESS Power Allocation Methods

Various power allocation methods were studied for the HESSs. The most basic battery-supercapacitor HESS power allocation approach is a filter based method. In the filter based power allocation method, a low pass filter or a high pass filter is employed to divert the high frequency power variations to the supercapacitor while the battery responds to the low frequency power requirement [12, 27, 37, 43, 46, 58]. The filter based HESSs reduce the battery current fluctuations, the second-order harmonic ripple component of the battery current and also indirectly reduce the peak power requirement of the battery. In some cases, an additional supercapacitor voltage controller is employed with the power allocation filter to maintain the supercapacitor voltage around a reference value [12, 27].

Battery power limiting based HESS power allocation is another commonly used approach. In this method, the power supplied by the battery is limited to a pre-defined maximum power value and the required balance power component is allocated to the supercapacitor [6, 17, 24, 59]. This method reduces the peak power requirement of the battery and hence reduces the peak battery current. For example, authors in [6] reported 49% to 80% peak battery current reduction. However, the battery current waveform is not smoothened and furthermore, this method is not effective in reducing the battery current ripple components.

Many other alternative power allocation methods were studied in the literature. In [10], a battery power ramp rate control method was proposed to allocate the power components between the battery and the supercapacitor. Predictive algorithms were used to allocate the power components in [28, 30] while [35, 60] employed fuzzy logic control algorithms. In [23], the authors used neural networks to achieve the power allocation and the authors in [34,42] used optimization methods. The authors in [50] employed a wavelet based power allocation method for battery-supercapacitor HESS. Even though many different power allocation methods were studied for the battery-supercapacitor HESSs, the main objective of all the methods can be identified as limiting the battery current fluctuations and the peak battery power while the supercapacitor responds to the fast, high and varying power components.

After studying the existing battery-supercapacitor HESSs, the following main

limitations can be identified in the previously proposed single phase battery-supercapacitor HESSs.

- 1. Majority of the HESSs were designed using the DC link based converter configurations. In the DC link based ESSs, multiple power processing stages can be observed which reduces the efficiency of the system. Hence, direct AC line integrated ESSs were proposed as a way of reducing the power processing stages. Even though a few direct AC line integrated HESSs were studied in the literature, power allocation methods and energy storage element sizing were not analyzed. Furthermore, none of the single phase direct grid connected battery-supercapacitor HESSs were justified using experimental results.
- 2. None of the previously proposed battery-supercapacitor HESSs considered the switching frequency ripple component reduction in the battery and the supercapacitor currents. The switching frequency ripple current component increases the DC source internal temperature and degrades its lifetime.
- 3. In the battery-supercapacitor HESSs proposed for single phase applications (DC link based and direct AC line integrated), the second-order harmonic ripple current component was allocated to the supercapacitor. However, according to [61], such continuous ripple current leads to supercapacitor overheating which in turns reduces its lifetime [61].

1.2 Objectives and Contributions

1.2.1 Objectives of the Research

The key objective of the research is to study and develop a single phase direct grid integrated battery-supercapacitor HESS and the associated sub-objectives of the research can be identified as follows.

1. Develop a power allocation strategy between the battery and the supercapacitor, and identify a suitable storage elements sizing method.

Power allocation strategy is one of the key aspects of the HESS design. In the HESS, fast and large power fluctuations has to be allocated to the supercapacitor while the battery supplies the steady power requirement. This helps to reduce the power requirement from the battery and also the reduction in battery current fluctuations helps to increase the battery lifetime. Energy storage element sizing is strongly coupled with the HESS power allocation method. The battery and the

supercapacitor have to be properly sized based on the designed power allocation method to ensure proper operation of the HESS for a considered application.

2. Design the HESS with a proper battery and supercapacitor SOC control method.

The battery and the supercapacitor SOC control is essential in HESSs to avoid energy storage element over-charging and over-discharging. Over-charging of the energy storage elements can lead to hazardous situations while over-discharging leads to a degradation of the battery and the supercapacitor lifetime. Furthermore, low battery and supercapacitor voltages reduce the efficiency of the HESS. Hence, it is important to control the battery and the supercapacitor SOC to achieve the desired HESS performances.

3. Apply advanced control methods to achieve better performance and ease of implementation.

Advanced control methods are essential in HESSs in order to incorporate the non-linear behavior of the power converters and also to improve the performance of the system.

4. Identify ways of reducing the switching frequency ripple components in the battery and the supercapacitor current.

Due to the switching operation of the power converters, the battery and the supercapacitor have to sustain a switching frequency ripple current component. The ripple current components increase the internal heating of the energy storage elements. Hence, by developing a suitable switching frequency ripple current reduction method, both the battery and the supercapacitor internal heating can be reduced.

5. Develop ways of mitigating the second-order harmonic ripple component from the supercapacitor current.

Due to the single phase operation of the HESS, a second-order harmonic power component has to be supplied by the HESS and the relevant second-order harmonic current component appears in the DC side of the power converter. The second-order harmonic ripple current component adversely affects the lifetime of both the battery and the supercapacitor. Hence, it is important to identify suitable ways to mitigate the ripple current component in both the battery and supercapacitor currents.

6. Experimentally validate the designed single phase grid connected battery-supercapacitor HESS.
In order to demonstrate the applicability of the proposed HESS and control methods in an actual environment, it is important to validate the performance of the HESS experimentally.

1.2.2 Contributions of the Research

The contributions of this research can be summarized as follows.

- 1. Chapter 2 and Chapter 3: An interleaved boost inverter based single phase grid integrated battery-supercapacitor HESS was proposed in [62–64]. The interleaved boost inverter based HESS provided both boosting and inversion functions in a single power processing stage and the interleaved operation reduced the switching frequency ripple component of both the battery and the supercapacitor currents. The proposed HESS was able to allocate the fast power fluctuations to the supercapacitor while the battery responds to the slow varying power component. In this thesis, power fluctuations greater than a pre-determined cut-off frequency is considered as fast power fluctuations. The cut-off frequency or the power allocation frequency is a design parameter of the HESS which is selected at the design phase of the HESS. The proposed HESS controller enabled the HESS operation within the battery SOC limits. A supercapacitor voltage controller (SCVC) was employed to maintain the supercapacitor voltage around a reference voltage level to reduce the risk of supercapacitor over-charging and over-discharging. The performance of the proposed HESS was experimentally verified in [64]. The proposed HESS was the first single phase direct grid integrated battery-supercapacitor HESS which was validated experimentally. Furthermore, the proposed HESS was the only battery-supercapacitor HESS which was able to reduce the switching frequency ripple components of both the battery and the supercapacitor current waveforms using a phase shifted interleaved operation [64].
- 2. Chapter 4: A supercapacitor energy controller (SCEC) and a high-pass filter based power allocation strategy was proposed in [65] to circumvent the issues associated with the nonlinearity of the conventional SCVC and a high-pass filter based power allocation method. The dynamic performance of the supercapacitor energy controlled filter based battery-supercapacitor HESS was extensively analyzed. Then, a HESS power allocation parameter selection method and an energy storage element sizing method were proposed for a supercapacitor energy controlled filter based HESS [65]. The proposed method enabled the dynamic analysis of the power allocation method and precise selection of power allocation parameters and energy

storage element sizes for a considered application.

- 3. Chapter 5: A novel fixed frequency sliding mode (SM) controlled boost inverter based HESS was proposed in [66] to achieve better output capacitor voltage reference following performance and hence to reduce the risk of DC current injection to the grid. The fixed frequency SM controller was able to overcome the issues associated with the traditional SM controllers for the boost inverter topology such as high and variable frequency switching operation and component selection complexity. In order to allocate the high frequency power components to the supercapacitor, a PI controller based SCEC was proposed and the dynamic performance of the HESS power allocation method was analyzed. Furthermore, compared to the existing fixed frequency SM controllers for the boost DC/DC converters, the SM controller was implemented using variable amplitude PWM carrier signals generated using the output capacitor voltage and inductor current measurements thus eliminating the requirement of the output capacitor currents measurement.
- 4. Chapter 6: A rule-based controller was proposed and analyzed to reduce the second-order harmonic ripple component of the boost inverter input current [67]. The method was designed to eliminate the limitations associated with the existing waveform control ripple reduction method. The ripple reduction performance of the waveform control method was affected by the boost inverter internal resistance and parameter variations. The rule-based controller was able to reduce the second-order harmonic ripple current by at least a factor of 6.
- 5. Chapter 7: A novel current feedback ripple reduction method was proposed to mitigate the second-order harmonic ripple component in the boost inverter input current [68]. The method demonstrated superior performance compared to the existing methods by achieving a significant second-order harmonic ripple current reduction without increasing other harmonic current components. The current feedback ripple reduction method was able to reduce the second-order harmonic current component by at least a factor of 20. Additionally, the method was able to achieve ripple current reduction even during output power transient conditions. Then, the current feedback ripple reduction method was applied to the boost inverter based HESS to mitigate the second order harmonic ripple component in the supercapacitor current [69]. None of the existing battery-supercapacitor HESSs designed for single phase applications considered the second-order harmonic ripple component reduction in the supercapacitor current.

1.3 List of Publications

1.3.1 Journal Publications

- [J1] Abeywardana, D.B.W.; Hredzak, B.; Agelidis, V.G., "Single-Phase Grid-Connected LiFePO4 Battery-Supercapacitor Hybrid Energy Storage System with Interleaved Boost Inverter," in *IEEE Transactions on Power Electronics*, vol.30, no.10, pp.5591-5604, Oct. 2015 doi: 10.1109/TPEL.2014.2372774
- [J2] Abeywardana, D.B.W.; Hredzak, B.; Agelidis, V.G., "A Rule-Based Controller to Mitigate DC-Side Second-Order Harmonic Current in a Single-Phase Boost Inverter," in *IEEE Transactions on Power Electronics*, vol.31, no.2, pp.1665-1679, Feb. 2016 doi: 10.1109/TPEL.2015.2421494
- [J3] Abeywardana, D.B.W.; Hredzak, B.; Agelidis, V.G., "An Input Current Feedback Method to Mitigate the DC-Side Low-Frequency Ripple Current in a Single-Phase Boost Inverter," in *IEEE Transactions on Power Electronics*, vol. 31, no. 6, pp. 4594-4603, June 2016. doi: 10.1109/TPEL.2015.2473170
- [J4] Abeywardana, D.B.W; Hredzak, B.; Agelidis, V.G., "A Fixed Frequency Sliding Mode Controller for a Boost Inverter Based Battery-Supercapacitor Hybrid Energy Storage System," in *IEEE Transactions on Power Electronics*, vol.PP, no.99, pp.1-1 doi: 10.1109/TPEL.2016.2527051
- [J5] Abeywardana, D.B.W; Hredzak, B.; Agelidis, V.G; Demetriades, G., "Supercapacitor Sizing Method for Energy Controlled Filter Based Hybrid Energy Storage Systems," in IEEE Transactions on Power Electronics , vol.PP, no.99, pp.1-1 doi: 10.1109/TPEL.2016.2552198

1.3.2 Conference Publications

- [C1] Abeywardana, D.B.W.; Hredzak, B.; Agelidis, V.G., "An Integration Scheme for a Direct AC Line Battery-Supercapacitor Hybrid Energy Storage System," in 2013 Australasian Universities Power Engineering Conference (AUPEC), vol., no., pp.1-6, Sept. 29 2013-Oct. 3 2013 doi: 10.1109/AUPEC.2013.6725474
- [C2] Abeywardana, D.B.W.; Hredzak, B.; Agelidis, V.G., "A Single Phase Grid Integration Scheme for Battery-Supercapacitor AC Line Hybrid Storage System," in 2014 IEEE International Conference on Industrial Technology (ICIT), vol., no., pp.235-240, Feb. 26 2014-March 1 2014 doi: 10.1109/ICIT.2014.6894873

Thesis Chapter	Publication
Chapter 2	[J1]
Chapter 3	[J1],[C1],[C2]
Chapter 4	[J5]
Chapter 5	[J4]
Chapter 6	[J2]
Chapter 7	[J3], [C3]

Table 1.1: The publications relevant to this thesis

[C3] Abeywardana, D.B.W.; Hredzak, B.; Agelidis, V.G., "Battery-supercapacitor hybrid energy storage system with reduced low frequency input current ripple," 2015 International Conference on Renewable Energy Research and Applications (ICRERA), Palermo, Italy, 2015, pp. 328-332. doi: 10.1109/ICRERA.2015.7418719

The journal and conference publications related to the thesis chapters are as follows.

1.4 Thesis Outline

The remainder of the thesis is organized as follows.

- Chapter 2 discusses operation of the boost inverter topology which is used to design the single phase grid connected battery-supercapacitor HESS. Switching frequency ripple component of the boost inverter DC side current is analyzed with the conventional in-phase PWM operation. Then, a phase shifted interleaved operation for the boost inverter topology is proposed and its effectiveness in reducing the switching frequency ripple component of the boost inverter DC side current is illustrated.
- Chapter 3 presents the proposed interleaved boost inverter based single phase grid connected battery-supercapacitor HESS. At the initial stage of the research, a filter based power allocation method with a SCVC is employed to achieve the power allocation between the battery and the supercapacitor. A double-loop (DL) control strategy is used to control the power converter structure. A battery SOC controller with an extended Kalman filter based battery SOC estimation method is developed. Experimental results are used to validate the operation of the proposed interleaved boost inverter based battery-supercapacitor HESS.
- Chapter 4 analyzes the power allocation between the battery and the supercapacitor for a supercapacitor SOC controlled filter based battery-supercapacitor HESS. A

SCEC and a filter based power allocation method is proposed to circumvent the challenges associated with the non-linearity of the SCVC and a filter based HESS power allocation method [65]. Furthermore, a power allocation parameter selection method and an energy storage element sizing method for a supercapacitor energy controlled filter based HESS are proposed for a given HESS application.

- Chapter 5 presents the proposed fixed-frequency SM controller for the boost inverter based battery-supercapacitor HESS. A better output capacitor voltage reference following performance is achieved with the SM controller mitigating the risk of DC current grid injection when the boost converter leg parameters are non-ideal. DC current grid injection can cause adverse effects such as transformer saturation and power quality issues. The fixed frequency SM controller is able to mitigate the issues associate with the traditional SM controllers for the boost inverters such as high and variable switching frequency operation and component selection complexity. The SM controller considers the non-linear behavior of the boost inverter topology and also simplifies the controller implementation.
- Chapter 6 studies the second-order harmonic ripple component reduction in the DC side current of the battery-supercapacitor HESS. In the initial stages of the HESS, the second-order harmonic ripple current component was allocated to the supercapacitor. However, this ripple current component can adversely affect the lifetime of the supercapacitor [61]. Hence, a suitable second-order harmonic ripple reduction method has to be identified for the HESS. For that, the existing second-order harmonic ripple reduction methods for a single boost inverter are studied. Chapter 6 evaluates performance and limitations of the existing waveform control ripple reduction method. Then, a rule-based control method. The performance of the rule-based controller is validated using experimental results.
- Chapter 7 proposed a novel current feedback ripple reduction method for the boost inverter topology. Compared to the rule-based control method and the waveform control method, the current feedback method is able to achieve a significant second-order harmonic ripple current reduction in the boost inverter DC side current without increasing other harmonic components. Furthermore, the current feedback method is able to mitigate the second-order harmonic current component even in output power transient conditions and hence the method can be applicable to the boost inverter based battery-supercapacitor HESS. Then, a boost inverter based battery-supercapacitor HESS with reduced low frequency input current ripple is

proposed using the current feedback ripple reduction method.

• Chapter 8 summarizes the conclusions of the research and presents a discussion on possible future work.

Chapter 2

Operation of the Boost Inverter

In this chapter, operation of a grid connected boost inverter is discussed. The boost inverter topology provides means of low voltage DC source grid integration by providing both boosting and inversion functions in a single power processing stage. The switching frequency ripple component of the boost inverter with the conventional in-phase PWM operation is extensively analyzed in Subsection 2.3.1. Subsection 2.3.2 presents the proposed phase shifted interleaved operation for the boost inverter topology [64] and the component selection for the proposed interleaved boost inverter is discussed in Subsection 2.3.3. Finally, the input current switching frequency ripple reduction performance of the interleaved boost inverter is presented in Subsection 2.3.4.

2.1 Introduction

Power converter topologies are essential to integrate low voltage DC energy storage devices to single phase AC power grid, and numerous DC/AC power converter topologies have been studied for grid connected energy storage system applications. Depending on the power flow direction, DC/AC power converters can be classified in to two groups namely, unidirectional power converters and bidirectional power converters. Bidirectional power flow is essential in a battery storage system to facilitate the charging and discharging of the battery.

H-bridge inverter is a widely used DC/AC power converter topology in grid applications [70]. The amplitude of the H-bridge inverter output voltage is less than the inverter input voltage and hence, additional DC/DC power converters are required to boost the DC source voltage when low voltage energy storage devices are used. In this type of power conversion systems, two power processing stages can be observed. Further, the output voltage of the H-bridge inverter has to be properly filtered to obtain the required sinusoidal output voltage and current.

Boost inverter is an alternative DC/AC power converter topology which can be used to integrate low voltage DC energy storage systems to a single phase AC line [55]. The boost inverter topology facilitates the DC source AC line integration by providing both the boosting and inversion functions in a single power processing stage. Further, the boost inverter naturally generates a sinusoidal output voltage and hence reduces the output filter requirement.

Numerous boost inverter based energy storage systems were studied in the literature both for stand-alone applications as well as for grid connected applications [54, 55, 57, 71– 79].

A variable frequency sliding mode (SM) controlled stand-alone boost inverter was presented in [55] and in [54], the authors presented a double loop (DL) control strategy for a stand-alone boost inverter. A dynamic SM controller for the stand-alone boost inverter topology was studied in [71]. In [72] and [73] the authors presented grid integrated boost inverter systems using variable frequency SM control method and grid current control method respectively. A waveform control ripple reduction method was studied for the boost inverter topology in [74–77]. Boost inverter based fuel cell-battery energy storage systems were presented in [57, 78]. A modified boost inverter based multi input energy storage system was discussed in [79].

In the boost inverter topology, two inductors are connected to the input DC source. The switching frequency ripple amplitude of the input current depends on the individual ripple components of the inductor currents. As a result of the in-phase PWM operation of the boost inverter, the inductor current switching frequency ripple components are also in-phase and hence the DC source switching frequency ripple current component is greater than each individual inductor current ripple due to addition of the two ripple components.

Interleaving is a widely accepted concept used to reduce the switching frequency ripple component in the DC source current by using parallel power converters [80]. Interleaving concept for the conventional DC/DC boost converter topology has been well documented in literature [80, 81]. In this research, the interleaving concept is applied to the boost inverter topology as a way of reducing the DC source switching frequency ripple current.

In this chapter, the basic operation of the conventional grid connected boost inverter is discussed first. Subsection 2.3.1 provides a detailed analysis on the switching frequency ripple component in the DC source current with the conventional in-Phase PWM operation. The proposed phase shifted interleaved operation for the boost inverter is discussed in Subsection 3.3.2 while the component selection for the interleaved boot inverter is presented in Subsection 2.3.3. Subsection 2.3.4 evaluates the switching frequency ripple current reduction performance of the proposed phase shifted interleaved operation while conclusions of the chapter are summarized in Section 2.4.

2.2 Operation of the Single Phase Grid Connected Boost Inverter

The boost inverter configuration consists of two boost converter legs as shown in Fig. 2.1 [55]. The left hand side boost converter leg is denoted by subscript 1. Each boost converter leg has two power switches, and gate signals for the top switches (subscript T) and the bottom switches (subscript B) of each converter leg are complimentary. The two switches in the each boost converter leg ensure the continuous conduction mode operation and facilitate the bi-directional power flow.

Objective of the boost inverter is to generate a differential sinusoidal output voltage given by,

$$v_{o,ref} = V_o sin(\omega t + \delta) \tag{2.1}$$

where V_o , ω and δ are the inverter output voltage amplitude, frequency and phase angle respectively. In order to obtain the required differential output voltage, the left hand side boost converter legs and the right hand side boost converter legs are required to follow the voltage references $v_{o1,ref}$ and $v_{o2,ref}$ across the output capacitors C_1 and C_2 ;

$$v_{o1,ref} = V_{DC,ref} + \frac{V_o}{2}sin(\omega t + \delta)$$
(2.2)

$$v_{o2,ref} = V_{DC,ref} - \frac{V_o}{2}sin(\omega t + \delta).$$
(2.3)

 $V_{DC,ref}$ is a reference DC shift introduced to ensure the boost mode operation and v_{in} is the DC input voltage to the boost inverter. Hence, $V_{DC,ref} \ge v_{in} + \frac{V_o}{2}$. The main



Fig. 2.1: Single phase grid connected boost inverter

disadvantage of using independent voltage references as in (2.2) and (2.3) is that the inverter output voltage v_o is not directly controlled and in order to alleviate the problem, the voltage across the output capacitor C_2 can be controlled using the reference signal given as [55],

$$v_{o2,ref} = v_{o1} - v_{o,ref}.$$
 (2.4)

The average mode of the left hand side boost converter leg can be illustrated as in Fig.2.2, and the model equations of the boost converter leg can be written as,

$$v_{in} - v_{L1} = (1 - d_1)v_{o1} \tag{2.5}$$

$$i_{C1} + i_{o1} = (1 - d_1)i_{L1} \tag{2.6}$$

where v_{o1} and i_{C1} are the voltage and current of the output capacitor C_1 , and v_{L1} and i_{L1} are the voltage and current of the inductor L_1 , respectively. d_1 is the duty cycle time average values of the switch $Sw_{1,B}$, and i_{o1} is the output current from the left hand side boost converter leg. The inductor and capacitor differential equations can be written as,

$$v_{L1} = R_{L1}i_{L1} + L_1 \frac{di_{L1}}{dt} \tag{2.7}$$

$$i_{C1} + R_{C1}C_1 \frac{di_{C1}}{dt} = C_1 \frac{dv_{o1}}{dt}$$
(2.8)

where R_{L1} and R_{C1} are the internal resistance of the inductor L_1 and capacitor C_1 , respectively.

The grid integration of the boost inverter can be achieved through an interfacing inductor as shown in Fig.2.1. Equivalent circuit of the grid connected boost inverter can be obtained by replacing the inverter and the grid using AC voltage sources as shown in Fig.2.3 In the figure, V_g , V_o , and δ are the grid voltage amplitude, inverter voltage amplitude, and phase angle of the inverter voltage with respect to the phase angle of the grid voltage, respectively. The active power (P) and the reactive power (Q) delivered to the grid can be calculated as,

$$P = \frac{V_g V_o}{2\omega L_g} sin\delta \tag{2.9}$$



Fig. 2.2: Average model of the left hand side boost converter leg.



Fig. 2.3: Equivalent circuit of the grid connected boost inverter

$$Q = \frac{V_g V_o}{2\omega L_g} \cos\delta - \frac{V_g^2}{2\omega L_g}.$$
(2.10)

Note that, the peak voltage and current values are considered for the calculations. Then, for small δ values (2.9) and (2.10) can be approximated as,

$$P = \frac{V_g V_o}{2\omega L_g} \delta \tag{2.11}$$

$$Q = \frac{V_g(V_o - V_g)}{2\omega L_g}.$$
(2.12)

Hence, by proper control of the inverter output voltage amplitude and phase-angle with respect to the grid voltage, the active and reactive power exchange between the grid and the inverter can be achieved.

Fig.2.4 depicts the operation of the boost inverter when the inverter supplies active power to the grid. v_{o1} and v_{o2} waveforms are DC shifted, and the DC shift is selected such that v_{o1} and v_{o2} are always greater than the input voltage v_{in} . The output capacitor voltage AC component amplitude equals to half of the required output voltage amplitude. A second-order harmonic ripple current component can be observed in the boost inverter input current in Fig.2.4 due to the single phase grid integration.

2.3 Switching Frequency Ripple Component of the Boost Inverter DC Side Current

The DC side switching frequency ripple current component of the boost inverter depends on the individual ripple components of the currents through inductors L_1 and L_2 , which are connected to the DC source. In the boost converter operation, the switching frequency ripple component of i_{L1} and i_{L2} are in-phase and hence, the DC source switching frequency ripple component is greater than each individual inductor current ripple due to addition of the two ripple components.



Fig. 2.4: Boost inverter operating waveforms

2.3.1 In-phase PWM Operation

Fig. 2.5 illustrates the gate signals of the switches $Sw_{1,B}$ and $Sw_{2,B}$ with the inductor current and DC source current switching frequency ripple waveforms. In this thesis, PWM is achieved by comparing the duty ratio values with symmetric triangular carrier signals which range between 0 and 1. It can be observed that two gate signals are in-phase (i.e., the centers of the logic high levels of the two gate signals and the centers of the logic low levels of the two gate signals are aligned separately). Duty ratio values of the switches $Sw_{1,B}$ and $Sw_{2,B}$ are d_1 and d_2 , respectively. Inductor currents i_{L1} and i_{L2} can be written as,

$$i_{L1} = I_{L1} + \Delta i_{L1}$$

 $i_{L2} = I_{L2} + \Delta i_{L2}.$
(2.13)

Hence, the total DC source current is,

$$i_{in} = I_{in} + \Delta i_{in} \tag{2.14}$$

where I_{in} , I_{L1} , and I_{L2} are the average values of the DC source current and inductor currents over one switching period, T_s . I_{in} is the sum of two average components I_{L1} and



Fig. 2.5: Ripple current waveforms when the PWM signals for two converter legs are in-phase

 I_{L2} . Δi_{in} is the instantaneous ripple component of the DC source current which is given by $\Delta i_{in} = \Delta i_{L1} + \Delta i_{L2}$.

In this converter structure, d_1 and d_2 are changing over the time. First, $d_1 \ge d_2$ situation is considered as shown in Fig. 2.5. The change of the DC source current from time t_x to t_y can be written as,

$$\Delta i_{in,tx,ty} = i_{in}(t_y) - i_{in}(t_x) \tag{2.15}$$

where, $i_{in}(t_x)$ and $i_{in}(t_y)$ are the instantaneous DC source currents at time t_x and t_y respectively.

From time t_1 to t_2 , $Sw_{1,B}$ is ON and $Sw_{2,B}$ is OFF. Hence, i_{L1} is increasing and i_{L2} is decreasing. The total DC source current can either increase or decrease depending on the amount of increment in i_{L1} and amount of decrement in i_{L2} . Fig. 2.5 shows the waveforms related to the situation where the total DC source current is increasing.

From time t_2 to t_3 , both switches are ON, and therefore, both inductor currents are increasing. Because of that, the total DC source current is also increasing.

 $Sw_{1,B}$ is ON and $Sw_{2,B}$ is OFF from time t_3 to t_4 . Hence, i_{L1} is increasing while i_{L2} is decreasing. In this case also, the total DC source current can either increase or decrease depending on i_{L1} and i_{L2} .

From t_4 to t_5 , both switches are OFF and both inductor currents are decreasing. Hence, the total DC source current is decreasing.

Without a loss of generality, the DC source current switching frequency ripple

magnitude $\Delta i_{in,rip}$ can be written as,

$$\Delta i_{in,rip} = max(|\Delta i_{in,t1,t2}|, |\Delta i_{in,t2,t3}|, |\Delta i_{in,t3,t4}|, |\Delta i_{in,t4,t5}|).$$
(2.16)

It can be observed that,

$$\Delta i_{in,t1,t2} = \Delta i_{in,t3,t4}.\tag{2.17}$$

Moreover,

$$\Delta i_{in,t1,t2} + \Delta i_{in,t2,t3} + \Delta i_{in,t3,t4} + \Delta i_{in,t4,t5} = 0.$$
(2.18)

Because of that, (2.16) can be reduced to

$$\Delta i_{in,rip} = max(\Delta i_{in,t2,t3}, \Delta i_{in,t4,t5}) \tag{2.19}$$

$$\Delta i_{in,t2,t3} = \left(\frac{2v_{in}}{L}\right) d_2 T s \tag{2.20}$$

$$\Delta i_{in,t4,t5} = \left(\frac{(v_{o1} - v_{in})}{L} + \frac{(v_{o2} - v_{in})}{L}\right)(1 - d_1)T_s \tag{2.21}$$

where, $L_1 = L_2 = L$. Since the left-hand side and right-hand side output capacitor voltages are DC shifted sinusoidal waveforms and they are phase shifted, the duty ratio values d_1 and d_2 are changing over the time. Hence, d_1 can be greater than or equal to d_2 or less than d_2 . These two situations should be analyzed separately when calculating the switching frequency ripple amplitudes. Using (2.19), (2.20), and (2.21), a general equation for the DC source switching frequency ripple amplitude when $d_1 \ge d_2$ can be written as,

$$\Delta i_{in,rip} = \begin{cases} \frac{2v_{in}T_s d_2}{L} & \text{if } d_2 \ge 0.5\\ \left| \frac{v_{in}T_s}{L} \left(\frac{d_1 + d_2 - 2d_1 d_2}{1 - d_2} \right) \right| & \text{if } d_2 < 0.5. \end{cases}$$
(2.22)

Similarly, when $d_2 > d_1$,

$$\Delta i_{in,rip} = \begin{cases} \frac{2v_{in}T_s d_1}{L} & \text{if } d_1 \ge 0.5\\ \left| \frac{v_{in}T_s}{L} \left(\frac{d_1 + d_2 - 2d_1 d_2}{1 - d_1} \right) \right| & \text{if } d_1 < 0.5. \end{cases}$$
(2.23)

2.3.2 Phase Shifted Interleaved Operation

Interleaving is a popular concept, which is used to reduce the switching frequency ripple component of power converter input current [80, 81]. In a conventional interleaved boost DC/DC converter configuration, N number of boost converter modules are connected in parallel so that all the converters have the same input voltage and the same output voltage. Moreover, the duty ratio and the switching frequency are same for all the parallel power converters. The PWM signals for the converters are phase shifted by $2\pi/_N$ rad and hence, the inductor current ripple components are also phase shifted, and when summed, the current ripples partially cancel each other, resulting in a lower input current ripple. As a result, the maximum input current ripple equals to $1/_N$ times the inductor current ripple of each boost converter module.

In this research, the interleaving concept is applied to the boost inverter topology to reduce the high frequency ripple component of the DC source current. Before studying the interleaved operation for the boost inverter topology, it is worth to consider the differences between the traditional interleaved boost DC/DC converter operation and the interleaved boost inverter operation.

In the traditional interleaved DC/DC boost converter, the boost converter modules are connected in parallel and hence, all the modules have the same input voltage and the same output voltage, and the modules operate with the same duty ratio value. However, in the boost inverter topology, both left-hand side and right-hand side boost converter legs have the same input voltage v_{in} , but different output capacitor voltages, v_{o1} and v_{o2} . The two boost converter legs operate at different duty ratio values and hence, a detailed analysis on the boost inverter phase-shifted operation is required.

Analysis of the switching frequency ripple component of the interleaved-boost inverter DC source current depends on the summation of two duty ratio components, d_1 and d_2 . Since two boost converter legs are connected to the DC source, relevant PWM signals are phase shifted by 180° from each other. Assuming that $v_{L1,2} = 0$, (i.e. neglecting losses and energy stored in the inductor), the sum of the two duty ratios can be approximated as,

$$d_1 + d_2 = \frac{v_{o1} - v_{in}}{v_{o1}} + \frac{v_{o2} - v_{in}}{v_{o2}}.$$
(2.24)

Case 1: $d_1 + d_2 \ge 1$

Using (2.24), (2.1), (2.2), and (2.3), condition for *Case1* can be written as,

$$d_{1} + d_{2} = \frac{V_{DC} + \frac{V_{o}}{2}sin(\omega t + \delta) - v_{in}}{V_{DC} + \frac{V_{o}}{2}sin(\omega t + \delta)} + \frac{V_{DC} - \frac{V_{o}}{2}sin(\omega t + \delta) - v_{in}}{V_{DC} - \frac{V_{o}}{2}sin(\omega t + \delta)} \ge 1.$$
(2.25)

By simplifying (2.25)

(

$$2V_{DC}^2 - 4v_{in}V_{DC} + \left(\frac{V_o}{2}\right)^2 \cos(2\omega t + \delta) \ge 0.$$
(2.26)

If the DC shift of the output capacitor voltages satisfies (2.27), the total operating region of the boost inverter can be defined by *Case 1*.

$$V_{DC} \ge v_{in} + \sqrt{v_{in}^2 + \left(\frac{V_o}{2}\right)^2}$$
 (2.27)

Fig. 2.6 depicts the gate signals and current waveforms related to the *Case 1*. The center of the logic high region of $Sw_{1,B}$ gate signal is aligned with the center of the logic low region of $Sw_{2,B}$ gate signal and illustrates the 180° phase shift between the two gate signals.

From t_1 to t_2 , both switches $Sw_{1,B}$ and $Sw_{2,B}$ are ON. Hence, both the inductor currents and the DC source current are increasing.

From t_2 to t_3 , $Sw_{1,B}$ is ON while $Sw_{2,B}$ is OFF, resulting in increasing inductor current i_{L1} while decreasing inductor current i_{L2} . However, the resulting DC source current can either increase or decrease depending on i_{L1} and i_{L2} . Fig. 2.6 shows the DC source current increase from t_2 to t_3 .

From t_3 to t_4 , the inductor currents and the DC source current are increasing since both switches are turned ON.

From t_4 to t_5 , i_{L1} is decreasing as $Sw_{1,B}$ is OFF and i_{L2} is increasing as $Sw_{2,B}$ is ON. The total DC source current can either increase or decrease as explained earlier. Equations (2.16)-(2.19) are valid for *Case 1* as well where,

$$\Delta i_{in,t2,t3} = \left(\frac{v_{in}}{L} - \frac{v_{in}}{L} \left(\frac{d_2}{1 - d_2}\right)\right) (1 - d_2) T_s$$
(2.28)

$$\Delta i_{in,t4,t5} = \left(\frac{v_{in}}{L} - \frac{v_{in}}{L} \left(\frac{d_1}{1 - d_1}\right)\right) (1 - d_1) T_s.$$
(2.29)

Using (2.19), (2.28), and (2.29), a general equation for the interleaved operation DC source current switching frequency ripple for the *Case 1* can be written as,

$$\Delta i_{in,rip} = \begin{cases} \frac{v_{in}T_s(2d_2 - 1)}{L} & \text{if } d_2 \ge d_1\\ \frac{v_{in}T_s(2d_1 - 1)}{L} & \text{if } d_2 < d_1. \end{cases}$$
(2.30)



Fig. 2.6: Ripple current waveforms for interleaved operation when $d_1 + d_2 \ge 1$

The maximum switching frequency ripple amplitude of the DC source current for *Case 1*, $max(\Delta i_{in,rip})_{Case1}$ can be obtained as,

$$max(\Delta i_{in,rip})_{Case1} = \frac{v_{in}T_s}{L}(2d_{max} - 1)$$
(2.31)

where d_{max} is the maximum duty ratio of the boost inverter which corresponds to the maximum voltage across the output capacitors.

Case 2: $d_1 + d_2 < 1$

Fig. 2.7 illustrates the relevant gate signals and the current waveforms for the *Case* 2. From t_1 to t_2 , both switches are OFF and hence both inductor currents are decreasing and a decrement in the total DC source current can be observed.

From t_2 to t_3 , $Sw_{1,B}$ is OFF while $Sw_{2,B}$ is ON. Hence, i_{L2} is increasing while i_{L1} is decreasing. The DC source current will increase or decrease depending on the i_{L1} and i_{L2} . Fig. 2.7 illustrates the situation where the DC source current is decreasing.

From t_3 to t_4 , again both switches are OFF and the current waveforms are decreasing.

From t_4 to t_5 , $Sw_{1,B}$ is ON and $Sw_{2,B}$ is OFF. Hence, i_{L1} is increasing while i_{L2} is decreasing. The total DC source current will either increase or decrease as explained earlier.

Similar to Case 2, the analysis provide in equations (2.16)-(2.19) are valid for the Case 2 as well, where,

$$\Delta i_{in,t2,t3} = \left(\frac{v_{in}}{L} - \frac{d_1}{(1-d_1)}\frac{v_{in}}{L}\right) d_2 T_s \tag{2.32}$$

$$\Delta i_{in,t4,t5} = \left(\frac{v_{in}}{L} - \frac{d_2}{(1-d_2)}\frac{v_{in}}{L}\right) d_1 T_s.$$
(2.33)



Fig. 2.7: Ripple current waveforms for interleaved operation when $d_1 + d_2 < 1$

Using, (2.19), (2.32), and (2.33), the DC source switching frequency ripple amplitude with the interleaved operation for the case $d_1 + d_2 < 1$ can be written as,

$$\Delta i_{in,rip} = \begin{cases} \frac{v_{in}T_s}{L} \frac{d_1(1-2d_2)}{(1-d_2)} & \text{if } d_1 \ge d_2\\ \frac{v_{in}T_s}{L} \frac{d_2(1-2d_1)}{(1-d_1)} & \text{if } d_1 < d_2. \end{cases}$$
(2.34)

The maximum switching frequency ripple amplitude of the DC source current for Case 2, $max(\Delta i_{in,rip})_{Case2}$ is given by,

$$max(\Delta i_{in,rip})_{Case2} = \frac{v_{in}T_s}{L} \frac{d_{max}(1 - 2_{dmin})}{(1 - d_{min})}$$
(2.35)

where d_{max} is the maximum duty ratio of the boost inverter which corresponds to the maximum voltage across the output capacitors and d_{min} is the minimum duty ratio of the boost inverter which corresponds to the minimum voltage across the output capacitors.

2.3.3 Boost Inverter Component Selection with the Interleaved Operation

The required inductors for the interleaved boost inverter can be selected to limit the switching frequency ripple amplitude of the DC source current. Once the DC source voltage v_{in} , the inverter output voltage amplitude V_o and the required output capacitor DC shift V_{DC} are determined, the maximum and the minimum duty ratio values of the boost inverter can be calculated as,

$$d_{max} = \frac{v_{o1,max} - v_{in}}{v_{o1,max}} = \frac{V_{DC} + \frac{V_o}{2} - v_{in}}{V_{DC} + \frac{V_o}{2}}$$
(2.36)

$$d_{max} = \frac{v_{o1,min} - v_{in}}{v_{o1,min}} = \frac{V_{DC} - \frac{V_o}{2} - v_{in}}{V_{DC} - \frac{V_o}{2}}.$$
(2.37)

Then, the required minimum inductor value to limit the DC source current switching frequency ripple amplitude to $max(\Delta i_{in,rip})$ can be calculated as,

$$L_{min} = \begin{cases} \frac{v_{in}T_s}{max(\Delta i_{in,rip})} (2d_{max} - 1) & \text{for } Case \ 1\\ \frac{v_{in}T_s}{max(\Delta i_{in,rip})} \frac{d_{max}(1 - 2d_{min})}{(1 - d_{min})} & \text{for } Case \ 2 \ . \end{cases}$$
(2.38)

For both in-phase PWM operation and interleaved operation, the switching frequency ripple amplitudes $\Delta v_{o1,rip}$ and $\Delta v_{o2,rip}$ of the output capacitor voltages v_{o1} and v_{o2} are determined by the output capacitor values. The switching frequency voltage ripple amplitude of v_{o1} is given by,

$$\Delta v_{o1,rip} = \frac{I_{o,max} d_{1,max} T_s}{C_1}.$$
(2.39)

Then using the maximum allowable switching frequency ripple amplitude in the output capacitor voltage, $max(\Delta v_{o1,rip})$, the required minimum output capacitor value $C_{1,min}$ can be calculated as,

$$C_{1,min} = \frac{I_{o,max}T_s}{max(\Delta v_{o1,rip})} \left(\frac{V_{DC} + \frac{V_o}{2} - v_{in}}{V_{DC} + \frac{V_o}{2}}\right).$$
 (2.40)

2.3.4 Comparison Between the In-Phase PWM Operation and the Interleaved Operation

A comparison of the DC source current switching frequency ripple amplitude with in-phase PWM operation and with the interleaved operation is illustrated in Fig. 2.8. Parameters of the considered boost inverter are summarized in Table 2.1. A DL control strategy as explained in [55] is used to control the boost inverter. Fig. 2.8 (a) illustrates the output capacitor voltage waveforms. Figs. 2.8 (b) and 2.8 (c) illustrate the DC source current waveforms with the in-phase PWM operation and with the interleaved operation, respectively. Fig. 2.8 (d) compares the switching frequency ripple current amplitude for the two operating methods. A significant reduction in the switching frequency ripple amplitude can be observed with the interleaved operation. A switching frequency ripple reduction factor R_{sf} is defined as (2.41) to compare the performance of the two configurations.

$$R_{sf} = \frac{\Delta i_{in,rip(In-phasePWM)}}{\Delta i_{in,rip(Interleaved)}}$$
(2.41)

During the interleaved operation, the DC source switching frequency current ripple component is reduced at least 1.6 times compared to the in-phase PWM operation.

2.4 Conclusions

Boost inverter topology enables the integration of low voltage DC sources to a single phase grid by providing both boosting and inversion functions in a single power processing

Parameter	Value	Parameter	Value
V_o	40 V	V_{DC}	40 V
v_{in}	12.8 V	C_1, C_2	60 uF
L_{1}, L_{2}	$1 \mathrm{mH}$	f	$50~\mathrm{Hz}$
T_s	50 us	R_{L1}, R_{L2}	0.24Ω
R_{C1}, R_{C2}	0.01 Ω		

Table 2.1	: Boost	inverter	parameters



Fig. 2.8: Switching frequency ripple current amplitude comparison with the interleaved and in-phase PWM operation. (a) Output capacitor voltage waveforms, (b) DC source current waveform with in-phase PWM operation, (c) DC source current with interleaved operation, (d) DC source current switching frequency ripple amplitude and (e) ripple reduction factor.

stage.

It was shown that the conventional in-phase PWM operation of the boost inverter leads to a higher switching frequency ripple component in the DC source current due to the addition of the ripple components of the two inductor currents. The switching frequency ripple current component increases the internal heating of the DC source and adversely affects its lifetime.

In this chapter, a novel phase-shifted PWM operation for the boost inverter topology was presented. A detailed analysis on the switching frequency ripple current amplitude with the in-phase PWM operation and with the phase-shifted interleaved operation was presented. Furthermore, the component selection for the proposed interleaved boost inverter was discussed. The proposed interleaved operation achieved a significant switching frequency ripple component reduction in the DC source current. The reduction in the switching frequency ripple amplitude helps to reduce the internal heating of the boost inverter DC power source.

Chapter 3

Interleaved Boost Inverter Based Battery-Supercapacitor Hybrid Energy Storage System

This chapter presents the proposed interleaved boost inverter based battery-supercapacitor HESS [64]. The interleaved operation reduces the switching frequency component of the battery and the supercapacitor currents and helps to reduce the internal heating of both the battery and the supercapacitor. Section 3.3 presents the proposed HESS control system which allocates the fast power variations to the supercapacitor while the battery supplies the slow varying power component. A high pass filter based power allocation method is used where the second-order harmonic ripple current component is allocated to the supercapacitor. A supercapacitor voltage controller (SCVC) is employed to reduce the risk of supercapacitor over-charging and over-discharging by maintaining its voltage around a reference voltage level. An Extended Kalman filter based battery SOC estimator is designed for the selected battery module. The proposed system is able to satisfy the output power requirements while operating within the battery and the supercapacitor SOC limits. The proposed battery-supercapacitor HESS is able to increase the battery lifetime by reducing the battery current fluctuations and peak currents. The proposed boost inverter based HESS is the first experimentally verified single phase direct grid integrated battery-supercapacitor HESS available in the literature.

3.1 Introduction

Boost inverter topology provides means of DC source grid integration by providing both boosting and inversion functions in a single power processing stage [55]. Chapter 2 presented the basic operating principles of a grid connected boost inverter system. Furthermore, a detailed analysis on the in-phase PWM operation of the boost inverter was provided and an interleaved operation for the boost inverter topology was proposed to reduce the DC source switching frequency current component.

In this chapter, an interleaved boost inverter based single phase grid connected battery-supercapacitor HESS is proposed [64]. The interleaved operation helps to reduce the switching frequency ripple component of both the battery and the supercapacitor currents. None of existing battery-supercapacitor HESSs considered the switching frequency ripple reduction in the DC source currents. The proposed controller for the HESS is presented in Section 3.3. The controller contains two double loop (DL) controllers to control the left hand side and the right hand side boost converter legs. A high pass filter based power allocation method is employed to allocate the fast power fluctuations to the supercapacitor. The high pass filter allocates the second-order harmonic current component to the supercapacitor since such ripple current component adversely affect the battery lifetime due to internal heating [13] Since it is important to avoid supercapacitor over-charging and over-discharging, a SCVC is used. An active and reactive power controller is implemented using the droop control theory to control the power exchange between the grid and the HESS as presented in Subsection 3.3.2. A second order nonlinear RC battery model used to model the LiFePO₄ battery and an extended Kalman filter based battery SOC estimation method are described in Section 3.4. A battery SOC controller is implemented to ensure the operation of the battery within a safe operating region. The operation of the interleaved boost inverter based battery-supercapacitor HESS is validated using experimental results obtained from a laboratory HESS prototype in Section 3.5.

3.2 Boost Inverter Based Battery- Supercapacitor Hybrid Energy Storage System

The system configuration of the boost inverter based single phase grid connected battery-supercapacitor HESS is shown in Fig. 3.1. The configuration contains two boost inverters one for the battery and one for the supercapacitor. Each boost inverter has two boost converter legs; namely the left hand side boost converter leg (subscript 1) and the right band side boost converter leg (subscript 2).





Dynamic behavior of the boost converter legs can be described by the boost converter average model. The model equations for the left-hand-side boost converter legs can be expressed as follows:

$$v_{sc} - v_{Lsc1} = (1 - d_{sc1})v_{o1} \tag{3.1}$$

$$v_{batt} - v_{Lbatt1} = (1 - d_{batt1})v_{o1}$$
(3.2)

$$i_{C1} + i_{o1} = (1 - d_{batt1})i_{Lbatt1} + (1 - d_{sc1})i_{Lsc1}$$
(3.3)

where v_{o1} and i_{C1} are the voltage and current of the output capacitor C_1 , and v_{Lbatt1} and i_{Lbatt1} are the voltage and current of the inductor L_{batt1} , respectively. v_{Lsc1} and i_{Lsc1} are the voltage and the current of the inductor L_{sc1} , and v_{batt} and v_{sc} are the battery voltage and the supercapacitor voltage, respectively. Duty cycle time average values of the gate signals for the switches $Sw_{batt1,B}$ and $Sw_{sc1,B}$ are denoted as d_{batt1} and d_{sc1} , respectively. i_{o1} is the output current from the left band side boost converter legs. The inductor and capacitor differential equations can be written as,

$$L_{batt1}\frac{di_{Lbatt1}}{dt} = v_{Lbatt1} - R_{Lbatt1}i_{Lbatt1}$$
(3.4)

$$L_{sc1}\frac{di_{Lsc1}}{dt} = v_{Lsc1} - R_{Lsc1}i_{Lsc1}$$
(3.5)

$$C_1 \frac{dv_{o1}}{dt} = i_{C1} + R_{C1} C_1 \frac{di_{C1}}{dt}$$
(3.6)

where R_{C1} , R_{Lbatt1} , and R_{Lsc1} are the ESR values of the capacitor C_1 and the inductors L_{batt1} and L_{sc1} , respectively.

3.3 Control System Description

3.3.1 Power Converter Control Strategy

The overall block diagram of the HESS controller is illustrated in Fig.3.2. The objective of the HESS controller is to maintain the inverter output voltage such that the HESS fulfills the active and reactive power requirements. Additionally, the controller has to allocate the second-order harmonic ripple component and fast power fluctuations to the supercapacitor. Further, the controller has to maintain the SOC of the battery and the supercapacitor within a safe operating region to ensure safe and sustainable operation of the HESS.

The HESS output voltage has to follow a sinusoidal reference voltage,

$$v_{o,ref} = V_o \sin(\omega t + \delta) \tag{3.7}$$



Fig. 3.2: Overall block diagram of the proposed controller.

where the HESS output voltage amplitude, frequency, and phase angle with respect to the grid voltage are denoted as V_o , ω , and δ , respectively. In order to obtain the required HESS differential output voltage, the left hand side and the right hand side boost converter legs are required to follow voltage references $v_{o1,ref}$ and $v_{o2,ref}$ across the output capacitors C_1 and C_2 .

$$v_{o1,ref} = V_{DC,ref} + \frac{V_o}{2}\sin(\omega t + \delta)$$
(3.8)

$$v_{o2,ref} = V_{DC,ref} - \frac{V_o}{2}\sin(\omega t + \delta)$$
(3.9)

The DC shift of the output capacitor reference voltages $v_{o1,ref}$ and $v_{o2,ref}$ is selected as,

$$V_{DC,ref} \ge v_{in} + \frac{V_o}{2} \tag{3.10}$$

where v_{in} is the input voltage to the boost converter legs (v_{batt} or v_{sc}). In order to achieve direct control over the HESS output voltage v_o , the voltage v_{o2} across the capacitor C_2 can be controlled using (3.11) as mentioned in Chapter 2.

$$v_{o2,ref} = v_{o1} - v_{o,ref} \tag{3.11}$$

Two DL controllers are used to control the left hand side and right hand side boost converter legs. Each DL controller has two inner current control loops and an outer voltage control loop. The objective of the voltage control loop for the left hand side boost converter legs is to generate such inductor current reference signals for the current control loops that, 1) maintain the reference voltage v_{o1} , 2) allocate the high frequency current variations to the supercapacitor, 3) maintain the supercapacitor voltage around a pre-defined reference voltage, and 4) supply the initial supercapacitor charging current from the battery. The block diagram of the outer voltage control loop and the inner current control loops for the left hand side boost converter legs is illustrated in Fig. 3.3 where,

$$H_{Lsc1}(s) = \frac{1}{(R_{Lsc1} + L_{sc1}s)}$$
(3.12)

$$H_{Lbatt1}(s) = \frac{1}{\left(R_{Lbatt1} + L_{batt1}s\right)} \tag{3.13}$$

$$H_{C1}(s) = \frac{1 + R_{C1}C_1s}{C_1s} \tag{3.14}$$

The voltage and current measurements are filtered using low pass filters as shown in Fig. 3.3 where T_F is filter parameter.

A proportional resonant (PR) controller with a transfer function $H_{PR}(s)$ is used in the voltage control loop to achieve better sinusoidal tracking performance.

$$H_{PR}(s) = K_{p,PR} + \frac{2sK_{i,PR}}{s^2 + \omega^2}$$
(3.15)

where $K_{p,PR}$ and $K_{i,PR}$ are constants and ω is the frequency of the output voltage in rad/s.





The total instantaneous power requirement from the left hand side boost converter legs is given by P_{tot1} . A high pass filter with a cut off frequency ω_{HPF} is employed to divert the second-order harmonic ripple component and fast power fluctuations to the supercapacitor while battery supplies the slow varying power requirement.

The supercapacitor tends to fully discharge or fully charge during the operation of the HESS, if the supercapacitor voltage is not maintained around a reference voltage value [29]. Hence, a PI controller is implemented to maintain the supercapacitor voltage around a pre-defined reference voltage $v_{sc,ref}$. Additionally, the controller compensates for the supercapacitor self-discharging. By selecting the supercapacitor operating voltage range between 50% and 100% of its rated voltage $v_{sc,max}$, 75% of the overall supercapacitor stored energy can be utilized [19]. Then the supercapacitor reference voltage $v_{sc,ref}$ is selected such that equal amount of energy is available above and below the reference level for future power delivering or absorbing situations [18].

A constant saturation block Sat_{sc} (see Fig. 3.3) based ripple current controller is used to allocate the ripple current to the battery, when the supercapacitor voltage is less than a pre-defined voltage $v_{sc,min}$. The constant saturation block saturates at the voltage $v_{sc,min}$ and if the supercapacitor voltage is less than $v_{sc,min}$, the ripple current controller allocates a fraction of the ripple current to the battery. In the initial supercapacitor charging process, the total required current and the supercapacitor charging current are supplied by the battery and as the supercapacitor gets charged, the ripple current is gradually allocated to the supercapacitor.

The PR controller for the outer voltage control loop is designed to achieve 400 Hz bandwidth and the PI controllers for the inner current control loops are implemented with 4 kHz bandwidth. These values enable the accurate tracking of 50 Hz reference voltages. The Bode diagram of the outer voltage control loop is shown in Fig. 3.4 (a). The Bode diagrams of the inner current control loops for the left band side battery connected boost converter leg and for the left band side supercapacitor connected boost converter leg are illustrated in Figs. 3.4 (b) and 3.4 (c), respectively.

The SCVC has to suppress the low frequency voltage errors and at the same time it has to be insensitive to high frequency errors so that it will not affect the smoothness of the battery current. For this reason, SCVC is designed to have 0.01Hz crossover frequency as shown in Fig. 3.4 (d).

3.3.2 Hybrid Energy Storage System Grid Integration

The HESS is connected to the grid through an interfacing inductor as shown in Fig. 3.1. The active power and the reactive power delivered from the HESS to the grid are denoted



Fig. 3.4: Bode diagrams of the HESS controller. (a) Open loop Bode plot of the left hand side voltage control loop, (b) open loop Bode plot of the current control loop for the left hand side battery connected boost converter leg, (c) open loop Bode plot of the current control loop for the left hand side supercapacitor connected boost converter leg and (d) open loop Bode plot of the SCVC.

as P_{HESS} and Q_{HESS} . For small δ values, P_{HESS} and Q_{HESS} can be approximated as,

$$P_{HESS} \approx \frac{V_g V_o}{2\omega L_g} \delta \tag{3.16}$$

$$Q_{HESS} \approx \frac{V_g(V_o - V_g)}{2\omega L_q} \tag{3.17}$$

Hence, by proper control of the inverter output voltage amplitude and phase-angle with respect to the grid voltage, the active and reactive power exchange between the grid and the HESS can be achieved. Furthermore, the fast and accurate measurement of the active and reactive power is essential to achieve an accurate power exchange between the grid and the HESS. P_{HESS} and Q_{HESS} can be calculated using orthogonal components of the grid voltage and current at the point of common coupling (PCC) as [78],

$$P_{HESS} = \frac{1}{2} (v_{g,\alpha} i_{g,\alpha} + v_{b,\beta} i_{g,\beta})$$
(3.18)

$$Q_{HESS} = \frac{1}{2} (v_{g,\beta} i_{g,\alpha} - v_{g,\alpha} i_{g,\beta})$$
(3.19)

where $v_{g,\alpha}$, $v_{g,\beta}$, and $i_{g,\alpha}$, $i_{g,\beta}$ are the orthogonal components of the instantaneous grid voltage and current at the PCC, respectively. The orthogonal components are generated using a second order generalized integrator based quadrature signal generator (SOGI-QSG) [82]. Two PI controllers are designed based on (3.16) and (3.17) to control the active and reactive power delivered to the grid by controlling the inverter output voltage amplitude and phase angle with respect to the grid voltage. The block diagram of the active and reactive power PI controllers along with the battery SOC controller, which will be described in Section 3.4 are illustrated in Fig.3.5. The PI controllers are designed to achieve rise time less than 0.1s for better reference following.

3.4 Battery State of Charge Estimation and Control

3.4.1 Battery Model

A Lithium iron phosphate (LiFePO₄) battery by K2 Energy with 6.4Ah capacity and 12.8V rated voltage is used in the experimental setup. A suitable battery model for the selected battery should be identified to design a battery SOC estimation algorithm. Various types of battery models have been studied in the literature depending on the battery chemistry. A second-order RC battery model shown in Fig. 3.6, which is proven to be a suitable battery model for LiFePO₄ batteries is selected for the SOC estimation algorithm [83]. The effects of temperature on battery SOC estimation are not considered



Fig. 3.5: Active and reactive power controller (PQ controller), and battery SOC controller



Fig. 3.6: Second order RC battery model

in this thesis. The battery open circuit voltage (OCV), E_o is a nonlinear function of the battery SOC, SOC_{batt} and can be written as,

$$E_o = f(SOC_{batt}). \tag{3.20}$$

From the RC battery model, following equations can be written.

$$v_{batt} = E_o - v_{1,batt} - v_{2,batt} - i_{batt} R_{0,batt}$$
(3.21)

$$i_{batt} = \frac{v_{1,batt}}{R_{1,batt}} + C_{1,batt} \frac{dv_{1,batt}}{dt} = \frac{v_{2,batt}}{R_{2,batt}} + C_{2,batt} \frac{dv_{2,batt}}{dt}$$
(3.22)

where $R_{0,batt}$, $R_{1,batt}$, $R_{2,batt}$, $C_{1,batt}$, and $C_{2,batt}$ are the battery internal parameters which reflect the battery dynamic response. $v_{1,batt}$ and $v_{2,batt}$ are the voltages across the capacitors $C_{1,batt}$ and $C_{2,batt}$, respectively. Then the state space equations for the considered battery model can be written as (23) by selecting the capacitor voltage drops and the battery SOC as the state variables.

$$\dot{x} = \begin{bmatrix} \frac{-1}{R_{1,batt}C_{1,batt}} & 0 & 0\\ 0 & \frac{-1}{R_{2,batt}C_{2,batt}} & 0\\ 0 & 0 & 0 \end{bmatrix} x \times \begin{bmatrix} \frac{1}{C_{1,batt}} \\ \frac{1}{C_{2,batt}} \\ -\eta \end{bmatrix} i_{batt}$$
(3.23)

where $x = \begin{bmatrix} v_{1,batt} & v_{2,batt} & SOC_{batt} \end{bmatrix}^T$, $\eta = \begin{pmatrix} \frac{1}{3600C_{batt}} \end{pmatrix}$, and C_{batt} is the total battery capacity in Ah. A discretized state space model of the RC battery model can be found as,

$$x_k = Ax_{k-1} + Bi_{batt,k-1} \tag{3.24}$$

where
$$A = \begin{bmatrix} \frac{-\Delta t}{R_{1,batt}C_{1,batt}} + 1 & 0 & 0\\ 0 & \frac{-\Delta t}{R_{2,batt}C_{2,batt}} + 1 & 0\\ 0 & 0 & 1 \end{bmatrix}$$
, $B = \begin{bmatrix} \frac{\Delta t}{C_{1,batt}} \\ \frac{\Delta t}{C_{2,batt}} \\ -\eta\Delta t \end{bmatrix}$, and Δt is the sampling interval

sampling interval.

3.4.2 Battery Model Parameter Estimation

In this section, the relationship between the battery OCV and the battery SOC, and the battery RC model parameter estimation are presented.

The relationship between the battery OCV and the battery SOC is identified by conducting a battery charge and discharge test as explained in [83]. A 3A, 14.4V LiFePO₄ battery charger from Fuyuan and a PLZ1004WH Kikusui DC electronic load are used to charge and discharge the LiFePO₄ battery. For the charge test, the battery is charged from 0% SOC to 100% SOC using C/2 charge pulses which last for 5min, followed by a rest period of 45 min which allows the battery to reach the charge equilibrium condition

before applying the next cycle. The battery terminal voltage is measured after each rest period. In the discharge test, the battery is discharged from 100% SOC to 0% SOC by using 1C discharge pulses which last for 5min followed by a 45min rest period. After each rest period, the battery terminal voltage is measured. It is assumed that the measured battery terminal voltage values can be considered as the battery OCV at each time instant since the battery is in the charge equilibrium condition [83]. The measured voltage values at different SOC values are plotted in Fig. 3.7. It can be observed that the charging OCV curve is different from the discharging OCV curve. Hence, the average OCV values are considered, and a monotonically increasing approximated OCV curve as shown in Fig. 3.7 is used for future calculations. The approximated OCV curve can be represented by a seventh order polynomial equation as

$$E_{o} = a_{1}SOC_{batt}^{7} + a_{2}SOC_{batt}^{6} + a_{3}SOC_{batt}^{5} + a_{4}SOC_{batt}^{4} + a_{5}SOC_{batt}^{3} + a_{6}SOC_{batt}^{2} + a_{7}SOC_{batt}^{1} + a_{8} \quad (3.25)$$

The curve parameters can be obtained by curve fitting using least square method as a_1 =-41.263, a_2 =203.702, a_3 =-393.99, a_4 =384.57, a_5 =-200.36, a_6 =52.823, a_7 =-5.063, and a_8 =12.99. The battery model developed using the approximated curve is able to estimate the battery SOC accurately as demonstrated later in the chapter. The battery RC model parameters are identified by conducting a battery discharge test as illustrated in Fig. 3.8. Each discharge cycle consists of 5min discharge phase and 45 min rest phase. A least square estimation technique is employed to obtain the RC model parameters as described in [83, 84] and the obtained RC model parameters are summarized in Table 3.1.



Fig. 3.7: Relationship between the battery OCV and SOC



Fig. 3.8: Battery current and voltage profile for the battery RC model parameter estimation

Table 3.1: Estimated battery RC model parameters

Parameter	Estimated Value	
$R_{0,batt}$	$0.1028~\Omega$	
$R_{1,batt}$	$0.061 \ \Omega$	
$R_{2,batt}$	$0.008 \ \Omega$	
$C_{1,batt}$	1772 F	
$C_{2,batt}$	52.2 F	

3.4.3 Extended Kalman Filter Based Battery SOC Estimation

Battery SOC indicates the available battery capacity as a ratio of the total battery capacity and it can be defined as,

$$SOC_{batt} = SOC_{batt,init} - \frac{\int_0^t i_{batt}(\tau) d\tau}{3600C_{batt}}$$
(3.26)

where $SOC_{batt,init}$ and C_{batt} are the initial battery SOC and the capacity of the battery in Ah. Knowledge of the battery SOC is important in energy storage system operation to prevent possible battery overcharging which can lead to hazardous situations or over discharging condition which can lead to battery degradation. However, battery SOC calculation using (3.26) is not practical due to possible error and noise accumulation with the integration of the battery current. [85]. Moreover, (3.26) requires the initial battery SOC which may not be available for the SOC calculation. Hence, a reliable battery SOC estimation algorithm is vital to ensure proper and sustainable operation of the HESS.

In this research, an extended Kalman filter based battery SOC estimation method is used to accurately estimate the SOC of the battery. The battery current i_{batt} is taken as the input and the measured battery voltage v_{batt} is taken as the measured output of the Kalman filter. Then, the battery state space model can be re-written as,

$$x_{k+1} = Ax_k + Bi_{batt,k} + w_k$$

$$y_k = v_{batt,k} = E_{o,k} - v_{1,batt,k} - v_{2,batt,k} - i_{batt,k} R_{0,batt} + v_k$$

$$y_k = f(SOC_{batt}) - v_{1,batt,k} - v_{2,batt,k} - i_{batt,k} R_{0,batt} + v_k = g(x_k, i_{batt,k}) + v_k$$
(3.27)

where $g(x_k, i_{batt,k})$ is a nonlinear function. w_k and v_k are zero-mean white Gaussian processes with covariance matrices Q_{Kalman} and R_{Kalman} , respectively. To design an extended Kalman filter for the system in (3.27), a measurement matrix C_k is defined as (3.28) to linearize the nonlinear system model around the last state estimate [83,85].

$$C_k = \left. \frac{\partial g(x_k, i_{batt,k})}{\partial x_k} \right|_{x = \hat{x}_k^-} \tag{3.28}$$

Using (3.27), (3.28) can be written as [83],

$$C_{k} = \begin{bmatrix} -1 & -1 & \frac{\partial f(SOC_{batt,k})}{\partial SOC_{batt,k}} \Big|_{SOC_{batt,k} = S\hat{O}C_{batt,k}^{-}} \end{bmatrix}$$
(3.29)

where superscript ' \wedge ' denotes the estimated states. The Kalman filter generates an error covariance matrix $P_{Kalman} = E\left[\tilde{x}_k \tilde{x}_k^T\right]$ where $\tilde{x}_k = x_k - \hat{x}_k$ and E[.] is the statistical expectation operator. The error covariance matrix indicates the uncertainty of the estimated states. In each time step, Kalman filter produces two set of estimations for the states as well as for the error covariance matrix. The first estimation is computed before any system measurements are made and the estimate is denoted by superscript '-'. After obtaining the system measurements, Kalman filter produces updated state estimation which is denoted by superscript '+'

The extended Kalman filter is initiated by providing information about the system states and the error covariance matrix. In each time interval, the extended Kalman filter performs two steps of estimation. In the first step, the extended Kalman filter generates an expected state value at the time instant k as,

$$\hat{x}_{k}^{-} = A\hat{x}_{k-1}^{+} + Bi_{batt,k-1} \tag{3.30}$$

This prediction step is known as the 'time update'. Then, the error covariance is calculated as,

$$P_{Kalman,k}^{-} = AP_{Kalman,k-1}^{+}A^{T} + Q_{Kalman}$$

$$(3.31)$$

The Kalman filter gain K_{Kalman} is calculated as,

$$K_{Kalman,k} = P_{Kalman,k}^{-} C_{k}^{T} [C_{k} P_{k}^{-} C_{k}^{T} + R_{Kalman}]^{-1}$$
(3.32)

Following the system measurements, the state estimation is updated as,

$$\hat{x}_{k}^{+} = \hat{x}_{k}^{-} + K_{Kalman,k}[y_{k} - g(\hat{x}_{k}^{-}, u_{k})]$$
(3.33)

Finally, the error covariance matrix is updated as,

$$P_{Kalman,k}^{+} = (I - K_{Kalman,k}C_k)P_{Kalman,k}^{-}$$
(3.34)

where I is the identity matrix.

Using the above calculations, the SOC of the battery can be estimated accurately. In the designed extended Kalman filter, initial parameters are selected as follows. The measurement covariance R_{Kalman} is tuned to 1000 in order to cope with measurement errors and noises. A diagonal process covariance matrix Q_{Kalman} is selected assuming uncorrelated noises, disturbances and model imperfections. Initially, all the diagonal elements were set to a constant value and elements of the matrix are tuned by obtaining the estimated states using a real-time system. After some tuning, the process variance matrix Q_{Kalman} is selected as,

$$Q_{Kalman} = \begin{bmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 10^{-3} \end{bmatrix}$$
(3.35)

A pulse discharge test is carried out to verify the performance of the extended Kalman filter based battery SOC estimation system. Fig. 3.9 compares the estimated SOC using the extended Kalman filter $(SOC_{batt,K})$ with the SOC calculated using the current integration method $(SOC_{batt,i})$.

3.4.4 Battery SOC Controller

The battery SOC has to be maintained within a safe operating region in order to eliminate possible negative impacts on the battery. The battery SOC controller illustrated in Fig. 3.5 is employed to maintain the battery operation in a safe operating region. If the battery SOC estimated using the extended Kalman filter is less than or equal to a minimum allowable level $SOC_{batt,min}$ and if the battery is continuing to discharge due to a positive or zero reference power signal, the actual reference power signal $P_{HESS,ref}$ is ignored. Then, a modified active power reference $P_{HESS,ref,mod}$ is generated using a PI controller as shown in Fig. 3.5, which maintains the total battery current at zero


Fig. 3.9: Estimated battery SOC using the extended Kalman filter $(SOC_{batt,K})$ and the current integration method $(SOC_{batt,i})$.

avoiding the battery discharge. Similarly, if the battery SOC is greater than or equal to a maximum SOC level $SOC_{batt,max}$ and if the battery is continuing to charge due to a negative reference power signal, the SOC controller ignores the actual reference power signal and maintains the battery current at zero.

3.5 Experimental Results

The performance of the proposed interleaved boost inverter based battery-supercapacitor HESS is verified using the experimental prototype illustrated in Fig. 3.10. A 6.4Ah, 12.8V, 81.92Wh LiFePO₄ battery from K2 Energy and a 16V, 58F, 7424J Maxwell supercapacitor module are used in the prototype, and the parameters of the HESS prototype are summarized in Table 3.2. The power electronics converter is implemented using Semikron SEMiX 302GB066HD IGBT modules. LEM LV20 voltage sensors and LEM LA 55-P current sensors are used in the experimental setup. The HESS control system is implemented on a DSpace DS1006 system using Matlab Simulink and the control system parameters are summarized in Table 3.2. Since the experimental setup is a low voltage prototype, the grid integration of the HESS is achieved through a step-up transformer.

Fig. 3.11 depicts the HESS output voltage, grid voltage and current at the PCC when the HESS delivers 30W active power to the grid. It can be observed the active and reactive power control is achieved by modifying the HESS output voltage amplitude and

phase angle with respect to the grid voltage.

Fig. 3.12 illustrates the inductor current waveforms i_{Lbatt1} and i_{Lbatt2} when the HESS supplies 30W active power to the grid.



Fig. 3.10: HESS experimental setup



Fig. 3.11: Current waveform of the grid interfacing inductor, i_g (CH2), HESS output voltage, v_o (CH3), and grid voltage, v_g (CH4), when the HESS with interleaved operation delivers 30W active power to the grid.



Fig. 3.12: Inductor current waveforms i_{Lsc1} (CH1) and i_{Lbatt1} (CH2) when the HESS with interleaved operation delivers 30W active power to the grid.

Furthermore, the total supercapacitor current and the battery current are shown in Fig. 3.13 with the interleaved operation. It is evident that the HESS controller allocates the second-order harmonic current component to the supercapacitor while battery supplies the required average power component.

Total supercapacitor current and the total battery current with the in-phase PWM operation at 30W active power output is shown in Fig. 3.14. When compared with the interleaved operation illustrated in Fig. 3.13, a significant reduction in the switching frequency ripple component of the battery and the supercapacitor current can be observed with the interleaved operation of the boost inverter based HESS.

Operation of the HESS with a step changing power reference signal is shown in Fig. 3.15. The high pass filter cut-off frequency is set to 0.05rads^{-1} in this experiment. It is evident that the proposed control system is able to properly track the output power reference signals. The supercapacitor responds to the sudden power changes while the battery current gradually increases to supply the required output power. Due to the initial supercapacitor response to the step power change, its voltage diverts from the reference value. The battery supplies an additional amount of current to recharge the supercapacitor to the reference voltage level. After supplying the required charging current to the supercapacitor, the battery current gradually reduces and settles down to provide the required average output power component. It can be observed that throughout the



Fig. 3.13: The total battery current, i_{batt} (CH2) and the total supercapacitor current, i_{sc} (CH1) when the HESS with interleaved operation delivers 30W active power to the grid.



Fig. 3.14: The total battery current, i_{batt} (CH2) and the total supercapacitor current, i_{sc} (CH1) when the HESS with in-phase PWM operation delivers 30W active power to the grid.

Parameter	Value	Parameter	Value					
Power converter parameters								
V_o	40 V	$V_{DC,ref}$	40 V					
L_{batt1}, L_{batt2}	$1 \mathrm{mH}$	L_{sc1}, L_{sc2}	$210~\mu\mathrm{H}$					
C_1, C_2	$60 \ \mu F$	f	$50 \mathrm{~Hz}$					
T_s	$50~\mu s$	T_F	0.001					
$V_{sc,ref}$	11.3 V	R_{Lsc1}, R_{Lsc2}	0.24Ω					
R_{Lbatt1}, R_{Lbatt2}	0.24Ω							
Grid parameters								
V_g	40 V	L_g	$20 \mathrm{mH}$					
Voltage control loop parameters								
$K_p(PR)$	0.1	$K_p(PI_{scv})$	50					
$K_i(PR)$	4	$K_i(PI_{scv})$	1					
Current control loop parameters								
$K_p(PI_{Ibatt})$	15.529	$K_p(PI_{Isc})$	2.615					
$K_i(PI_{Ibatt})$	8.44×10^{-5}	$K_i(PI_{Isc})$	8.44×10^{-5}					
PQ controller parameters								
$K_p(PI_P)$	2×10^{-6}	$K_p(PI_Q)$	1×10^{-5}					
$K_i(PI_P)$	0.14	$K_i(PI_Q)$	4					
Battery SOC controller parameters								
$K_p(PI_{SOC})$	0.8	$K_i(PI_{SOC})$	0.5					

Table 3.2: Hybrid energy storage system parameters

experiment, the supercapacitor voltage is maintained around the reference voltage level. In the supercapacitor current waveforms, the second-order harmonic current component is filtered out so that the supercapacitor response to the fast power changes can be clearly seen.

Next, operation of the HESS in a renewable power generation scenario as illustrated in Fig. 3.16 is considered and the power reference signals are generated to emulate the PV power generation. The PV power generation data was obtained from [86] and scaled down to fit the power range of the low voltage low power prototype. The cut-off frequency of the high-pass filter is set to 3.3×10^{-3} rads⁻¹ so that the battery responds to the average power requirement while the supercapacitor responds to the fast PV power changes. Fig. 3.17 illustrates the experimental waveforms and as expected, the battery is charged using the average power component generated by the PV source while the supercapacitor handles



Fig. 3.15: Operation of the HESS with step changing output power reference signals.



Fig. 3.16: Example application of a direct AC line integrated battery-supercapacitor HESS with a renewable power generation system.

the fluctuating current component due to the intermittency of the PV power generation.

Fig. 3.18 demonstrates the operation of the battery-supercapacitor HESS when the



Fig. 3.17: Operation of the HESS when the power reference signals are generated to emulate PV power generation situation.

battery SOC limits are reached. For the illustrative purposes, the maximum battery SOC limit, $SOC_{batt,max}$ and the minimum battery SOC limit $SOC_{batt,min}$ are selected as 0.7 and 0.65 respectively. In the experiment, initially HESS absorbs active power from the grid and hence the battery gets charged. Once the battery reaches its maximum SOC limit, the HESS reference power signal is modified by the SOC controller to bring the battery current to zero so that the battery SOC will not increase over the maximum limit. Next, active power is supplied by the HESS to the grid in order to illustrate the operation of the HESS at the minimum battery SOC limit. When the battery SOC reaches its minimum value, the active power reference signal is modified by the SOC controller to avoid further discharging of the battery as illustrate in Fig. 3.18.

3.6 Conclusions

A single phase grid connected battery-supercapacitor HESS was proposed in this chapter. The HESS was designed using the interleaved boost inverter topology proposed



Fig. 3.18: Operation of the HESS when the battery SOC limits are reached.

in Chapter 2 which reduces the switching frequency ripple current component in both the battery and the supercapacitor currents and helps to reduce the internal heating of the DC energy storage devices. A filter based power allocation method and a DL control strategy were used to control the HESS. A SCVC was employed to reduce the risk of supercapacitor over-charging and over-discharging by maintaining the supercapacitor voltage around a reference voltage level. Furthermore, an extended Kalman filter based battery SOC estimator was employed to maintain the battery SOC within a safe operating region.

Experimental results showed that the HESS operated as required and allocated the fast current variations and the second-order harmonic current to the supercapacitor while the battery responded to the slow varying current demands. This in turn increases the battery lifetime due to reduction in its internal heating. Moreover, the results demonstrated that the control system maintained the supercapacitor voltage at around a reference voltage value and additionally, the battery SOC was maintained within the specified SOC limits. The proposed interleaved HESS is the first experimentally verified single phase direct grid integrated HESS with reduced switching frequency ripple current component in literature.

Chapter 4

Supercapacitor Energy Controlled Filter Based Battery-Supercapacitor Hybrid Energy Storage System

The power allocation parameter selection and energy storage element sizing are important aspects of a battery-supercapacitor HESS design. Furthermore, it is vital to control the SOC of the supercapacitor to avoid supercapacitor over-charging and over-discharging. A SCVC and a high-pass filter based power allocation strategy was used in the proposed HESS in Chapter 3. Due to the non-linearity of the SCVC, it is challenging to accurately size the energy storage elements as well as to identify the suitable high-pass filter and SCVC parameters. In this chapter, a supercapacitor energy controller (SCEC) is proposed which helps to circumvent the issues associated with the non-linearity of the SCVC and filter based power allocation method [65]. The effects of the SCEC on the filter based HESS is extensively analyzed. Then, a HESS power allocation parameter selection method and an energy storage element sizing method are proposed for a given application. Experimental results are used to validate the operation, parameter selection and sizing of the supercapacitor energy controlled filter based HESS.

4.1 Introduction

The operation of an interleaved boost inverter based, single phase grid integrated battery-supercapacitor HESS was proposed in Chapter 3. A high pass filter based power allocation strategy along with a supercapacitor voltage controller (SCVC) was used to allocate the fast power variations and the second-order harmonic current component to the supercapacitor. A DL controller strategy was used to control the boost converter legs and a droop control method was used to control the HESS power exchange with the grid. The battery SOC estimation obtained from an extended Kalman filter based SOC estimator was used to control the battery SOC within a predefined safe SOC range.

In this chapter, energy storage element sizing and HESS power allocation parameter selection for a supercapacitor SOC controlled filter based batter-supercapacitor HESS is considered.

Energy storage sizing is an important aspect of the HESS design. Numerous researches have been conducted to identify suitable energy storage sizing methods for battery-supercapacitor HESSs and the sizing methods are strongly coupled with the employed power allocation methods [10, 24, 26, 29, 31, 34, 37, 38, 49, 50]. The energy storage sizing methods presented in [10, 24, 29, 31, 34, 38, 50] were studied using non-filter based power allocation methods and hence cannot be used in filter based battery-supercapacitor HESSs. In [34], authors presented a HESS cost minimization method to select the power allocation frequency and the supercapacitor size. In [37, 49] the filter based HESS design was done based on the Ragone theory. A supercapacitor sizing method for a filter based HESS with a pulsed power load was analyzed in [26]. However, in these papers, the supercapacitor SOC control was not considered.

In the battery-supercapacitor HESSs, the supercapacitor voltage tends to fluctuate in a large range due to the supercapacitor power fluctuations. Hence, the supercapacitor SOC control is important to reduce the risk of supercapacitor over-charging or over-discharging [29, 87, 88]. Using a supercapacitor SOC controller, the SOC can be maintain within a safe SOC region. SCVCs were used as supercapacitor SOC controllers in filter based HESSs presented in [12, 20, 27]. However, the effect of SCVC on HESS power allocation or energy storage element sizing was not presented in the papers. Due to the non-linearity introduced by the SCVC, it is challenging to analyze the dynamic behavior of the power allocation method. Furthermore, the relationship between the power component supplied by the supercapacitor and the total HESS output power depends on the supercapacitor SOC controllers on the HESS power allocation and dynamic performance are not analyzed in the current literature.

Studying the effects of the supercapacitor SOC controller on the HESS power allocation and dynamic performance is important, a) to select the high pass filter cut-off frequency and the supercapacitor SOC controller parameters required to achieve the desired HESS performance and b) to obtain the required energy storage sizes for a considered HESS application.

In this chapter a HESS power allocation parameter selection method and an energy storage element sizing method are studied for a supercapacitor SOC controlled, filter based battery-supercapacitor HESS. The operation and associated challenges of a filter based HESS with the conventional SCVC are presented first. Then, a SCEC is proposed to circumvent the issues associated with the non-linearity of the SCVC and filter based power allocation method. The effects of the SCEC on the filter based HESS is analyzed. Then, a HESS power allocation parameter selection method and an energy storage element sizing method are proposed for a given application. Experimental results are used to validate the operation, parameter selection and sizing of the supercapacitor energy controlled filter based HESS.

4.2 HESS Operation with SCVC

The block diagram of the outer voltage control loop for the left-hand side boost converter legs is shown in Fig. 4.1. In this chapter, a simplified voltage control loop is considered. The ripple allocator presented in Chapter 3 is not considered in the analysis since it is only used for the initial supercapacitor charging as discussed in Chapter 3 Further, a proportional controller is considered in the SCVC. In the voltage control loop, P_{tot1} is the total input power supplied by the left-hand side boost converter legs. A high-pass filter is employed to allocate the high frequency power variations to the supercapacitor. The high pass filter transfer function is given by,

$$H_{HPF} = \frac{as}{as+1} \tag{4.1}$$

where $a = \frac{1}{\omega_{HPF}}$ and ω_{HPF} is the high pass filter cutoff frequency. To maintain the supercapacitor voltage around a reference voltage value, the power reference signal is then modified using the SCVC. The supercapacitor reference voltage is given by $v_{sc,ref}$ which



Fig. 4.1: Outer voltage control loop for the left-hand side boost converter legs with SCVC

can be selected as explained in Chapter 3. Additionally, the SCVC helps to handle the self-discharging of the supercapacitor. The safe operating region of the supercapacitor is given by,

$$v_{sc,min} \le v_{sc} \le v_{sc,max} \tag{4.2}$$

where $v_{sc,min}$ and $v_{sc,max}$ are the minimum and maximum allowed supercapacitor voltages, respectively.

The total power supplied by the left hand side and the right hand side boost converter legs, P_{tot} can be written as,

$$P_{tot} = P_{tot1} + P_{tot2} = \frac{1}{\eta} P_{HESS}.$$
 (4.3)

 P_{HESS} is the total HESS output power and η represents the efficiency of the HESS where $\eta = \left(\frac{P_{HESS}}{P_{tot}}\right)$. In this analysis, power components are considered as the average values over one output voltage cycle.

A simplified model of the HESS power allocation method is considered as shown in Fig. 4.2 to analyze the dynamic behavior of the HESS. The supercapacitor is modeled as an ideal capacitor by neglecting the ESR value of the supercapacitor. The simplification is valid due to relatively small supercapacitor ESR and the simplification can have an impact on the analysis when ESR of the supercapacitor cannot be neglected [26,89]. Using the model, the relationship between the total power supplied by the supercapacitor, P_{sc} and its voltage can be written as,

$$v_{sc} = v_{sc,init} - \frac{1}{C_{sc}} \int \frac{P_{sc}}{v_{sc}} dt$$
(4.4)

where $v_{sc,init}$ and C_{sc} are the initial supercapacitor voltage and the capacitance of the supercapacitor, respectively. A factor of '2' is used in the model shown in Fig. 4.2 since the SCVC modifies both the left hand side and the right hand side boost converter legs.



Fig. 4.2: Simplified block diagram of the HESS power allocation with the SCVC

Then, the relationship between P_{tot} and P_{sc} can be obtained by,

$$P_{sc} = HPF\left(P_{tot}\right) - 2K_{p,vsc}\left[\left(v_{sc,ref} - v_{sc,init}\right) + \int \frac{P_{sc}}{v_{sc}C_{sc}}dt\right]$$
(4.5)

where $HPF(P_{tot})$ is the output from the high pass filter.

In order to obtain the required energy storage element sizes and also to select the high pass filter cut-off frequency and SCVC gain value, the dynamic relationship between P_{tot} and the power supplied by the individual energy storage elements (P_{sc} and P_{batt}) has to be obtained. But, due to non-linearity in (4.4) and (4.5), it is challenging to obtain the relationships between P_{tot} and power components supplied by the battery and the supercapacitor.

4.3 Supercapacitor Energy Controller

In this section, a SCEC is analyzed for the filter based battery-supercapacitor HESS and the SCEC is used to overcome the issues related to the non-linearity of the SCVC. The modified voltage control loop for the left hand side boost converter leg with the SCEC is illustrated in Fig. 4.3. Instead of controlling the supercapacitor voltage around a reference voltage value, the energy stored in the supercapacitor E_{sc} is controlled around a reference energy level $E_{sc,ref}$. The stored energy in the supercapacitor E_{sc} is calculated using the simplified supercapacitor model and can be written as,

$$E_{sc} = \frac{1}{2} C_{sc} v_{sc}^2$$
 (4.6)

A simplified block diagram as illustrated in Fig. 4.4 is considered to obtain the dynamic behavior of the HESS power allocation method. Using the simplified model, the relationship between the power supplied by the supercapacitor and its stored energy can be written as,

$$E_{sc} = E_{sc,init} - \int P_{sc}dt \tag{4.7}$$



Fig. 4.3: Outer voltage control loop for the left-hand side boost converter legs with SCEC

where $E_{sc,init}$ is the initial stored energy of the supercapacitor which corresponds to $v_{sc,init}$ value. Then using Fig. 4.4, the relationship between the HESS output power and the power component supplied by the supercapacitor can be written as (refer Appendix A.1.1),

$$P_{sc} = \frac{as^2}{as^2 + (1 + 2aK_{p,Esc})s + 2K_{p,Esc}} \frac{1}{\eta} P_{HESS} - \frac{2K_{p,Esc}s}{(s + 2K_{p,Esc})} (E_{sc,ref} - E_{sc,init})$$
(4.8)

Assuming,

$$E_{sc,init} = E_{sc,ref},\tag{4.9}$$

the transfer function between P_{HESS} and P_{sc} can be obtained from (4.8) as,

$$H_{Psc}(s) = \frac{P_{sc}}{P_{HESS}} = \frac{as^2}{as^2 + (1 + 2aK_{p,Esc})s + 2K_{p,Esc}} \frac{1}{\eta}$$
(4.10)

Similarly, the transfer function between P_{batt} and P_{HESS} is,

$$H_{Pbatt}(s) = \frac{P_{batt}}{P_{HESS}} = \frac{(1 + 2aK_{p,Esc})s + 2K_{p,Esc}}{as^2 + (1 + 2aK_{p,Esc})s + 2K_{p,Esc}} \frac{1}{\eta}$$
(4.11)

Further, the transfer function between the supercapacitor energy change, dE_{sc} and P_{HESS} is,

$$H_{dEsc}(s) = \frac{dE_{sc}}{P_{HESS}} = \frac{as}{as^2 + (1 + 2aK_{p,Esc})s + 2K_{p,Esc}} \frac{1}{\eta}$$
(4.12)

4.3.1 Effects of the Supercapacitor Energy Controller on the HESS Power Allocation Strategy

In order to identify the effects of the SCEC on the HESS power allocation strategy, obtained transfer functions in (4.10), (4.11), and (4.12) are further analyzed.

It can be observed that (4.11) has a zero z_1 and two poles p_1 and p_2 where,

$$z_1 = \frac{-2K_{p,Esc}}{(1+2aK_{p,Esc})}; \qquad p_1 = \frac{-1}{a}; \qquad p_2 = -2K_{p,Esc} \qquad (4.13)$$



Fig. 4.4: Simplified block diagram of the HESS power allocation with the SCEC.

It can be shown that,

$$|z_1| < |p_1| |z_1| < |p_2|$$
(4.14)

for non-negative, non-zero a and $K_{p,Esc}$ values. Then, an asymptotic Bode plot of H_{Pbatt} can be drawn as shown in Fig. 4.5. It can be observed that the transfer function H_{Pbatt} demonstrates low pass filter characteristics. Since $H_{Psc} = 1 - H_{Pbatt}$, the transfer function H_{Psc} demonstrates high pass filter characteristics. Using the Bode plot shown in Fig. 4.5, following relationship can be obtained.

$$\frac{\omega_c}{\max(|p_1|, |p_2|)} = \frac{\min(|p_1|, |p_2|)}{|z_1|} \tag{4.15}$$

where ω_c is the gain cross-over frequency of the transfer function H_{Pbatt} . Substituting p_1 and p_2 from (4.13) to (4.15), the gain cross-over frequency can be obtained as,

$$\omega_c = \frac{1 + 2aK_{p,Esc}}{a} \tag{4.16}$$

The gain cross-over frequency of the HESS without any supercapacitor SOC control method can be obtained by substituting $K_{p,Esc}=0$ to (4.16) as,

$$\omega_c' = \frac{1}{a} = \omega_{HPF} \tag{4.17}$$

which is similar to the cut-off frequency of the high pass filter. Superscript '7' denotes the HESS operation without any supercapacitor SOC controller. From (4.16) and (4.17), it is evident that the introduction of SCEC leads to a modification of the HESS power allocation frequency by a factor of $(1 + 2aK_{p,Esc})$. To further investigate the effect of the SCEC on HESS power allocation,(4.11) can be re-written as,

$$H_{Pbatt} = \left[\frac{\omega_c s + \gamma}{s^2 + \omega_c s + \gamma}\right] \frac{1}{\eta}$$
(4.18)

where

$$\gamma = \frac{2K_{p,Esc}}{a} \tag{4.19}$$



Fig. 4.5: Asymptotic Bode plot of the transfer function H_{Pbatt} .

Assume the required H_{Pbatt} gain cross-over frequency is $\omega_{c,req}$. Then, the high pass filter parameter, a and SCEC gain $K_{p,Esc}$ have to be selected such that

$$\frac{1 + 2aK_{p,Esc}}{a} = \omega_{c,req} \tag{4.20}$$

However, from (4.18), infinite number of H_{Pbatt} transfer functions with $\omega_{c,req}$ gain cross-over frequency can be obtained by changing the γ value. Substituting $K_{p,Esc}$ from (4.19) into (4.20) and solving for a gives,

$$a = \frac{\omega_{c,req} \pm \sqrt{\omega_{c,req}^2 - 4\gamma}}{2\gamma} \tag{4.21}$$

It can be observed from (4.21) that there are two combinations of a and $K_{p,Esc}$ values which result in the same H_{Pbatt} transfer function. Moreover from (4.21), for a required gain cross-over frequency $\omega_{c,req}$, γ has to satisfy,

$$0 \le \gamma \le \frac{\omega_{c,req}^2}{4} \tag{4.22}$$

Using,

$$n = \frac{\gamma}{\omega_{c,req}^2} \tag{4.23}$$

and (4.22), a set of all possible H_{Pbatt} transfer functions can be obtained by changing n in the following range.

$$0 \le n \le 0.25 \tag{4.24}$$

Figs. 4.6 (a) and 4.6 (b) demonstrates the effect of n on the step response of the transfer functions H_{Pbatt} and H_{dEsc} for a sample gain cross-over frequency $\omega_{c,req} = 0.05 \text{rads}^{-1}$. The battery power profile demonstrates a low pass filter response to a step output power change as shown in Fig. 4.6 (a). The supercapacitor supplies the balance power component to meet the output power requirement, and hence, the stored energy in the supercapacitor changes as shown in Fig. 4.6 (b). It can be observed that when n = 0, the supercapacitor energy is not controlled and it does not reach the reference energy level (at the steady state $dE_{sc} \neq 0$). When n = 0.25, the fastest battery response can be observed. Further, when n = 0.25, change in the supercapacitor energy dE_{sc} is also minimum.

4.3.2 Calculating the Capacitance of the Supercapacitor C_{sc} for a Set of HESS Power Allocation Parameters and for a Given HESS Power Profile

Next, the supercapacitor sizing for a set of HESS power allocation parameters ($\omega_{c,req}$ and n) and for a given HESS output power profile is considered.



Fig. 4.6: Step response of (a) H_{Pbatt} and (b) H_{dEsc} for various *n* values for a sample gain cross-over frequency $\omega_{c,req} = 0.05 \text{ rads}^{-1}$.

First, the maximum and minimum supercapacitor voltage limits $v_{sc,max}$ and $v_{sc,min}$ have to be selected based on the operation and specifications of the power converter. For the considered boost inverter based HESS, $v_{sc,min}$ can be selected based on the maximum possible gain of the converter while $v_{sc,max}$ has to satisfy,

$$v_{sc,max} \le V_{DC,ref} - \frac{V_o}{2} \tag{4.25}$$

Please refer to (3.10). Using the maximum and minimum supercapacitor voltage limits $v_{sc,max}$ and $v_{sc,min}$, the utilized energy of the supercapacitor $E_{sc,u}$ can be calculated as

$$E_{sc,u} = E_{sc,max} - E_{sc,min} = 0.5C_{sc}(v_{sc,max}^2 - v_{sc,min}^2).$$
(4.26)

 $E_{sc,max}$ and $E_{sc,min}$ are the supercapacitor energy levels corresponding to the maximum and minimum supercapacitor voltage limits $v_{sc,max}$ and $v_{sc,min}$, respectively. If the supercapacitor reference energy level is chosen as,

$$E_{sc,ref} = E_{sc,min} + 0.5E_{sc,u} \tag{4.27}$$

then equal amount of energy $(0.5E_{sc,u})$ is available for supercapacitor charging and discharging. The supercapacitor voltage level corresponding to the reference energy level can be obtained as,

$$v_{sc,ref} = \sqrt{0.5(v_{sc,min}^2 + v_{sc,max}^2)}$$
(4.28)

Then, using the considered HESS power profile and the set of HESS power allocation parameters ($\omega_{c,req}$ and n), the supercapacitor energy variation profile dE_{sc} can be obtained using the transfer function (4.12). Since only half of the utilized energy is available for either the supercapacitor charging or discharging,

$$0.5E_{sc,u} = max(|dE_{sc}|)$$
(4.29)

Using (4.26) and (4.29), the required supercapacitor value for the considered HESS output power profile and for the set of power allocation parameters, can be obtained as,

$$C_{sc} = \frac{4max(|dE_{sc}|)}{(v_{sc,max}^2 - v_{sc,min}^2)}$$
(4.30)

4.3.3 Calculating the Battery Specifications for a Set of HESS Power Allocation Parameters and for a Given HESS Output Power Profile

For a given HESS output power profile and for a set of HESS power allocation parameters ($\omega_{c,req}$ and n), the corresponding battery power profile P_{batt} can be obtained from (4.11). Then, the required power rating of the battery $P_{batt,rated}$ can be selected as,

$$P_{batt,rated} \ge max(P_{batt}) \tag{4.31}$$

Further, the change in the battery energy dE_{batt} due to the considered HESS operation can be obtained using

$$H_{dEbatt} = \frac{dE_{batt}}{P_{HESS}} = \frac{1}{s} \left[\frac{(1 + 2aK_{p,Esc})s + 2K_{p,Esc}}{as^2 + (1 + 2aK_{p,Esc})s + 2K_{p,Esc}} \right] \frac{1}{\eta}$$
(4.32)

Then, the required energy rating of the battery $E_{batt,rated}$ should be selected such that,

$$E_{batt,rated} \ge max(|dE_{batt}|). \tag{4.33}$$

4.4 HESS Power Allocation Parameter Selection and Supercapacitor Sizing for a PV Power Smoothing Application

This section presents a HESS power allocation parameter selection and supercapacitor sizing method for a HESS operation in a PV power smoothing application. The power allocation parameters and the required supercapacitor size are selected to achieve required smoothness of the battery power profile. As a measure of smoothness, the gradient of the battery power profile is considered. The filter and SCEC based power allocation strategy considered in this chapter reduce the variability of the battery power profile. Additionally, the power allocation method indirectly reduces the peak power requirement from the battery as presented later in the chapter.

Fig. 4.7 illustrates an example application of the HESS where it is used to smooth the PV power injected to the grid at the common coupling point. The PV power profiles are obtained by scaling the data available in [86] to match the HESS experimental setup parameters summarized in Table 4.2. The compensating power supplied by the HESS can be obtained as the difference between the generated PV power and the power supplied to the grid. In this research, a simple rate limiter is used to limit the per minute ramp rate to 10% of the rated power of the PV plant [90]. Relevant power profiles P_{PV} , P_{grid} and P_{HESS} are illustrated in Fig. 4.8 (a) where P_{PV} and P_{grid} are the generated power from the PV plant and the power injected to the grid at the common coupling point respectively. Fig. 4.8 (b) depicts the total power supplied by the battery and the supercapacitor, P_{tot} . The gradient of the P_{tot} power profile is considered to quantify the variability, and the histogram of non-zero gradient values of the P_{tot} power profile is shown in Fig. 4.9.

To observe the relationship between the HESS power allocation parameters and the battery power gradient, a set of battery power profiles is obtained using the P_{HESS} power profile in Fig. 4.8 and (4.18) for a set of ω_c and n values. Then, the maximum gradient value of the battery power profile $max(|\nabla P_{batt}|)$ is plotted against ω_c and n as illustrated in Fig. 4.10 (a). Additionally, the required supercapacitor values for each set of power allocation parameters are calculated and shown in Fig. 4.10(b).

Then, a linear search method and the data in Fig. 4.10 are used to obtain the required HESS power allocation parameters which minimizes the required capacitance value of the supercapacitor, C_{sc} subject to $max(\nabla P_{batt}) \leq \nabla P_{batt,limit}$. Here $\nabla P_{batt,limit}$ is a predefined maximum allowable gradient value for the battery power profile. In the linear search method, first a set of HESS power allocation parameters which satisfy $max(\nabla P_{batt}) \leq \Delta P_{batt,limit}$.



Fig. 4.7: Block diagram of the HESS in a PV power smoothing application



Fig. 4.8: Power profiles considered for the HESS parameter selection. (a) PV power, power injected to the grid and the HESS power and (b) the total power profile of the battery and the supercapacitor



Fig. 4.9: Histogram of the non-zero gradient values of the P_{tot} power profile

 $\nabla P_{batt,limit}$ is selected and then the ω_c and *n* values corresponding to the minimum required supercapacitor value, $C_{sc,req}$ in the selected set are identified. After the required $\omega_{c,req}$ and n_{req} are identified, the high pass filter parameter and the SCEC gain can be found using (4.16), (4.21), and (4.23) as,

$$a = \frac{\omega_{c,req} + \sqrt{\omega_{c,req}^2 - 4n_{req}\omega_{c,req}^2}}{2n_{req}\omega_{c,req}^2}$$
(4.34)



Fig. 4.10: Variation of (a) $max(|\nabla P_{batt}|)$ and (b) C_{sc} with n and ω_c .

$$K_{p,Esc} = \frac{a\omega_{c,req} - 1}{2a} \tag{4.35}$$

Table 4.1 summarizes the high pass filter parameter a, SCEC gain $K_{p,Esc}$, and the required supercapacitor values, $C_{sc,req}$ obtained for various values. The $\nabla P_{batt,limit}$ is calculated as the ρ^{th} percentile value of the non-zero gradient values of the P_{tot} power profile. The considered maximum and minimum supercapacitor voltage limits $v_{sc,max}$ and $v_{sc,min}$ are given in Table 4.2.

The operation of the HESS with the selected power allocation parameters and the supercapacitor value for the 75th percentile value is illustrated in Fig. 4.11. The percentile value ρ can be selected based on the required smoothness of the battery power profile. From Fig. 4.11 (a) it can be observed that the battery supplies only the low frequency power component of the P_{tot} power variation. The selected HESS power allocation



Fig. 4.11: Operation of the HESS when $\rho = 75$. (a) P_{tot} and P_{batt} power profiles, (b) gradient of the battery power profile, (c) supercapacitor energy variation and (d) voltage variation of the supercapacitor.

parameters ensure the HESS operation within the battery power gradient limits as shown in Fig. 4.11 (b). The SCEC maintains the supercapacitor energy variation around the reference energy level as illustrated in Fig. 4.11 (c). Furthermore, the selected power allocation parameters and the supercapacitor value ensures the supercapacitor operation within the pre-defined voltage limits as shown in Fig. 4.11 (d)

Next, the HESS power allocation parameter selection and the supercapacitor sizing for the HESS operation with the SCVC is considered. When the conventional SCVC is used it is difficult to properly size the supercapacitor due to the nonlinear relationship between

	$\nabla P_{batt,limit}$ [W/s]	HESS	operati	on with	HESS	operation without	
ρ		SCEC			any supercapacitor SoC		
					controller		
		a	$K_{p,Esc}$	$C_{sc,req}$ [F]	<i>a'</i>	$C_{sc,req}'$ [F]	
$90^{\rm th}$	2.5863	244.88	0.0230	8.35	19.42	11.09	
85^{th}	1.9098	325.82	0.0160	11.13	28.57	15.28	
$80^{\rm th}$	1.4571	191.89	0.0099	13.66	37.74	18.83	
$75^{\rm th}$	1.1451	132.44	0.0062	15.44	47.62	22.06	
$70^{\rm th}$	0.8979	117.65	0.0043	16.97	60.61	25.60	
65^{th}	0.7177	142.86	0.0035	18.96	71.43	28.25	
$60^{\rm th}$	0.5825	628.57	0.0052	20.77	86.96	31.79	

Table 4.1: HESS power allocation parameters and supercapacitor values for various $\nabla P_{batt,limit}$ values with and without SCEC

 P_{HESS} and P_{sc} . Hence, the only way to estimate a suitable size of the supercapacitor is to size it without considering the effects of the SCVC. The transfer function between the supercapacitor energy change dE'_{sc} and P_{HESS} for the operation of the HESS without any supercapacitor SOC controller can be written as,

$$H'_{dEsc} = \frac{dE'_{sc}}{P_{HESS}} = \left[\frac{a'}{a's+1}\right]\frac{1}{\eta} = \left[\frac{\frac{1}{\omega'_c}}{\frac{1}{\omega'_c}s+1}\right]\frac{1}{\eta}$$
(4.36)

The superscript '' denotes the HESS operation without supercapacitor SOC controllers. It can be observed that the power allocation frequency is equal to the high pass filter cut-off frequency. The energy variation of the supercapacitor dE'_{sc} for the HESS output power profile and any ω'_c value can be obtained using (4.36). Then the required supercapacitor value can be obtained from,

$$C'_{sc} = \frac{4max\left(|dE'_{sc}|\right)}{\left(v_{sc,max}^2 - v_{sc,min}^2\right)} \tag{4.37}$$

Similar to the HESS operation with the SCEC, the required power allocation frequency $\omega'_{c,req}$ is calculated to minimize the supercapacitor value C'_{sc} , subject to $max(|\nabla P'_{batt}|) \leq \nabla P_{batt,limit}$. The corresponding minimum required supercapacitor values $C'_{sc,req}$ are summarized in Table 4.1 and compared against the required supercapacitor values calculated for the HESS operation with the SCEC. It is evident that using the SCEC not only allows the precise supercapacitor sizing but also the required supercapacitor value with the SCEC is significantly lower compared to the estimated values for the operation with the SCVC.

4.5 Experimental Results

The performance of the proposed system is verified using the experimental prototype illustrated in Fig. 4.12. A 9.6Ah, 25.6V LiFePO₄ battery from K2 Energy and two 16V, 58F Maxwell supercapacitor modules connected in series are used in the prototype and the parameters of the HESS prototype are summarized in Table 4.2. The controllers are implemented on a DSpace DS1006 platform.

A 900s HESS power profile was considered in the experiments as shown in Fig. 4.13 (a). For the filter and SCEC parameter calculation, $\nabla P_{batt,limit}$ was chosen as the 75th percentile of the non-zero gradient values of the P_{tot} power profile (0.6986W/s). Then, the required HESS power allocation parameters were obtained as $\omega_{c,req}=0.013$ rads⁻¹ and $n_{req}=0.208$ using the linear search method. The corresponding *a* and $K_{p,Esc}$ values are 255.81 and 0.0045, respectively. The required supercapacitor value for the considered HESS output power profile and for the calculated HESS power allocation parameters is



Fig. 4.12: HESS experimental setup

27.19F. Hence, a commercially available 29F supercapacitor was used in the experiments.

Fig.	4.13	depicts	the	operation	of	the	HESS	with	the	calculated	power	allocation
------	------	---------	-----	-----------	----	-----	------	------	-----	------------	-------	------------

Parameter	Value	Parameter	Value						
Power converter parameters									
V_o	60 V	60 V							
L_{batt1}, L_{batt2}	$210~\mu\mathrm{H}$	L_{sc1}, L_{sc2}	$210~\mu\mathrm{H}$						
C_1, C_2	$60 \ \mu F$	f	$50~\mathrm{Hz}$						
$v_{sc,max}$	28 V	20 V							
Grid parameters									
V_g	60 V	L_g	$20 \mathrm{~mH}$						
Voltage control loop parameters									
$K_p(PR)$	0.1	$K_i(PR)$	4						
Current control loop parameters									
$K_p(PI_{Ibatt})$	2.615	$K_p(PI_{Isc})$	2.615						
$K_i(PI_{Ibatt})$	8.44×10^{-5}	$K_i(PI_{Isc})$	8.44×10^{-5}						
PQ controller parameters									
$K_p(PI_P)$	2×10^{-6}	$K_p(PI_Q)$	1×10^{-5}						
$K_i(PI_P)$	0.14	$K_i(PI_Q)$	4						

 Table 4.2: Hybrid energy storage system parameters



Fig. 4.13: Operation of the HESS with SCEC (a) power reference of the HESS, (b) variation of P_{tot} , P_{batt} and P_{sc} , (c) gradient of the P_{tot} profile, (d) gradient of the battery power profile, (e) supercapacitor energy variation, (f) supercapacitor voltage variation, (g) battery current variation and (h) supercapacitor current variation.

parameters and the 29F supercapacitor. According to Fig. 4.13 (b) the battery supplies the slow varying power component of the P_{tot} power profile while supercapacitor supplies the fast power fluctuations. According to Fig. 4.13 (c), the maximum gradient of the P_{tot} power profile is close to 20W/s which is clearly higher than the $\nabla P_{batt,limit}$ value. However, the battery power gradient remains within the predefined limits as illustrated in Fig. 4.13 (d). The energy variation of the supercapacitor is depicted in Fig. 4.13 (e) and it is evident that the SCEC maintains the supercapacitor energy around the reference energy level. Fig. 4.13 (f) shows the supercapacitor voltage which lies within the maximum and minimum voltage limits throughout the experiment. The battery and the supercapacitor current waveforms are illustrated in Figs. 4.13(g) and 4.13 (h) respectively. As expected, the battery supplies the slow varying current while the supercapacitor supplies the high frequency current fluctuations. Further from Figs. 4.13(b) and 4.13(g), it can be observed that, by limiting the maximum gradient of the battery power profile, the peak battery power and the number of charge/discharge cycles of the battery are also reduced.

4.6 Conclusions

In this chapter, the HESS power allocation parameter selection and energy storage element sizing for a supercapacitor SOC controlled filter based HESS was considered. None of the existing literature studied the power allocation parameter selection, energy storage element sizing and dynamic behavior of a supercapacitor SOC controlled filter based battery-supercapacitor HESS.

Due to the nonlinearity of the conventional SCVC and filter based HESS power allocation method, it was difficult to properly design the HESS power allocation parameters as well as size the supercapacitor. Hence, a SCEC is proposed in this chapter to circumvent the issues associated with the SCVC and its effects on the HESS power allocation method were extensively analyzed. It was shown that the SCEC has a significant impact on the HESS power allocation frequency. Subsequently, a HESS power allocation parameter selection and energy storage element sizing method for a given HESS output power profile were proposed. Finally, the experimental results were presented to validate the operation of the filter based HESS with the SCEC and it was shown that the selected power allocation parameters and the supercapacitor size can ensure the operation of the HESS within the supercapacitor voltage limits as well as within the battery power gradient limits.

Chapter 5

A Fixed Frequency Sliding Mode Controller for the Boost Inverter Based Hybrid Energy Storage System

This chapter presents a fixed frequency SM controller for the boost inverter based battery-supercapacitor HESS [66]. The SM controller enables better output capacitor reference following and the main advantage of the proposed SM controller, as compared with the traditional DL control method, is in eliminating possible DC current injection into the grid when the equivalent series resistance (ESR) values of the boost inductors become unequal due to the tolerances and temperature variations. The SM controller is implemented using variable amplitude PWM carrier signals which are generated using the output capacitor voltage and inductor current measurements which allow operation without using the output capacitor currents measurement. The battery connected inductor reference currents for the SM controller are generated by a supercapacitor energy controller (SCEC) which is responsible for the HESS power allocation. A supercapacitor sizing method for a given application is also presented for the SM controlled boost inverter based battery-supercapacitor HESS. The performance of the proposed SM controlled, grid connected battery-supercapacitor HESS is experimentally verified using a laboratory prototype.

5.1 Introduction

The operation of an interleaved boost inverter based battery-supercapacitor HESS was discussed in Chapter 3. A filter based power allocation strategy along with a SCVC was employed to allocate the power components among the two energy storage elements. As a way of circumventing the issues related to the non-linearity of the SCVC, a SCEC was introduced in Chapter 4 and the power allocation parameter selection and the energy storage element sizing for a filter based HESS with the SCEC was extensively analyzed for a PV power smoothing application. A double loop (DL) control method with proportional resonant (PR) controller based outer voltage control loops and PI controller based inner current control loops were used to control the boost inverter based HESS. However, the voltage drop across the boost converter leg inductors were neglected when designing the outer voltage control loop, which leads to a DC current injection into the grid when the ESR values of the left hand side and right hand side boost converter legs are not equal.

The SM techniques were extensively used to control the boost inverter topology [54,71,72]. Unlike the DL control method, the SM control methods are robust to parameter variations and modeling errors and achieve better reference following performance. However, the existing SM controllers for the boost inverter were based on the traditional SM control technique which leads to a high and variable switching frequency operation which can increase the power losses and system components design complexity [91]. Fixed frequency SM controllers were studied in literature for the boost DC/DC converters but not for the boost inverter. The fixed frequency SM controller in [92] was designed for a constant load and hence the method is not applicable to a grid connected boost inverter based HESS. A fixed frequency SM current controller was proposed in [93] for the boost DC/DC converter, however, the method required a PI controller based outer voltage control loop to generate the required inductor current reference signals. A pulse width modulation (PWM) based fixed frequency SM controller was proposed for the boost DC/DC converter in [94], but the controller required a capacitor current measurement which increased the required number of current sensors in the system and this may also influence the filter performance of the capacitor due to the impedance modification of the capacitor path [91].

A few SM controlled battery-supercapacitor HESS were proposed in literature. A simulation based comparison documented in [39] reveals that SM controller based HESS has a higher robustness and stability compared to the HESS operation with PI controllers. SM controlled battery, supercapacitor and fuel cell HESSs were presented in [25, 40], however the design procedure of the SM controller was not analyzed. A SM coefficient modification method was discussed in [41] for a battery-supercapacitor HESS in order to

control the SOC of the energy storage elements, but the possible modification ranges were not presented in the paper. None of the above mentioned SM controllers proposed for the HESSs were validated using experimental results and none of the papers analyzed the power allocation between the battery and the supercapacitor. Further, all the proposed SM controlled HESSs were DC link based structures and hence, many design issues related to the implementation of the SM controlled direct grid integrated HESSs have not been sufficiently covered.

In this chapter, a fixed frequency SM controlled boost inverter based battery-supercapacitor HESS is presented [66]. The SM controller is implemented using variable amplitude PWM carrier signals which are generated using the output capacitor voltage and inductor current measurements which allow operation without using the output capacitor currents measurement. The battery connected inductor reference currents for the SM controller are generated by a SCEC which is responsible for the HESS power allocation. A supercapacitor sizing method for a given application The performance of the proposed SM controlled, grid connected is also presented. battery-supercapacitor HESS is experimentally verified using a laboratory prototype. Furthermore, the performance of the SM controller is compared against a DL controlled boost inverter based battery-supercapacitor HESS. The main advantage of the proposed SM controller, as compared with the traditional DL control method, is in eliminating possible DC current injection into the grid when the ESR values of the boost inductors become unequal due to the tolerances and temperature variations.

5.2 Effect of the Inductor ESR Values on Output Capacitor Reference Voltage Tracking with DL Control Method

The DL controller for the HESS presented in Chapter 3 was designed using PR controller based outer voltage control loops and PI controller based inner current control loops. However, in the voltage control loop design, the voltage drops across the ESR of the boost inductors were neglected [55, 63, 64]. This results in a DC shift tracking error in the output capacitor voltages when the ESR values are significant. A simulation test was conducted to observe the effect of the boost inductor ESR values on the output capacitor voltage DC shift error using the prototype parameters summarized in Table 5.1 and the DL controller presented in [64]. In the simulation, the ESR values of the boost inductors L_{sc1} , L_{sc2} , L_{batt1} , and L_{batt2} are assumed to be equal. Fig. 5.1 illustrates the effect of the boost inductor ESR values on the DC shift of the output capacitor voltage



Fig. 5.1: Comparison of the effect of the boost inductor ESR on the output capacitor voltage reference tracking with the double loop control method and with the sliding mode control method, (a) DC shift of the output capacitor voltage and (b) DC shift error of the output capacitor voltage

for a sample case when the HESS supplies 50W active power and 25VAr reactive power to the grid. A significant increment in the output capacitor DC shift error can be observed with the increasing boost inductor ESR values. If the ESR values of the left hand side boost converter leg inductors are different from the right hand side boost converter leg inductors, due to the tolerances and temperature variations, then there are unequal DC voltage shifts in v_{o1} and v_{o2} and hence a DC voltage component ΔV_{DC} appears in the HESS differential output voltage. The DC voltage component results in undesirable DC current flowing into the grid.

The DC component of the HESS output voltage ΔV_{DC} , is given by,

$$\Delta V_{DC} = V_{DC1} - V_{DC2} \tag{5.1}$$

where V_{DC1} and V_{DC2} are the DC shifts of the output capacitor voltages v_{o1} and v_{o2} respectively. Then, the current supplied to the grid, i_g can be written as,

$$i_g = I_o \sin(\omega t + \alpha) + \frac{\Delta V_{DC}}{R_{Lg}}$$
(5.2)

where R_{Lg} is the ESR value of the grid interfacing inductor L_g . Hence, a slight DC shift in the HESS output voltage can lead to a significant output current DC component due to the low ESR of the grid interfacing inductor (less than 1 Ω). According to the IEEE standards, it is required to limit the DC current injection to 0.5% of the rated current of the inverter [95].

5.3 Sliding Mode Controller Analysis

The configuration of the boost inverter based battery-supercapacitor HESS is illustrated in Fig. 5.2 and the gate signals for the top switches (subscript T) and for the bottom switches (subscript B) of each converter leg are complementary. In order to design a SM controller for the HESS, the four boost converter legs of the HESS configuration are considered as four subsystems with unique objectives and each boost converter leg can be defined as a single input system. This way of sub-dividing a complex system into several independent single input subsystems is known as decentralized switching scheme and widely used in SM controller design for complex power converters [27,29].

In the SM controller design, the supercapacitor connected left hand side and right hand side boost converter legs are controlled to track the output capacitor reference voltages $v_{o1,ref}$ and $v_{o2,ref}$ where,

$$v_{o1,ref} = V_{DC,ref} + \frac{V_o}{2}\sin(\omega t + \delta)$$
(5.3)

$$v_{o2,ref} = V_{DC,ref} - \frac{V_o}{2}\sin(\omega t + \delta).$$
(5.4)

The battery connected boost converter legs are controlled in the inductor current controlled mode to facilitate the power allocation of the HESS. In this chapter, SM controller analysis is provided only for the left hand side converter legs, but a similar analysis can be performed for the right hand side converter legs as well.

5.3.1 Left Hand Side Supercapacitor Connected Boost Converter Leg

The objective of the SM controller for the left hand side supercapacitor connected boost converter leg is to follow the output capacitor reference voltage given by (5.3). Equivalent circuit of the left hand side boost converter leg is illustrated in Fig. 5.3. The left hand side battery connected boost converter leg is operated in the inductor current controlled mode as explained later and hence the effect of the left hand side battery connected boost converter leg operation on the supercapacitor connected boost converter leg can be modeled using a current source with i_{obatt1} current as shown in Fig. 5.3. The left hand side supercapacitor connected boost converter leg controller has to control v_{o1} according to the reference voltage $v_{o1,ref}$, irrespective of the current i_{obatt1} which acts as a disturbance to its operation. The internal resistance of the output capacitors, R_{C1} , and R_{C2} are neglected to simplify the analysis.







Fig. 5.3: Equivalent circuit of the HESS in the viewpoint of the left hand side supercapacitor connected boost converter leg

The dynamic behavior of the system in the viewpoint of the left hand side supercapacitor connected boost converter leg can be described by following differential equations.

$$\frac{di_{Lsc1}}{dt} = \frac{v_{sc} - i_{Lsc1}R_{Lsc1} - (1 - u_{sc1})v_{o1}}{L_{sc1}}$$
(5.5)

$$\frac{dv_{o1}}{dt} = \frac{i_{obatt1} - i_{o1} + (1 - u_{sc1})i_{Lsc1}}{C_1}$$
(5.6)

where u_{sc1} is the logic state of the switch $S_{wsc1,B}$. Sliding surface for the left hand side supercapacitor connected boost converter leg, S_{sc1} is chosen as,

$$S_{sc1} = K_{sc1,a}(v_{o1} - v_{o1,ref}) + K_{sc1,b} \int (v_{o1} - v_{o1,ref})dt + K_{sc1,c}i_{Lsc1}$$
(5.7)

with the switching low,

$$u_{sc1} = \begin{cases} 1 & S_{sc1} < 0 \\ 0 & S_{sc1} > 0 \end{cases}$$
(5.8)

where $K_{sc1,a}$, $K_{sc1,b}$, and $K_{sc1,c}$ represents the sliding coefficients. Similar sliding surfaces were used in literature to control various power converter configurations with SM control techniques [39, 71, 96–99]. In (5.7), the integral of the voltage error is introduced to reduce the output capacitor voltage tracking error as well as to retain the voltage reference information in the equivalent control input which is obtained by solving $\frac{dS_{sc1}}{dt} = 0$ [96]. Once the sliding mode operation is achieved ($S_{sc1} = 0$), the left hand side supercapacitor connected inductor current is given by,

$$i_{Lsc1} = \frac{K_{sc1,a}}{K_{sc1,c}} (v_{o1,ref} - v_{o1}) + \frac{K_{sc1,b}}{K_{sc1,c}} \int (v_{o1,ref} - v_{o1}) dt$$
(5.9)

which is similar to traditional inductor current control using the voltage tracking error and its integral [71].

The existence condition of the SM controller for the left hand side supercapacitor connected boost converter leg can be written as,

$$\lim_{S_{sc1}\to 0} \left(S_{sc1} \frac{dS_{sc1}}{dt} \right) < 0 \tag{5.10}$$

Then,

$$\begin{cases} \frac{dS_{sc1}}{dt} < 0 & \text{if } S_{sc1} > 0\\ \frac{dS_{sc1}}{dt} > 0 & \text{if } S_{sc1} < 0 \end{cases}$$
(5.11)

When $S_{sc1} > 0$, $u_{sc1} = 0$ according to the switching low given in (5.8). Then using (5.5), (5.6), (5.7), and (5.11)

$$K_{sc1,a}\left(\frac{i_{obatt1} - i_{o1} + i_{Lsc1}}{C_1}\right) + K_{sc1,c}\left(\frac{v_{sc} - i_{Lsc1}R_{Lsc1} - v_{o1}}{L_{sc1}}\right) < 0.$$
(5.12)

Since the switching frequency is high compared to the frequency of the output capacitor reference voltage, $v_{o1,ref} \approx 0$ for SM controller analysis [54]. When $S_{sc1} < 0$, according to the switching law in (5.8), $u_{sc1} = 1$. Using (5.5), (5.6), (5.7), and (5.11),

$$K_{sc1,a}\left(\frac{i_{obatt1} - i_{o1}}{C_1}\right) + K_{sc1,c}\left(\frac{v_{sc} - i_{Lsc1}R_{Lsc1}}{L_{sc1}}\right) > 0.$$
(5.13)

The sliding coefficients $K_{sc1,a}$, $K_{sc1,b}$, and $K_{sc1,c}$ must comply with the inequalities (5.12) and (5.13) to satisfy the existence condition of the SM for the left hand side supercapacitor connected boost converter leg. In order to reach the sliding surface from all possible initial conditions, it is sufficient that $(K_{sc1,a}, K_{sc1,b}, K_{sc1,c}) > 0$ provided that SM exists [54].

The equivalent control concept is employed to design the fixed frequency SM controller for the boost inverter based HESS and the equivalent control input for the SM controller is obtained by solving,

$$\frac{dS_{sc1}}{dt} = 0 \tag{5.14}$$

Then, using (5.5), (5.6), and (5.7), the equivalent control input can be obtained as,

$$\bar{u}_{sc1,eq} = \frac{k_{sc1,a}(i_{obatt1} - i_{o1}) + k_{sc1,b}(v_{o1} - v_{o1,ref}) + k_{sc1,c}(v_{sc} - i_{Lsc1}R_{Lsc1})}{k_{sc1,c}v_{o1} - k_{sc1,a}i_{Lsc1}}$$
(5.15)

where,

$$\begin{cases}
k_{sc1,a} = K_{sc1,a}L_{sc1} \\
k_{sc1,b} = K_{sc1,b}L_{sc1}C_1 \\
k_{sc1,c} = K_{sc1,c}C_1
\end{cases}$$
(5.16)

are the fixed gain parameters of the proposed SM controller for the left hand side supercapacitor connected boost converter leg. $\bar{u}_{sc1,eq} = (1 - u_{sc1,eq})$ where $u_{sc1,eq}$ is the equivalent control input to the switches $Sw_{sc1,B}$. Please refer Appendix A.1 for the derivation. If the SM exists for the left hand side supercapacitor connected boost converter leg, $0 < \bar{u}_{sc1,eq} < 1$.

In this chapter, the fixed frequency operation of the SM controller is achieved by using a pulse width modulator. In the pulse width modulation, the duty ratio d is obtained by comparing a control signal against a triangular carrier signal with amplitude v_{tri} [94]. Symmetrical triangular carrier signals in the range of 0 and 1 are used in pulse width modulation. Then,

$$0 < \left[d = \frac{v_c}{v_{tri}}\right] < 1 \tag{5.17}$$

The equivalent control input for the SM controller corresponds to the duty ratio of the PWM control method [94]. By equating $\bar{u}_{sc1,eq} = d$ and using (5.15) and (5.17), the SM controller can be implemented using a control signal $v_{c,sc1}$ and a triangular carrier signal with $v_{tri,sc1}$ amplitude as,

$$v_{c,sc1} = k_{sc1,m} [k_{sc1,a}(i_{obatt1} - i_{o1}) + k_{sc1,b}(v_{o1} - v_{o1,ref}) + k_{sc1,c}(v_{sc} - i_{Lsc1}R_{Lsc1})] = k_{sc1,ma}(i_{obatt1} - i_{o1}) + k_{sc1,mb}(v_{o1} - v_{o1,ref}) + k_{sc1,mc}(v_{sc} - i_{Lsc1}R_{Lsc1})$$
(5.18)

and

$$v_{tri,sc1} = k_{sc1,m} [k_{sc1,c} v_{o1} - k_{sc1,a} i_{Lsc1}]$$

$$= k_{sc1,mc} v_{o1} - k_{sc1,ma} i_{Lsc1}$$
(5.19)

where $k_{sc1,m}$ is a scaling factor introduced to scale the carrier signal amplitude to a practical magnitude level. The controller gains $k_{sc1,a}$, $k_{sc1,b}$ and $k_{sc1,c}$ have to be selected to ensure the reaching and existence condition of the SM.

5.3.2 Left Hand Side Battery Connected Boost Converter Leg

The equivalent circuit of the battery-supercapacitor HESS in the viewpoint of the left hand side battery connected boost converter leg can be illustrated as in Fig. 5.4. The battery connected boost converter legs are controlled in the inductor current control mode in order to facilitate the power allocation between the battery and the supercapacitor. Since the left hand side supercapacitor connected boost converter leg is used to control the output capacitor voltage v_{o1} , the effect of the left hand side supercapacitor connected boost converter leg on the left hand side battery connected boost converter leg operation can be modeled using a voltage source with v_{o1} amplitude as illustrated in the equivalent circuit shown in Fig. 5.4. The dynamic behavior of the left hand side battery connected boost converter leg can be described by following differential equation.

$$\frac{di_{Lbatt1}}{dt} = \frac{v_{batt} - i_{Lbatt1}R_{Lbatt1} - (1 - u_{batt1})v_{o1}}{L_{batt1}}$$
(5.20)

where, the logic state of the switch $Sw_{batt1,B}$ is denoted as u_{batt1} . The sliding surface for the left hand side battery connected boost converter leg is selected to mitigate the inductor current tracking error and its integral as

$$S_{batt1} = (i_{Lbatt1} - i_{Lbatt1,ref}) + K_{batt1} \int (i_{Lbatt1} - i_{Lbatt1,ref}) dt$$
(5.21)

with the switching law,

$$u_{batt1} = \begin{cases} 1 & S_{batt1} < 0 \\ 0 & S_{batt1} > 0 \end{cases}$$
(5.22)

where K_{batt1} represents the sliding coefficient. $i_{Lbatt,ref}$ is the inductor current reference signal which is generated by the HESS power allocation method explained in Section 5.3.4. Existence condition of the considered SM is given by,

$$\lim_{S_{batt1}\to 0} \left(S_{batt1} \frac{dS_{batt1}}{dt} \right) < 0 \tag{5.23}$$

Using (5.20), (5.21), and (5.22), the existence condition in (5.23) when $S_{batt1} < 0$ and $S_{batt1} > 0$ can be simplified as,

$$\frac{v_{batt} - i_{Lbatt1} R_{Lbatt1}}{L_{batt1}} > 0 \tag{5.24}$$

and

$$\frac{(v_{batt} - i_{Lbatt1}R_{Lbatt1}) - v_{o1}}{L_{batt1}} < 0 \tag{5.25}$$

respectively. With the inherent operation of the boost converter, it can be seen that $v_{batt} - i_{Lbatt1}R_{Lbatt1} > 0$ and $(v_{batt} - i_{Lbatt1}R_{Lbatt1}) - v_{o1} < 0$. Because of that, the existence condition for the SM is satisfied. Similar to the previous case, the equivalent control input for the SM can be obtain as,

$$\bar{u}_{batt,eq} = \frac{v_{batt} - i_{Lbatt1}R_{Lbatt1} + k_{batt1}(i_{Lbatt1} - i_{Lbatt1,ref})}{v_{o1}}$$
(5.26)

where

$$k_{batt1} = K_{batt1} L_{batt1} \tag{5.27}$$

is the fixed gain parameter for the left hand side battery connected boost converter leg. Please refer Appendix A.1 for the derivation. $\bar{u}_{batt1,eq} = (1 - u_{batt,eq})$ where $u_{batt,eq}$ is the equivalent control input to the switch $Sw_{batt1,B}$. Then,

$$0 < \left[\bar{u}_{batt,eq} = \frac{v_{batt} - i_{Lbatt1}R_{Lbatt1} + k_{batt1}(i_{Lbatt1} - i_{Lbatt1,ref})}{v_{o1}}\right] < 1.$$
(5.28)



Fig. 5.4: Equivalent circuit of the HESS in the viewpoint of the left hand side battery connected boost converter leg
The SM controller for the left hand side battery connected boost converter leg can be implemented using a control signal $v_{c,batt1}$ and a triangular carrier signal with amplitude $v_{tri,batt1}$ as,

$$v_{c,batt1} = k_{batt1,m} \left[v_{batt} - i_{Lbatt1} R_{Lbatt1} + k_{batt1} (i_{Lbatt1} - i_{Lbatt1,ref}) \right]$$
(5.29)

and

$$v_{tri,batt} = k_{batt1,m} v_{o1}.$$
(5.30)

In equations (5.29) and (5.30), a scaling factor $k_{batt1,m}$ is used to scale the carrier signal amplitude to a practical magnitude level.

Fig. 5.5 represents the block diagram of the SM controller for the boost inverter based battery-supercapacitor HESS.



Fig. 5.5: Block diagram of the proposed sliding mode controller.

5.3.3 Sliding Coefficients Selection for the HESS

The sliding coefficients of the SM controller have to be selected properly to satisfy the SM existence and reaching conditions. This has to be done considering the worst case operation of the system. The worst case operation of the supercapacitor connected boost converter leg occurs when the total power requirement of the HESS is allocated to the supercapacitor. Then, the existence condition for the left hand side supercapacitor connected boost converter leg SM controller can be re-written as,

$$max\left[K_{sc1,a}\left(\frac{-i_{o1}+i_{Lsc1}}{C_1}\right)+K_{sc1,c}\left(\frac{v_{sc}-v_{o1}}{L_{sc1}}\right)\right]<0$$
(5.31)

$$\min\left[K_{sc1,a}\left(\frac{-i_{o1}}{C_1}\right) + K_{sc1,c}\left(\frac{v_{sc}}{L_{sc1}}\right)\right] > 0$$
(5.32)

The HESS output current is given by,

$$i_{o1} = I_o \sin(\omega t + \alpha) \tag{5.33}$$

Then, the supercapacitor current when the total power requirement is allocated to the supercapacitor can be written as,

$$i_{Lsc1} = \frac{v_{o1}(i_{o1} + i_{c1})}{v_{sc}} \tag{5.34}$$

In order to simplify the analysis, the effect of the output capacitor current on i_{Lsc1} and the switching frequency ripple component can be neglected [72]. Furthermore, $\delta \approx 0$ and $V_o \approx V_g$. Then using (5.31), (5.32), and (5.34), the existence condition of the left hand side supercapacitor connected boost converter leg SM controller can be simplified as,

$$0 < \frac{K_{sc1,a}}{K_{sc1,c}} < \frac{v_{sc,min}C_1}{I_{o,max}L_{sc1}}$$
(5.35)

Using (5.16), the controller gains $k_{sc1,a}$ and $k_{sc1,c}$ have to be selected such that,

$$0 < \frac{k_{sc1,a}}{k_{sc1,c}} < \frac{v_{sc,min}}{I_{o,max}}$$

$$(5.36)$$

The inequality (5.36) provides the basis for the selection of the controller gains. Then, a simulation and experimental analysis based empirical approach can be followed as explained in [94] to obtain the exact values for the SM controller gains.

5.3.4 HESS Power Allocation

The objective of the HESS power allocation method is to divert the second-order harmonic ripple current component and the fast power fluctuations to the supercapacitor while the battery supplies a slow varying power component. The power variations greater than a pre-determined cut-off frequency, $\omega_{c,req}$ are considered as fast power fluctuations. Since the supercapacitor connected boost converter legs are used to control the output capacitor voltages, the second-order harmonic ripple component is supplied by the supercapacitor. In this chapter, a supercapacitor energy controller is used i) to maintain the stored energy in the supercapacitor around a reference energy level and ii) to allocate the fast power fluctuations to the supercapacitor. To achieve foregoing objectives, a PI controller is required in the SCEC as explained later. The current reference signals for the battery connected boost converter legs $i_{Lbatt1,ref}$ and $i_{Lbatt2,ref}$ are generated as,

$$i_{Lbatt1,ref} = i_{Lbatt2,ref} = \left(K_{p,Esc} + \frac{K_{i,Esc}}{s}\right) \left(E_{sc,ref} - E_{sc}\right)$$
(5.37)

where, $K_{p,Esc}$, $K_{i,Esc}$, E_{sc} , and $E_{sc,ref}$ are the proportional and integral gains of the PI controller, stored energy in the supercapacitor, and the reference energy level of the supercapacitor, respectively (See Fig. 5.5). The stored energy in the supercapacitor is given by,

$$E_{sc} = \frac{1}{2} C_{sc} v_{sc}^2 \tag{5.38}$$

where C_{sc} is the capacitance value of the supercapacitor.

In order to analyze the dynamic behavior of the HESS power allocation method, an equivalent block diagram as illustrated in Fig. 5.6 is considered. In Fig. 5.6, P_{HESS} , η , P_{tot} , P_{sc} , P_{batt} , dE_{sc} , and $E_{sc,init}$ are the output power supplied by the HESS, efficiency of the HESS, total power requirement from the battery and the supercapacitor, power supplied by the supercapacitor, power supplied by the battery, change in the supercapacitor energy, and the initial energy level of the supercapacitor, respectively. Then, the relationship between the power supplied by the battery and P_{tot} can be obtained as,

$$P_{batt} = \left(\frac{A_1s + A_2}{s^2 + A_1s + A_2}\right) P_{tot} + \left(\frac{A_1s^2 + A_2s}{s^2 + A_1s + A_2}\right) (E_{sc,ref} - E_{sc,init})$$
(5.39)

where, $A_1 = 2v_{batt}K_{p,Esc}$, $A_2 = 2v_{batt}K_{i,Esc}$ and $P_{tot} = P_{batt} + P_{sc}$. Please refer Appendix A.1 for the derivation. A transfer function between the power supplied by the battery,



Fig. 5.6: Equivalent block diagram of the HESS power allocation

 P_{batt} and the total output power of the HESS, P_{HESS} can be obtained assuming $E_{sc,init} = E_{sc,ref}$ as,

$$H_{Pbatt} = \frac{P_{batt}}{P_{HESS}} = \left(\frac{A_1 s + A_2}{s^2 + A_1 s + A_2}\right) \frac{1}{\eta}$$
(5.40)

Further, a transfer function between the power supplied by the supercapacitor, P_{sc} and P_{HESS} can be obtained as,

$$H_{Psc} = \frac{P_{sc}}{P_{HESS}} = \left(\frac{s^2}{s^2 + A_1 s + A_2}\right) \frac{1}{\eta}$$
(5.41)

Then, the transfer function between dE_{sc} and P_{HESS} can be written as,

$$H_{dEsc} = \frac{dE_{sc}}{P_{HESS}} = \left(\frac{s}{s^2 + A_1 s + A_2}\right) \frac{1}{\eta}$$
(5.42)

It can be observed that H_{Pbatt} has one zero z_1 and two poles p_1 and p_2 ,

$$z_1 = \frac{K_{i,Esc}}{K_{p,Esc}} \tag{5.43}$$

$$p_1, p_2 = -v_{batt} K_{p,Esc} \pm \sqrt{v_{batt}^2 K_{p,Esc}^2 - 2v_{batt} K_{i,Esc}}.$$
(5.44)

For real system poles,

$$K_{i,Esc} \le \frac{v_{batt} K_{p,Esc}^2}{2} \tag{5.45}$$

and further it can be shown that $p_1, p_2 < z_1$ for positive non zero $K_{p,Esc}$ and $K_{i,Esc}$ values. Using the relationships, an asymptotic Bode plot of H_{Pbatt} can be drawn as Fig. 5.7 when poles of the H_{Pbatt} are real. Using the Bode plot, the gain cross-over frequency of H_{Pbatt} , ω_c can be derived as,

$$\omega_c = \frac{|p_1| \times |p_2|}{|z_1|} = 2K_{p,Esc}v_{batt}$$
(5.46)

In order to observe the dynamic behavior of H_{Pbatt} with different PI controller parameters, a sample gain cross-over frequency of 0.05 rads⁻¹, the same gain cross-over frequency considered in Chapter 4, is considered. Figs. 5.8 (a), 5.8 (b) and 5.8 (c) illustrate the pole-zero plot of H_{Pbatt} , the step responses of H_{Pbatt} , and the step response of H_{dEsc} for the considered sample gain cross-over frequency. When $K_{i,Esc} = 0$, the system



Fig. 5.7: Asymptotic Bode plot of H_{Pbatt} with real system poles



Fig. 5.8: Effect of $K_{p,Esc}$ and $K_{i,Esc}$ on H_{Pbatt} and H_{dEsc} . (a) pole zero plot of H_{Pbatt} , (b) step response of H_{Pbatt} and (c) step response of H_{dEsc} .

is first order and the step response of H_{Pbatt} demonstrates a low pass filter characteristic. Further it can be observed from Fig. 5.8 (c) that the battery does not recharge the supercapacitor and hence, in the steady state $d_{Esc} \neq 0$. When $K_{i,Esc} < 0.5K_{p,Esc}^2 v_{batt}$, the system is second order and the system poles are real. According to the step response in Fig. 5.8 (b), the battery response shows a low pass filter characteristic and the battery supplies an additional amount of power to recharge the supercapacitor to the reference energy level. When $K_{i,Esc} = 0.5K_{p,Esc}^2 v_{batt}$, the poles of the system coincide and the system is critically damped. When $K_{i,Esc} > 0.5K_{p,Esc}^2 v_{batt}$, the poles of the system becomes imaginary and hence, an oscillatory behavior can be observed in the step response which is not desirable in the battery-supercapacitor HESS operation. It can be observed that the fastest battery response with supercapacitor recharging is occurred in the critically damped system.

In this chapter, the PI controller parameters for the SCEC are calculated to obtain the critically damped operation as follows. First, the proportional gain $K_{p,Esc}$ is calculated to achieve the required gain cross-over frequency $\omega_{c,req}$ using,

$$K_{p,Esc} = \frac{\omega_{c,req}}{2v_{batt}} \tag{5.47}$$

Then, $K_{i,Esc}$ is selected to obtain the critically damped system as,

$$K_{i,Esc} = \frac{K_{p,Esc}^2 v_{batt}}{2}.$$
 (5.48)

5.3.5 Supercapacitor Selection for the SM Controlled HESS

Using the parameters selected for the SCEC, the required capacitor value for the supercapacitor can be calculated for a given application as follows.

First, the maximum and minimum voltage limits for the supercapacitor, $v_{sc,max}$ and $v_{sc,min}$ has to be identified based on the power converter specification. Based on the

voltage limits, the utilized energy of the supercapacitor can be obtained by,

$$E_{sc,u} = 0.5C_{sc} \left(v_{sc,max}^2 - v_{sc,min}^2 \right)$$
(5.49)

The reference energy level for the supercapacitor can be selected such that equal amount of energy is available for both the supercapacitor charging and discharging as,

$$E_{sc,ref} = 0.5C_{sc}v_{sc,min}^2 + 0.5E_{sc,u}$$
(5.50)

For a considered application, the supercapacitor energy variation due to the HESS operation, dE_{sc} can be estimated from (5.42). Then, using the maximum value of the supercapacitor energy variation, $max(|dE_{sc}|)$ the required capacitance value for the supercapacitor can be calculated as,

$$C_{sc,req} = \frac{4max(|dE_{sc}|)}{v_{sc,max}^2 - v_{sc,min}^2}$$
(5.51)

5.4 Experimental Results

The performance of the SM controlled HESS is verified using a low voltage low power prototype. A 9.6Ah, 25.6V LiFePO₄ battery from K2 Energy and two 16V, 58F Maxwell supercapacitor modules connected in series are used in the prototype and the parameters of the HESS prototype are summarized in Table 5.1. The controllers are implemented on a DSpace DS1006 platform.

Operation of the battery-supercapacitor HESS when the system delivers 50W active power and 25VAr reactive power to the grid is illustrated in Figs. 5.9, 5.10, and 5.11.

The output capacitor voltages v_{o1} , v_{o2} , and the HESS output voltage v_o are illustrate in Fig. 5.9. The operation of the active and reactive power controller is illustrates in Fig. 5.10. It can be observed that the power exchange between the HESS and the grid is achieved by changing the HESS output voltage amplitude and phase angle with respect to



Fig. 5.9: The output capacitor voltage waveforms v_{o1} and v_{o2} , and the HESS output voltage v_o when the HESS delivers 50W active power and 25VAr reactive power to the grid



Fig. 5.10: The grid voltage v_g , the HESS output voltage v_o and the HESS output current i_g when the HESS delivers 50W active power and 25VAr reactive power to the grid

the grid voltage. The left hand side supercapacitor connected and the battery connected inductor current waveforms i_{Lsc1} and i_{Lbatt1} are illustrates in Fig. 5.11. As required, the supercapacitor supplies the required ripple current component while the battery supplies the required average current component.

Operation of the SM controlled battery-supercapacitor HESS with a step changing output power profile is illustrates in Fig. 5.12. In this experiment, a gain cross-over

Parameter	Value	Parameter	Value			
Power converter parameters						
Vo	60 V	$V_{DC,ref}$	60 V			
L_{batt1}, L_{batt2}	$210~\mu\mathrm{H}$	L_{sc1}, L_{sc2}	$210~\mu\mathrm{H}$			
C_1, C_2	$60 \ \mu F$	f	$50~\mathrm{Hz}$			
$v_{sc,max}$	28 V	$v_{sc,min}$	20 V			
R_{Lsc1}, R_{Lsc2}	$0.24 \ \Omega$	R_{Lbatt1}, R_{Lbatt2}	$0.24 \ \Omega$			
R_{Lg}	0.2					
Grid parameters						
V_g	60 V	L_g	$20 \mathrm{~mH}$			
Sliding mode controller parameters						
$k_{sc1,a}$	2.73×10^{-4}	$k_{sc1,b}$	8.19×10^{-5}			
$k_{sc1,c}$	$8.7 imes 10^{-5}$	k_{batt1}	3.78			
$k_{sc1,m}$	200	$k_{batt1,m}$	0.02			
f_{sw}	$20 \mathrm{~kHz}$					
PQ controller parameters						
$K_p(PI_P)$	2×10^{-6}	$\overline{K_p(PI_Q)}$	1×10^{-5}			
$K_i(PI_P)$	0.14	$K_i(PI_Q)$	4			

Table 5.1: Hybrid energy storage system parameters



Fig. 5.11: The inductor current waveforms i_{Lbatt1} and i_{Lsc1} when the HESS delivers 50W active power and 25VAr reactive power to the grid



Fig. 5.12: Experimental waveforms illustrating the operation of the HESS: Supercapacitor responds to sudden power changes.

frequency of 0.1 rad⁻¹ is considered to illustrate the performance of the system. The gain cross-over frequency, and hence the SCEC parameters, should be selected based on the specific application at the design stage of the HESS. In this experiment, the SCEC parameters $K_{p,Esc}$ and $K_{i,Esc}$ were selected as 0.00195 and 4.87×10^{-5} to obtain 0.1 rad⁻¹ gain cross-over frequency and a 29F supercapacitor is employed. It is evident that the supercapacitor responds to the sudden power variations as shown in Fig. 5.12 (d) and hence the stored energy in the supercapacitor diverts from the reference energy level as in Fig. 5.12 (e). In Fig. 5.12 (d), the double line frequency component of the supercapacitor current is filtered out to illustrate the supercapacitor response to the sudden power changes clearly. The battery current gradually increases to provide the required DC power component and the battery supplies an additional power component to recharge the

supercapacitor back to the reference energy level as illustrates in Fig. 5.12(c) and (e).

Next, the operation of the SM controlled HESS in a PV power smoothing application as discussed in Chapter 4 is considered. A 900s HESS reference power profile as shown in Fig. 5.13 (a) is generated to emulate the HESS operation in a PV power smoothing application. The SCEC parameters are selected as $K_{p,Esc} = 3.92 \times 10^{-4}$ and $K_{i,Esc} = 1.97 \times 10^{-6}$ to obtain a 0.02 rads^{-1} gain cross-over frequency. Then, the required supercapacitor size for the considered output power profile is obtained from (5.51) as 26.3F. Hence, a commercially available 29F supercapacitor is used in the experiment. Fig. 5.13 (b) illustrates the power supplied by the battery, P_{batt} the power supplied by the supercapacitor, P_{sc} and the total power supplied by the battery and the supercapacitor, P_{tot} . As expected the supercapacitor responds to the high frequency power variations while the battery supplies the slow varying power component. Furthermore, it can be observed that the HESS is able to reduce the peak battery power as well as the number of battery charge discharge cycles. The supercapacitor energy variation is depicted in Fig. 5.13 (c) and it is controlled around the reference supercapacitor energy level. From Fig. 5.13 (d) it is evident that the selected capacitance value enables the operation of the supercapacitor within the pre-defined voltage limits. The battery and the supercapacitor current waveforms are shown in Figs. 5.13 (e) and 5.13 (f) and as required the battery supplies the slow varying current while the supercapacitor supplies the fast varying current component.

In order to observe the effect of the boost inductor ESR tolerance on the HESS output



Fig. 5.13: Experimental waveforms illustrating the operation of the HESS in a PV power smoothing application.

current DC component, a 20% change in the nominal ESR value of 0.24Ω was simulated. A sample HESS output power condition of $P_{HESS}=50$ W and $Q_{HESS}=25$ VAr was considered and the boost inductor ESR values were selected as, $R_{Lsc1} = R_{Lbatt1} = 0.24 \times 0.8\Omega$ and $R_{Lsc2} = R_{Lbatt2} = 0.24 \times 1.2\Omega$. The SM controller was calculated using the nominal boost inductor ESR value. A DC component of 0.0814A which is 4.07% of the HESS rated output current was observed in the HESS output current with the DL control method. However, with the SM controller, the DC current component was only 0.0046A which is 0.23% of the HESS rated output current. According to the IEEE standards, it is required to limit the DC current injection to 0.5% of the rated current of the inverter [95]. This result demonstrates the advantage of the SM control method which is robust to parameters variation and therefore is unaffected by the ESR variation.

5.5 Conclusions

A fixed frequency SM controller for a boost inverter based single phase grid connected battery-supercapacitor HESS was presented in this chapter. The PWM based fixed frequency SM controller was employed to overcome high and variable frequency operation associated with the traditional SM control technique. Theoretical aspects of the SM controller design were discussed, and a SCEC based power allocation method was employed to allocate the fast power fluctuations to the supercapacitor without incorporating an additional filter. The power allocation parameter selection and the supercapacitor sizing were discussed for a given HESS application. The proposed control system for the boost inverter based HESS was experimentally verified using a laboratory prototype. The proposed controller was able to satisfy the HESS output power requirement while allocating the ripple current and fast power fluctuations to the supercapacitor. Moreover, the selected HESS power allocation parameters and the value of the supercapacitor enabled the operation of the HESS within the pre-determined supercapacitor voltage limits. Compared to the traditional DL controlled boost inverter based battery-supercapacitor HESS, the proposed SM controller was able to achieve better output capacitor reference tracking and hence reduce the risk of the DC current injection into the grid when the ESR values of the boost inductors become unequal due to the tolerances and temperature variations.

Chapter 6

A Rule-Based Controller to Mitigate DC-Side Second-Order Harmonic Current in a Single Phase Boost Inverter

In the HESS operation presented in the previous chapters, the second-order harmonic ripple current component was allocated to the supercapacitor and this leads to a supercapacitor lifetime reduction due to its internal heating. The objective of this chapter is to identify a suitable second-order harmonic ripple current reduction method for the boost inverter based battery-supercapacitor HESS. In order to simplify the analysis, only operation of a single boost inverter for a battery energy storage system is considered.

In this chapter, a rule-based control ripple current reduction method proposed in [67] for a boost inverter based battery energy storage system is presented. Operation of the existing waveform control ripple current reduction method is extensively analyzed. A key feature and advantage of the proposed controller compared to the waveform control ripple reduction method is its ability to reduce the ripple current amplitude in all four inverter output power operating quadrants without being affected by the capacitor tolerances and the equivalent series resistance (ESR) values of the inductors. Presented experimental results validate the performance of the proposed controller on a single-phase grid-connected DC/AC boost inverter based battery energy storage system (ESS).

6.1 Introduction

An interleaved boost inverter based single phase grid integrated battery-supercapacitor HESS was discussed in Chapter 3 to Chapter 5. The single-phase conversion from DC to AC using the boost inverter topology introduced a second-order harmonic ripple current at the DC side of the converter and hence the DC sources had to handle the ripple current component.

The effect of the low frequency current ripple on DC ESSs has been studied in the literature extensively. In [11], the authors reported low frequency ripple current can considerably degrade the lifetime of a lead acid battery due to internal heating. Furthermore, [13] reported a noticeable temperature increment in a Lithium Iron Phosphate (LFP) battery based community ESS, as a result of the low frequency ripple current component. In the power allocation methods discussed in Chapter 3 to Chapter 5, the second-order harmonic ripple current component was allocated to the supercapacitor as a way of extending the battery life time. However, [61] reported that a supercapacitor can also overheat when exposed to a continuous ripple current which in turn leads to deterioration of its lifetime.

A number of second-order harmonic ripple current reduction methods for the single-phase boost inverter topology have been presented in the technical literature [57, 74–78, 100] and can be classified into passive and active ripple current reduction methods. In the passive methods, a separate energy storage device is added to the converter to supply the second-order harmonic ripple current component while in the active methods, a modulation or control scheme is employed to reduce the ripple current component without incurring additional hardware. In [57,78], authors proposed a separate battery storage to compensate the second-order harmonic ripple current in a boost inverter based fuel cell ESS. However such method required additional hardware and cost and also it leads to a lifetime reduction in the battery. In the case of active methods, a waveform control method was studied in [74-77, 100] for a boost inverter based ESSs and in [74] authors reported a noticeable efficiency increase in the ESS due to the second-order harmonic ripple current reduction. However, the method has to be redesigned every time the load changes. A current feedback method for the boost-inverter waveform controller to deal with the changing load conditions was proposed in [75]. However, only the unity power factor operation was considered limiting its applicability for a boost-inverter ESS which is required to operate in all four output power quadrants. The waveform control method was extended to all possible output power conditions in [76]. In [74–77, 100], the compensation signal parameters were calculated using an ideal boost inverter model.

Hence, the effectiveness of the second-order harmonic ripple reduction was affected by the capacitance tolerance of the inverter output capacitors. In addition, the waveform control method cannot compensate for the effect of the ESR values of the inductors on the ripple. Also, the waveform control method in [74–77, 100] is not directly applicable to a grid connected boost inverter since in the grid connected boost inverter the output power control is achieved by changing the inverter output voltage amplitude and phase angle with respect to the grid voltage.

In this chapter, the waveform control ripple current reduction method is extended to the grid connected operation and the effects of the capacitor tolerances and the ESR values of the inductors on its effectiveness are analyzed. Then, a rule-based controller is proposed to reduce the second-order harmonic ripple current in the boost inverter based ESS. The main advantage of the proposed controller is that it can reduce the second-order harmonic ripple current in all four output power quadrants without being affected by the capacitor tolerances and the ESR values of the inductors. The effectiveness of the proposed controller is experimentally verified on a single-phase grid-connected DC/AC boost inverter based battery ESS prototype and it is shown that its performance is superior when compared with the waveform control method.

6.2 Second-Order Harmonic Ripple Current Analysis and Reduction Using the Waveform Control Method

6.2.1 Second-Order Harmonic Input Current Component in the Boost Inverter

The configuration of a single phase grid connected boost inverter is shown in Fig. 6.1 and the operation of the boost inverter was discussed in Chapter 2. The voltages across the output capacitors C_1 and C_2 are controlled to follow voltage references given by,

$$v_{o1,ref} = V_{DC,ref} + \frac{V_o}{2}\sin(\omega t + \delta)$$
(6.1)

and

$$v_{o2,ref} = V_{DC,ref} - \frac{V_o}{2}\sin(\omega t + \delta).$$
(6.2)

respectively. V_o , ω , and δ are the inverter output voltage amplitude, frequency and phase angle respectively. The boost inverter output current i_o is given by,

$$i_o = I_o \sin(\omega t + \alpha) \tag{6.3}$$

where α is the phase angle of the inverter output current with respect to the grid voltage. The output capacitor currents i_{C1} and i_{C2} are,



Fig. 6.1: Single phase grid connected boost inverter

$$i_{C1} = C_1 \omega \frac{V_o}{2} \cos(\omega t + \delta) \tag{6.4}$$

$$i_{C2} = -C_2 \omega \frac{V_o}{2} \cos(\omega t + \delta) \tag{6.5}$$

From (6.4) and (6.5), the capacitor current waveforms are in fundamental frequency. Using (6.1)-(6.5),

$$i_1 = I_o \sin(\omega t + \alpha) + C_1 \omega \frac{V_o}{2} \cos(\omega t + \delta)$$
(6.6)

Similarly, i_2 can be written as,

$$i_2 = -I_o \sin(\omega t + \alpha) - C_2 \omega \frac{V_o}{2} \cos(\omega t + \delta)$$
(6.7)

Then, the total input current to the boost inverter can be obtained as,

$$i_{in} = i_1 \frac{v_{o1}}{v_{in}} + i_2 \frac{v_{o2}}{v_{in}} \tag{6.8}$$

Then,

$$i_{in} = \frac{1}{v_{in}} \left[\frac{V_o I_o}{2} \cos(\alpha - \delta) + \frac{V_o}{2} \sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C \omega \sin(\alpha - \delta) \right]} \cos(2\omega t - \theta) \right]$$
(6.9)

where

$$\cos\theta = \frac{-I_o\cos(\alpha+\delta) + \frac{C\omega V_o\sin(2\delta)}{2}}{\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha-\delta)\right]}}$$

and $C = C_1 = C_2$. A detailed analysis when $C_1 \neq C_2$ can be found in [101]. The magnitude and phase angle of the second-order harmonic ripple component depend on the

inverter output current and the phase angle, and hence depend on the inverter output power. The inductor current waveforms i_{L1} and i_{L2} can be obtained using

$$i_{L1} = i_1 \frac{v_{o1}}{v_{in}} \tag{6.10}$$

and

$$i_{L2} = i_2 \frac{v_{o2}}{v_{in}} \tag{6.11}$$

By substituting (6.1), (6.2), (6.6), and (6.7) to (6.10) and (6.11), the AC components of the inductor currents i_{L1} and i_{L2} can be obtained as

$$i_{L1,AC} = A_{\omega,n} \cos(\omega t + \lambda_{\omega,n}) + A_{2\omega,n} \cos(2\omega t + \lambda_{2\omega,n})$$
(6.12)

$$i_{L2,AC} = -A_{\omega,n}\cos(\omega t + \lambda_{\omega,n}) + A_{2\omega,n}\cos(2\omega t + \lambda_{2\omega,n})$$
(6.13)

where,

$$\begin{split} A_{\omega,n} &= \frac{1}{v_{in}} \sqrt{(I_o V_{DC})^2 + \left(\frac{C\omega V_o V_{DC}}{2}\right)^2 + I_o V_{DC}^2 V_o C\omega \sin(\alpha - \delta)} \\ A_{2\omega,n} &= \frac{V_o}{4v_{in}} \sqrt{I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C\omega \sin(\alpha - \delta)} \\ \cos \lambda_{\omega,n} &= \frac{I_o V_{DC} \sin \alpha + \left(\frac{C\omega V_o V_{DC} \cos \delta}{2}\right)}{\sqrt{(I_o V_{DC})^2 + \left(\frac{C\omega V_o V_{DC}}{2}\right)^2 + I_o V_{DC}^2 V_o C\omega \sin(\alpha - \delta)}} \\ \cos \lambda_{2\omega,n} &= \frac{-I_o \cos(\alpha + \delta) + \left(\frac{C\omega V_o \sin(2\delta)}{2}\right)}{\sqrt{I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C\omega \sin(\alpha - \delta)}}. \end{split}$$

From (6.4), (6.5), (6.12) and (6.13), it is clear that the second-order harmonic current components flow only through the inductors and not through the output capacitors [74].

6.2.2 Second-Order Harmonic Input Current Ripple Reduction Using the Waveform Control Method

In [74], a waveform control method was proposed to reduce the second-order harmonic input current ripple of a boost inverter which was operated at a unity power factor. Operation of the waveform control method was extended for all possible power factor conditions in [76]. In this paper, the analysis of the waveform control method is extended to a grid connected boost inverter where the inverter output power is controlled by modifying the inverter output voltage amplitude and phase angle. Since the boost inverter is a differential inverter, the output capacitor voltages v_{o1} and v_{o2} can be modified as in (6.14) and (6.15), without affecting the required inverter output voltage.

$$v_{o1} = V_{DC,ref} + \frac{V_o}{2}\sin(\omega t + \delta) + B\sin(2\omega t + \varphi)$$
(6.14)

$$v_{o2} = V_{DC,ref} - \frac{V_o}{2}\sin(\omega t + \delta) + B\sin(2\omega t + \varphi)$$
(6.15)

where B is the amplitude of the second-order harmonic voltage component and φ is the phase angle of the second-order harmonic voltage component with respect to the grid voltage.

Then, the output capacitor currents can be written as,

$$i_{C1} = C\omega \frac{V_o}{2} \cos(\omega t + \delta) + 2BC\omega \sin(2\omega t + \varphi)$$
(6.16)

$$i_{C2} = -C\omega \frac{V_o}{2}\cos(\omega t + \delta) + 2BC\omega\sin(2\omega t + \varphi)$$
(6.17)

The total input current with the modified output capacitor voltages can be obtained by using (6.14), (6.15), (6.16), (6.17) and (6.8) as

$$i_{in} = \frac{V_o I_o}{2v_{in}} \cos(\alpha - \delta) + \frac{V_o}{2v_{in}} \sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C \omega \sin(\alpha - \delta)\right]} \cos(2\omega t - \theta) + \frac{4\omega B C V_{DC} \cos(2\omega t + \varphi)}{v_{in}} + \frac{2\omega B^2 C \sin(4\omega t + 2\varphi)}{v_{in}}.$$
(6.18)

The first term of (6.18) is the DC component of the input current which corresponds to the inverter output power. The second term is the uncompensated second-order harmonic ripple current component of the boost inverter. The second-order harmonic ripple current component due to the second-order harmonic output capacitor voltage reference component is given by the third term of the equation. It can be observed that due to the modification of the output capacitor reference voltages, a fourth-order harmonic ripple component is also present in the input current (the fourth term in (6.18)). By selecting B and φ such that,

$$\frac{V_o}{2}\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha - \delta)\right]\cos(2\omega t - \theta) + 4\omega BC V_{DC}\cos(2\omega t + \varphi)} = 0$$
(6.19)

the second-order harmonic ripple component of the input current can be theoretically eliminated. For that, the magnitude of the two terms in (6.19) should be equal and the two terms should have 180° phase shift. The parameters B and φ which satisfy (6.19) are given by,

$$B = \frac{V_o}{8V_{DC}\omega C} \sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega \sin(\alpha - \delta)\right]}$$
(6.20)

and

$$\cos(\varphi) = \frac{I_o \cos(\delta + \alpha) - \frac{C\omega V_o}{2} \sin(2\delta)}{\sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C\omega \sin(\alpha - \delta)\right]}}.$$
(6.21)

6.2.3 Effect of the Waveform Control Method on the Inductor Current Harmonic Content and Efficiency

With the waveform control method, the current waveforms i_1 and i_2 is given by

$$i_{C1} = I_o \sin(\omega t + \alpha) + C\omega \frac{V_o}{2} \cos(\omega t + \delta) + 2BC\omega \sin(2\omega t + \varphi)$$
(6.22)

$$i_{C1} = -I_o \sin(\omega t + \alpha) - C\omega \frac{V_o}{2} \cos(\omega t + \delta) + 2BC\omega \sin(2\omega t + \varphi)$$
(6.23)

Then, substituting (6.14), (6.15), (6.22), and (6.23) in to (6.10) and (6.11) and simplifying, the AC components of the inductor currents i_{L1} and i_{L2} can be obtained as,

$$i_{L1,AC} = A_{\omega,n} \cos(\omega t + \lambda_{\omega,n}) + A_{\omega,w} \cos(\omega t + \lambda_{\omega,w}) + A_{2\omega,n} \cos(2\omega t + \lambda_{2\omega,n}) + A_{2\omega,w} \cos(2\omega t + \lambda_{2\omega,w}) + A_{3\omega,w} \cos(3\omega t + \lambda_{3\omega,w}) + A_{4\omega,w} \cos(4\omega t + \lambda_{4\omega,w})$$
(6.24)

$$i_{L2,AC} = -A_{\omega,n}\cos(\omega t + \lambda_{\omega,n}) - A_{\omega,w}\cos(\omega t + \lambda_{\omega,w}) + A_{2\omega,n}\cos(2\omega t + \lambda_{2\omega,n}) + A_{2\omega,w}\cos(2\omega t + \lambda_{2\omega,w}) - A_{3\omega,w}\cos(3\omega t + \lambda_{3\omega,w}) + A_{4\omega,w}\cos(4\omega t + \lambda_{4\omega,w})$$
(6.25)

where

$$A_{\omega,w} = \frac{1}{v_{in}} \sqrt{\left(\frac{BI_o}{2}\right)^2 + \left(\frac{BC\omega V_o}{4}\right)^2 - \frac{B^2 V_o I_o C\omega \sin(\alpha - \delta)}{4}}{A_{2\omega,w}} = \frac{2BC\omega V_{DC}}{v_{in}}$$

$$A_{3\omega,w} = \frac{1}{v_{in}} \sqrt{\left(\frac{BI_o}{2}\right)^2 + \left(\frac{3BC\omega V_o}{4}\right)^2 + \frac{3B^2 V_o I_o C\omega \sin(\alpha - \delta)}{4}}{A_{4\omega,w}} = \frac{B^2 C\omega}{v_{in}}$$

$$\cos \lambda_{\omega,w} = \frac{\frac{BI_o}{2} \cos(\varphi - \alpha) - \frac{BV_o C\omega}{4} \sin(\varphi - \delta)}{\sqrt{\left(\frac{BI_o}{2}\right)^2 + \left(\frac{BC\omega V_o}{4}\right)^2 - \frac{B^2 V_o I_o C\omega \sin(\alpha - \delta)}{4}}}{\cos \lambda_{2\omega,w} = \cos \varphi}$$

$$\cos \lambda_{3\omega,w} = \frac{\frac{3BV_o C\omega}{4} \sin(\varphi + \delta) - \frac{BI_o}{2} \cos(\alpha + \varphi)}{\sqrt{\left(\frac{BI_o}{2}\right)^2 + \left(\frac{3BC\omega V_o}{4}\right)^2 + \frac{3B^2 V_o I_o C\omega \sin(\alpha - \delta)}{4}}}$$

and

$$\cos \lambda_{4\omega,w} = \cos 2\varphi.$$

Additionally, $A_{\omega,n}$ and $A_{2\omega,n}$ are as defined in (6.12) and (6.13). In (6.24) and (6.25), the subscripts *n* and *w* denote the amplitude and phase angle components during the normal operation (without the waveform control) and the operation with the waveform control method respectively. The second-order harmonic components of the inductor currents (6.24) and (6.25) can be re-written as,

$$i_{L1,2,2\omega} = \frac{V_o}{4} \sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C \omega \sin(\alpha - \delta)\right]} \cos(2\omega t - \theta) + 2\omega B C V_{DC} \cos(2\omega t + \varphi)$$

$$\tag{6.26}$$

Then, by selecting B and φ as in (6.20) and (6.21), the second-order harmonic components of the inductor currents can be theoretically eliminated and will only flow through the output capacitors as evident from (6.16) and (6.17).

The frequency content of the inductor current without and with the waveform control method can be obtained from (6.12) and (6.24) using the prototype parameters given in Table 6.1. Fig. 6.2 illustrates the calculated AC components of the boost inverter inductor current using (6.12) and (6.24) when the boost inverter supplies 10W active power and 15VAr reactive power to the grid. With the waveform control, the fundamental frequency component shows a slight reduction, the second-order harmonic component is completely eliminated and the third-order and the fourth-order harmonic current is eliminated from the inductor and allocated to the output capacitor, the power loss in the inductor is reduced which in turn increases the converter efficiency as presented in [74, 102].



Fig. 6.2: AC components of the inductor current i_{L1} , without and with the waveform control method when delivering 10W active power and 15 VAr reactive power to the grid.

6.2.4 Selection of the Output Capacitor Capacitance C

From the analysis shown in Subsection 6.2.3 it is obvious that due to the waveform control method, the output capacitors of the boost inverter have to supply the second-order harmonic power variation as opposed to the normal boost inverter operation in which the second-order harmonic power variation is supplied by the input power source. In the following paragraphs this effect is analyzed and it is shown how it affects selection of the required output capacitor capacitance as compared with the selection of the capacitance for the normal operation.

During operation with the waveform control the maximum value of B, B_{max} , can be found from (6.20) for the maximum output current $I_{o,max}$ as,

$$B_{max} = \frac{V_o}{8V_{DC}\omega C} \sqrt{\left[(I_{o,max})^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_{o,max} C\omega \right]}$$
(6.27)

Then, the maximum and minimum values of the output capacitor voltages are,

$$V_{o1,max} = V_{DC} + \frac{V_o}{2} + B_{max}$$
(6.28)

$$V_{o1,min} = V_{DC} - \frac{V_o}{2} - B_{max}$$
(6.29)

In addition, V_{DC} has to satisfy the inequality

$$V_{DC} > v_{in} + \frac{V_o}{2} + B_{max}$$
(6.30)

From (6.28) it is clear that the higher the V_{DC} the higher the amplitude of the maximum output capacitor reference voltage $V_{o1,max}$. Since a low value of V_{DC} is preferred to reduce the gain of the boost inverter, a minimum possible value of V_{DC} , $V_{DC,min}$, which satisfies (6.30) can be obtained from (6.27) and (6.30) as,

$$V_{DC,min} = \frac{1}{2} \left[v_{in} + \frac{V_o}{2} + \sqrt{\left(v_{in} + \frac{V_o}{2} \right)^2 + \left(\frac{V_o}{2\omega C} \sqrt{\left[(I_{o,max})^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_{o,max} C \omega \right]} \right)}$$
(6.31)

Substituting $V_{DC} = V_{DC,min}$ in (6.28), the maximum value of the output capacitor voltages, $V_{o1.max}$ is,

$$V_{o1,max} = V_{DC,min} + \frac{V_o}{2} + B_{max}$$
(6.32)

The relationship of $V_{DC,min}$, B_{max} , and $V_{o1,max}$ versus the output capacitance value Cfor $I_{o,max}=1.5$ A is shown in Fig. 6.3. From Fig. 6.3, it can be observed that higher C leads to lower B_{max} , $V_{DC,min}$, and $V_{o1,max}$. Hence, the required output capacitor capacitance value $C_{req,w}$ can be found as follows. First, the maximum allowable output capacitor voltage $V_{o1,max}$ is predetermined from the input voltage v_{in} and the maximum gain of the converter legs G_{max} , as $V_{o1,max} = v_{in} \times G_{max}$. Next, using (6.27), (6.31), and (6.32), values of B_{max} , $V_{DC,min}$, and $V_{o1,max}$ are plotted against a range of the output capacitor C values. Then, for the predetermined $V_{o1,max}$ value both the required output capacitor capacitance value $C_{req,w}$ and the required output capacitor reference voltage minimum DC shift $V_{DC,min}$ can be obtained graphically as illustrated in Fig. 6.3.

When considering the normal operation (without the waveform control method) the DC shift voltage V_{DC} has to satisfy the inequality

$$V_{DC} > v_{in} + \frac{V_o}{2} \tag{6.33}$$

Hence, the minimum possible DC shift, $V_{DC,min}$ is

$$V_{DC,min} = v_{in} + \frac{V_o}{2} \tag{6.34}$$

and the maximum output capacitor voltage, $V_{o1,max}$ is

$$V_{o1,max} = V_{DC} + \frac{V_o}{2}$$
(6.35)

Form (6.34) and (6.35), it is clear that $V_{DC,min}$ and $V_{o1,max}$ are independent of the output capacitor value C. Therefore, during the normal boost inverter operation, the required output capacitor value $C_{req,n}$ can be found from a selected maximum switching frequency voltage ripple in the output capacitor voltage waveform $\Delta V_{o1,max}$ which is given by

$$\Delta V_{o1,max} = \frac{I_{o,max} d_{1,max}}{f_{sw} C_{req,n}} \tag{6.36}$$



Fig. 6.3: Relationships between B_{max} , $v_{o1,max}$ and $V_{DC,min}$ versus the output capacitor C values during operation with the waveform control.

where

$$d_{1,max} = \frac{V_{o1,max} - v_{in}}{V_{o1,max}} \tag{6.37}$$

assuming a lossless conversion. From (6.36) and (6.37), the required output capacitor value $C_{req,n}$ during the normal operation is

$$C_{req,n} = \frac{I_{o,max}}{f_{sw}\Delta V_{o1,max}} \frac{V_{o1,max} - v_{in}}{V_{o1,max}}.$$
(6.38)

The following example illustrates selection of the required output capacitor values $C_{req,n}$ and $C_{req,w}$ for the normal operation and operation with the waveform control, respectively.

Consider a boost inverter with the input voltage v_{in} , the output voltage peak value V_o and the maximum peak output current $I_{o,max}$ equal to 12.8V, 40V, and 1.5A, respectively. Assume the switching frequency of the converter is 20kHz and the maximum gain of the boost converter leg G_{max} is 6. Then, according to (6.35) the maximum output capacitor voltage $V_{o1,max}$ for the normal operation is 52.8V which corresponds to the converter leg gain G equal to 4.125. If the required output capacitor switching frequency ripple $\Delta V_{o1,max}$ is less than 5% of the peak output voltage then the required output capacitor value $C_{req,n}$ for the normal operation can be calculated from (6.38) as 28.4 μ F.

If the value of $C_{req,n}$ found for the normal operation is used with the waveform control method, then the required peak output capacitor voltage $V_{o1,max}$ can be obtained from Fig. 6.3 as 89.56V. However, to achieve this voltage, the converter leg gain G has to be equal to 7. Since this gain exceeds the maximum possible gain G_{max} , the $C_{req,n}$ value cannot be used with the waveform control method and the $C_{req,w}$ value has to be found as follows. Considering the maximum gain of the boost converter leg $G_{max}=6$, the maximum possible output capacitor voltage $V_{o1,max}$ is 76.8V. Then, the required output capacitor value $C_{req,w}$ to maintain the output capacitor peak voltage less than 76.8V can be obtained from Fig. 6.3 as $C_{req,w} = 54.6 \ \mu\text{F}$. The nearest commercially available value of $60\ \mu\text{F}$ is then selected in the actual setup.

6.2.5 Effect of the Capacitor Tolerances and the Boost Inductor ESR Values on the Waveform Control Method Performance

In the waveform control method, (6.20) and (6.21) are used to calculate the required second-order harmonic output capacitor voltage reference signal to reduce the second-order harmonic input ripple current. However, in the derivation of (6.20) and (6.21), the ESR values of the boost inverter inductors are neglected for simplicity. Moreover, since the second-order harmonic voltage reference amplitude B and phase angle φ are functions of the capacitance of the output capacitor, the effect of the output capacitor tolerances on the ripple reduction using the waveform control method has to be investigated. In order to analyze the effect of the ESR values of the inductors and the capacitor tolerances on the total input current the inductor currents of the boost inverter can be written as,

$$i_{L1} = \frac{i_1 v_{o1}}{v_{in} - i_{L1} R_{L1}}, i_{L2} = \frac{i_2 v_{o2}}{v_{in} - i_{L2} R_{L2}}$$
(6.39)

Then, the total input current can be obtained from

$$i_{in} = \frac{v_{in} - \sqrt{\left(v_{in}^2 - 4R_L v_{o1} i_1\right)}}{2R_L} + \frac{v_{in} - \sqrt{\left(v_{in}^2 - 4R_L v_{o2} i_2\right)}}{2R_L}$$
(6.40)

where $R_L = R_{L1} = R_{L2}$. Please refer Appendix A.1 for the derivation.

The effect of the ESR values of the inductors and tolerance of the capacitors on the total input current can be observed by simulating (6.40). Inverter parameters equal to the prototype parameters summarized in Table 6.1 are used in the analysis. First, the second-order harmonic output capacitor voltage reference parameters are calculated from (6.20) and (6.21) using $C=60\mu$ F. Then, (6.40) is simulated for various ESR values and considering a $\pm 10\%$ change in C. A ripple reduction factor is defined as the ratio between the ripple current amplitude without and with the waveform control method. Fig. 6.4 shows the ripple current reduction factor for various values of the inductor internal resistance and $\pm 10\%$ change in the capacitance of the output capacitors when the inverter supplies 10W active power and 15VAr reactive power. Theoretically, a perfect ripple cancellation should result in an infinite ripple reduction factor. However, as it can be observed from Fig. 6.4, the capacitor tolerances and the inductor ESR values cause a significant performance degradation in the waveform control method because it is based on direct calculation of the second-order harmonic reference voltage parameters. Hence, a control method which is able to generate the required second-order harmonic voltage reference signal parameters without calculating them from (6.20) and (6.21) can avoid the problem.

6.3 The Proposed Rule Based Controller for Ripple Current Mitigation

A rule-based controller is proposed to mitigate the second-order harmonic input ripple current component by adjusting the output capacitor second-order harmonic voltage reference amplitude B and phase angle φ . The main advantage of the proposed controller is that it can reduce the second-order harmonic ripple current in all four output power quadrants without being affected by the capacitor tolerances and the ESR values of the inductors.



Fig. 6.4: Ripple reduction factor for various values of the inductor ESR values R_L and capacitance C of the output capacitors when the waveform control method is used (inverter supplies 10W active power and 15VAr reactive power).

Variation of the second-order harmonic ripple current amplitude as a function of Band φ is analyzed for all four operating power quadrants to verify that it contains a global minimum. The results in Fig. 6.5 illustrate that the function contains only a global minimum which corresponds to the B and φ values given by (6.20) and (6.21), respectively.

The proposed rule-based controller incorporates a perturb and observe (P&O) approach [103,104] to adjust the B and φ values. The flow chart of the proposed controller is shown in Fig. 6.6. Initially, both B and φ values are set to zero and the second-order harmonic ripple current amplitude A is measured as presented later in the section. The controller operates in two phases. In the first phase, the value of B is either increased or decreased to reduce the ripple current component and in the second phase, the value of φ is either increased or decreased to reduce the ripple current component (ϵ) or decrement ($-\epsilon$) of the ripple current amplitude cannot be obtained by increasing or decreasing either the value of B or φ (see Fig. 6.6).

Variable step size perturbations are used in the proposed controller algorithm in order to avoid design issues related to the fixed step size perturbations [103]. The variable step size perturbation method used for B is given by,

$$B_k = B_{k-1} + N_B A_{k-1} \tag{6.41}$$

where N_B is a scalar which is tuned at the design time of the controller and A is the measured second-order harmonic input current amplitude. Larger N_B results in faster dynamics with increased transition oscillations while smaller N_B results in slower dynamics with smaller transition oscillations. Since the controller incorporates a P&O method, an



Fig. 6.5: Second-order harmonic ripple current amplitude as a function of B and φ when (a) inverter supplies 15W active power and 10VAr reactive power,(b) inverter supplies 15W active power and absorbs 10VAr reactive power, (c) inverter absorbs 15W active power and supplies 10VAr reactive power and (d) inverter absorbs 15W active power and 10VAr reactive power

increment of the second-order harmonic ripple current amplitude may occur due to the initial perturbation of B. N_B can be used to limit the initial ripple current amplitude change ΔA_{init} to a maximum allowable limit $\Delta A_{init,max}$. From (6.18), the maximum possible change of the second-order harmonic ripple current amplitude ΔA_{max} due to a step change in B can be written as,

$$\Delta A_{max} = \frac{4C\omega V_{DC}\Delta B_{max}}{v_{in}} \tag{6.42}$$

where ΔB_{max} is the maximum possible step change in B.

Hence, using (6.41) and (6.42), the value of N_B can be calculated to limit the initial step change of the ripple current amplitude ΔA_{init} due to the initial perturbation of B as,

$$N_B = \left(\frac{v_{in}}{4\omega C V_{DC}}\right) \frac{\Delta A_{init,max}}{A_{max}} \tag{6.43}$$

where $\Delta A_{init,max}$ is the maximum allowable initial step change of the ripple current



Fig. 6.6: Flow diagram of the proposed rule-based controller for the second-order harmonic ripple current reduction

amplitude and A_{max} is the maximum possible uncompensated second-order harmonic ripple current amplitude which is given by,

$$A_{max} = \frac{V_o}{2v_{in}} \sqrt{\left[(I_{o,max})^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_{o,max} C \omega \right]}$$
(6.44)

The variable step size perturbation method used for φ is given by,

$$\varphi_k = \varphi_{k-1} + N_{\varphi} A_{k-1} \tag{6.45}$$

where N_{φ} is a scalar which is calculated for a maximum allowable phase-angle step change

 $\Delta \varphi_{max}$, by,

$$N_{\varphi} = \frac{\Delta \varphi_{max}}{A_{max}} \tag{6.46}$$

There has to be a specific time delay T_D between each perturbation and the ripple amplitude measurement instant. The value of the required time delay T_D depends on the converter response time to a step change in the voltage reference and the time required to measure the second-order harmonic ripple amplitude accurately. The faster the measurement of the second-order harmonic ripple current amplitude, the faster the dynamic performance of the controller. Hence, a Fast Fourier transformation (FFT) based amplitude measurement technique [105] is employed to ensure fast and accurate measurement. Fig. 6.7 shows the block diagram of the second-order harmonic ripple current amplitude measurement system. From (6.18), the total input current of the boost inverter with modified output capacitor reference signals consists of a DC component, a second-order harmonic component, and a fourth-order harmonic component. The total boost inverter input current in (6.18) can be re-written as,

$$i_{in} = I_{in,DC} + I_{in,2\omega} \cos(2\omega t + \gamma_{2\omega}) + I_{in,4\omega} \cos(4\omega t + \gamma_{4\omega})$$
(6.47)

The following equations can be obtained after multiplying (6.47) with the reference signals, $\cos(2\omega t)$ and $\sin(2\omega t)$.

$$i_{in}\cos(2\omega t) = I_{in,DC}\cos(2\omega t) + I_{in,2\omega} \left[\frac{\cos(\gamma_{2\omega}) + \cos(4\omega t + \gamma_{2\omega})}{2} \right] + I_{in,4\omega} \left[\frac{\cos(2\omega t + \gamma_{4\omega}) + \cos(6\omega t + \gamma_{4\omega})}{2} \right]$$
(6.48)
$$i_{in}\sin(2\omega t) = I_{in,DC}\sin(2\omega t) + I_{in,2\omega} \left[\frac{\sin(\gamma_{2\omega}) + \sin(4\omega t + \gamma_{2\omega})}{2} \right] + I_{in,4\omega} \left[\frac{\sin(2\omega t + \gamma_{4\omega}) + \sin(6\omega t + \gamma_{4\omega})}{2} \right]$$
(6.49)

By averaging (6.48) and (6.49) over a period of the second-order harmonic waveform (T_{avg}) ,



Fig. 6.7: Second order harmonic ripple current amplitude measurement technique

the following equations can be obtained.

$$i_{in}\cos(2\omega t)_{avg} = \frac{1}{T_{avg}} \int_0^{T_{avg}} i_{in}\cos(2\omega t)dt$$

$$= \frac{I_{in,2\omega}\cos(\gamma_{2\omega})}{2}$$
(6.50)

$$i_{in}\sin(2\omega t)_{avg} = \frac{1}{T_{avg}} \int_0^{T_{avg}} i_{in}\sin(2\omega t)dt$$

= $\frac{I_{in,2\omega}\sin(\gamma_{2\omega})}{2}$ (6.51)

Then, the second-order harmonic ripple current amplitude can be obtained as,

$$A = 2\sqrt{(i_{in}\cos(2\omega t)_{avg})^2 + (i_{in}\sin(2\omega t)_{avg})^2}$$
(6.52)

The method requires T_{avg} time to accurately measure the ripple current amplitude. Hence, for the proper operation of the rule-based controller, the delay time between the perturbations and the measurements T_D has to be selected such that,

$$T_D \ge T_{avg} \tag{6.53}$$

To avoid possible steady state oscillations, the controller should be disabled after a pre-determined number of iterations. The number of iterations can be determined based on the required ripple reduction performance, and in the experiments, the controller was disabled after four iterations.

6.4 Boost Inverter Control System

The overall block diagram of the grid-connected boost inverter with the proposed rule-based controller is shown in Fig. 6.8. The parameters of the experimental prototype are summarized in Table 6.1.

Two DL controllers are implemented to control the output capacitor voltages v_{o1} and v_{o2} as given by (6.14) and (6.15) respectively. Each DL controller has an inner current control loop and an outer voltage control loop. The block diagram of the DL controller for the left hand side boost converter leg is shown in Fig. 6.9. A proportional resonant (PR) controller with two resonant components which are tuned to 50 Hz and 100Hz is employed in the voltage control loop to achieve better sinusoidal reference following [106]. The transfer function of the PR controller is given by,

$$H_{PR}(s) = K_p + \frac{2K_{i1}\omega_{c1}s}{s^2 + 2\omega_{c1}s + \omega_1^2} + \frac{2K_{i2}\omega_{c2}s}{s^2 + 2\omega_{c2}s + \omega_2^2}$$
(6.54)

where K_p , K_{i1} , and K_{i2} , are gains of the PR controller. The bandwidth of the resonant components can be adjusted using ω_{c1} and ω_{c2} . The parameters ω_1 and ω_2 are selected





Parameter	Value	Parameter	Value			
Power converter parameters						
V_o	40 V	f	$50~\mathrm{Hz}$			
C_{1}, C_{2}	$60 \ \mu F$	L_1, L_2	$210~\mu\mathrm{H}$			
$I_{o,max}$	$1.5 \mathrm{A}$	T_F	0.001			
V_{DC}	$42 \mathrm{V}$	f_{sw}	$20 \mathrm{~kHz}$			
R_{L1}, R_{L2}	$0.24~\Omega$					
Grid parameters						
V_g	40 V	L_g	$20 \mathrm{~mH}$			
Voltage control loop parameters						
$K_{p,PR}$	0.1	$K_{i1,PR}$	4			
$K_{i2,PR}$	6	ω_{c1},ω_{c2}	0.5			
Current control loop parameters						
$K_{p,PI}$	2.615	$K_{i,PI}$	8.44×10^{-5}			
Rule based controlletr parameters						
N_B	1.1	N_{arphi}	0.25			
T_{avg}	$0.01 \mathrm{~s}$	T_D	$0.03~{ m s}$			
ϵ	0.01					

Table 6.1:	Experimental	$\operatorname{prototype}$	parameters
------------	--------------	----------------------------	------------

as 314.2 rads⁻¹ and 628.3 rads⁻¹ respectively. The PI controller based inner current control loops are designed for each boost converter leg using the average continuous time model of the boost converter. The PR controllers for the outer voltage control loops are designed with 400Hz bandwidth and the PI controllers for the inner current control loops are designed with 4kHz bandwidth [55].

The maximum possible uncompensated second-order harmonic input current amplitude A_{max} can be calculated using (6.44). The scalar N_B for the rule-based controller is calculated such that the maximum initial step change of the second-order harmonic ripple component is 25% of the maximum possible uncompensated second-order harmonic input current amplitude A_{max} . The scalar N_{φ} for the rule-based controller is selected such that maximum possible step change of the phase angle φ is 0.75rads⁻¹. A large value for ϵ can reduce the ripple reduction performance of the controller while a very small ϵ can lead to slower dynamics. Hence ϵ was selected as 0.01 to achieve better ripple reduction with faster dynamics. All controller parameters for the experimental prototype are summarized in Table 6.1.

6.5 Experimental Results

The proposed rule-based controller for the single-phase grid-connected boost inverter was verified using an experimental prototype. A 6.4 Ah, 12.8 V LiFePO₄ battery from K2 Energy was used as the input power source. Fig. 6.10 shows the operation of the grid-connected boost inverter without and with the proposed ripple current reduction method. The inverter is controlled to deliver 10W active power and 15VAr reactive power to the grid. Figs. 6.10(a) and 6.10(b) show the waveforms of the output capacitor voltages and the inverter output voltage, without and with the rule-based control respectively. As expected, the output voltage waveform is the same despite the difference in the output capacitor voltage waveforms. Figs. 6.10(c) and 6.10(d) depict the inverter output voltage, grid voltage and the output current waveforms. Both the active and reactive power supplied to the grid are controlled by changing the amplitude and phase angle of the inverter output voltage with respect to the grid voltage. The inductor current waveforms and the total input current waveforms of the boost inverter without and with the rule-base control are shown in the Figs. 6.10(e) and 6.10(f), respectively. Significant reduction in the DC-side current ripple component can be observed after applying the rule-based controller compared to the conventional operation of the boost inverter.

The frequency spectrum characteristics of the boost inverter input current are shown in Figs. 6.10(g) and 6.10(h) for both cases. The second-order harmonic component amplitude with conventional operation is 1.174A while the second-order harmonic current component amplitude with the rule-based controller is reduced to 0.156A achieving close to 7.5 times reduction in the second order harmonic ripple current amplitude. The fourth-order harmonic current component with the proposed controller is 32mA while it is 11mA with the conventional operation. The increment in the fourth-order harmonic current component is as expected due to modification in the output capacitor reference voltages. However, the effect of the increment in the fourth-order harmonic current component is negligible compared to the reduction in the second order harmonic input current component.

Dynamic performance of the rule-based controller is shown in Fig. 6.11. The controller is enabled at the time 1s. First the value of B is adjusted to reduce the second-order harmonic ripple current amplitude followed by a phase angle φ adjustment. After several iterations, significant ripple current reduction is achieved. To avoid steady-state oscillations, the controller is disabled after four cycles.

Performance of the proposed controller for a step change in the output active and reactive power is shown in Fig. 6.12. The rule-based controller is reactivated each time a



Fig. 6.10: Experimental results illustrating operation of the battery storage system when delivering 15VAr reactive power and 10W active power. (a) Output capacitor voltage waveforms and inverter output voltage without the rule-based controller, (b) Output capacitor voltage waveforms and inverter output voltage with the rule-based controller, (c) Inverter output voltage, grid voltage and inverter output current without the rule-based controller, (d) Inverter output voltage, grid voltage and inverter output current with the rule-based controller, (e) Inductor currents and total input current of the boost inverter without the rule-based controller, (f) Inductor currents and total input current of the boost inverter with the rule-based controller, (g) Frequency characteristics of the input current without the rule-based controller and (h) Frequency characteristics of the input current with the rule-based controller.

change in the reference active or reactive power P_{ref} , Q_{ref} occurs. Since it takes about 2 seconds to perform the P&O process this time interval limits how frequently the change in the reference active or reactive power can occur. Hence, the method is not suitable for PV power applications.

To emulate the effect of the capacitor tolerances on the input current ripple reduction,



Fig. 6.11: Experimental results illustrating operation of the proposed rule-based controller when the battery storage system supplies 10W active power and 15 VAr reactive power



Fig. 6.12: Operation of the rule-based controller with step-changing inverter output power

the capacitance value ($C = C_1 = C_2$) used for the second-order harmonic voltage reference parameter calculation was changed by $\pm 10\%$ from the nominal capacitor value 60μ F. Fig. 6.13 illustrates the effect of the change in the capacitance value on the second-order harmonic ripple current reduction when the waveform control method is used. For the considered output power operating point (10W active power and 15VAr reactive power), a -10% change in capacitance does not leads to a significant change in the ripple reduction. However, a +10% change in the capacitance value results in considerable reduction in the waveform controller performance.

Since the boost inverter second-order harmonic input current amplitude and phase angle vary with the inverter output power, the operation of the proposed controller was tested in all possible output power operating quadrants. The boost inverter input current ripple reductions obtained using the waveform control method and the proposed rule-based control method are compared for all possible output power conditions in Fig. 6.14 and



Fig. 6.13: Experimental results illustrating the effect of capacitor tolerances on the ripple current reduction when the waveform control method is used (a) Inductor currents and total input current of the boost inverter without any ripple current reduction method, (b) Inductor currents and total input current of the boost inverter with the waveform control (WFC) method and $C=60\mu$ F, (c) Inductor currents and total input current of the boost inverter with the waveform control method and $C=54\mu$ F, (d) Inductor currents and total input current of the boost inverter with the waveform control method and $C=66\mu$ F, (e) Inductor currents and total input current of the boost inverter with the waveform control method and $C=66\mu$ F, (e) Inductor currents and total input current of the boost inverter with the rule-based controller (RBC), (f) Comparison of the ripple amplitude and ripple reduction factor for cases (b) to (e).

summarized in Table 6.2. It can be observed that the proposed rule-based controller is able to reduce the ripple current at least six times in all conditions while the waveform control method fails to achieve significant ripple current reduction at some of the output power conditions. Percentage improvement in the ripple reduction R_{imp} when using the rule-based controller as compared with the waveform control method is also illustrated in Fig. 6.14.

$$R_{imp} = \frac{R_{RBC} - R_{WFC}}{R_{WFC}} \times 100\%$$
(6.55)

where, R_{RBC} and R_{WFC} are the ripple reduction factors obtained from the rule-based control method and the waveform control method respectively.

6.6 Conclusions

Due to the single phase operation of the boost inverter based battery-supercapacitor HESS, a second order harmonic ripple component exists at the DC side of the converter and the ripple component was allocated to the supercapacitor as a way of increasing the battery lifetime. However, such ripple current component increases the internal heating and losses in the supercapacitor and degrades its lifetime. Hence, it is vital to identify a suitable second-order harmonic ripple current reduction method for the boost inverter based battery-supercapacitor HESS. In order to identify a ripple reduction method, the operation of a single boost inverter based ESS was considered in this chapter.

First, the existing waveform control ripple current reduction method was extended to the grid connected boost inverter operation and the effects of the capacitor tolerances



Fig. 6.14: Comparison of the ripple current reduction using the waveform control method (WFC) and the rule-based controller (RBC)

Active power [W]	Reactive	Ripple	Ripple	Ripple
	power [VAr]	amplitude	amplitude	amplitude
		without any	reduction	reduction
		$\operatorname{compensation}$	factor with	factor with
		method [A]	RBC	WFC
			method	method
20	0	1.954	6.65	5.40
-20	0	1.067	6.28	5.10
0	15	0.605	6.80	3.30
0	-15	1.343	7.07	4.80
15	10	1.454	7.09	4.00
15	-10	1.843	7.26	5.20
-15	10	0.936	7.04	4.70
-15	-10	1.114	7.38	5.90
10	15	1.174	7.50	4.04

Table 6.2: Second-order harmonic ripple current mitigation for various output power conditions

and ESR of the inductors on its effectiveness were analyzed. It was illustrated that the capacitance tolerance and the inductor ESR values affect the performance of the waveform control method due to the direct calculation of the compensation signals. Then, a rule-based controller was proposed to mitigate the second-order harmonic ripple current in the boost inverter. The main advantage of the proposed controller was that it could reduce the second-order harmonic ripple current in all four output power quadrants without being affected by the capacitor tolerances and ESR values of the inductors. The proposed rule-based controller was validated experimentally and confirmed that the controller can achieve better ripple current reduction as compared to the waveform control approach without its performance being affected by the capacitor tolerances and the ESR values.

Chapter 7

An Input Current Feedback Method to Mitigate the DC-Side Low Frequency Ripple Current in a Single-Phase Boost Inverter

Due to the single phase operation of the boost inverter topology, a second-order harmonic ripple component can be observed at the DC side of the boost inverter based energy storage systems (ESSs). A novel current feedback ripple reduction method proposed in [68] is presented in this chapter. The proposed method modifies the boost inverter output capacitor reference voltages using an input current feedback signal. Compared to the existing waveform control ripple current reduction method, the performance of the proposed current feedback ripple reduction method does not get affected by the output capacitor tolerances and equivalent series resistance (ESR) of the boost inductors. Furthermore, compared to the rule-based control method presented in Chapter 6, the current feedback method demonstrates a superior ripple current reduction and dynamic performance in all output power quadrants and hence it is most suitable for the boost inverter based energy storage system. The performance of the proposed current feedback ripple reduction method is verified experimentally.

Later in the chapter, the proposed current feedback ripple reduction method is applied to the interleaved boost inverter based battery-supercapacitor HESS and its effectiveness in mitigating the second-order harmonic ripple component in the supercapacitor current is demonstrated [69]. The proposed HESS is the only single phase battery-supercapacitor HESS presented in literature which is able to mitigate the second-order harmonic ripple component in the supercapacitor current as well as able to reduce the switching frequency
ripple component in both the battery and the supercapacitor currents while allocating the fast power fluctuations to the supercapacitor.

7.1 Introduction

A boost inverter based battery-supercapacitor HESS was presented in Chapter 3 to Chapter 5. In all the HESS power allocation methods, the second order harmonic current component present at the DC side of the converter due to the single phase grid integration was allocated to the supercapacitor. However, such continuous ripple current component can increase the internal temperature of the supercapacitor and hence deteriorate its lifetime [61]. Hence, it is vital to identify a suitable second-order harmonic ripple current reduction method for the boost inverter based battery-supercapacitor HESS.

Chapter 6 extensively analyzed the operation of an existing waveform control method for the boost inverter and demonstrated that the capacitance tolerances and ESR of the inductors affect the performance of the waveform control method. A rule-based controller which generates the required output capacitor reference voltages for the waveform control ripple current reduction method was proposed in Chapter 6 [67]. The rule-based controller eliminated possible inaccuracies occurred in the traditional waveform control method due to the capacitor tolerances and inductor ESR values and was able to achieve better performance.

The rule-based control method employed a perturb and observe (P&O) approach to generate the required output capacitor reference voltages and the controller required about 2 seconds to perform the P&O process. The time required for the P&O operation limited how frequently the reference active or reactive power can change which is not desirable in the battery-supercapacitor HESS since the HESS should be able to operate under variable output power conditions.

A current feedback ripple reduction method proposed in [68] is presented in this chapter. First, the ripple current reduction in a single boost inverter is considered. Effectiveness of the proposed method is theoretically analyzed. In the proposed method, the output capacitor voltage reference signals are modified using the DC side current feedback signal as opposed to the complex calculations required in the waveform control method. Hence, performance of the proposed method is not affected by the capacitance tolerances of the output capacitors. Moreover, the proposed method is able to reduce the second-order harmonic ripple current at the DC side of the boost inverter without increasing other frequency harmonic components. Compared to the rule-based control ripple reduction method, the current feedback method is able to

117

achieve superior second-order harmonic ripple current reduction as well as better dynamic performance. The method is experimentally verified using a single-phase grid connected battery ESS prototype. Later in the chapter, the current feedback method is applied to the boost inverter based battery-supercapacitor HESS [69]. Effectiveness of the current feedback method on the supercapacitor current second-order harmonic ripple reduction is demonstrated experimentally.

7.2 Current Feedback Ripple Current Reduction Method

The configuration of a single phase grid connected boost inverter is shown in Fig. 7.1 and the operation of the boost inverter was discussed in Chapter 2. The voltages across the output capacitors C_1 and C_2 are controlled to follow voltage references given by,

$$v_{o1,ref} = V_{DC,ref} + \frac{V_o}{2}\sin(\omega t + \delta)$$
(7.1)

and

$$v_{o2,ref} = V_{DC,ref} - \frac{V_o}{2}\sin(\omega t + \delta).$$
(7.2)

respectively. V_o , ω , and δ are the inverter output voltage amplitude, frequency, and phase angle respectively. The boost inverter output current i_o is given by,

$$i_o = I_o \sin(\omega t + \alpha) \tag{7.3}$$



Fig. 7.1: Single phase grid connected boost inverter

where α is the phase angle of the inverter output current with respect to the grid voltage. The total input current to the boost inverter is given by

$$i_{in} = \frac{1}{v_{in}} \left[\frac{V_o I_o}{2} \cos(\alpha - \delta) + \frac{V_o}{2} \sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C \omega \sin(\alpha - \delta) \right] \cos(2\omega t - \theta)} \right]$$
(7.4)

where

$$\cos\theta = \frac{-I_o\cos(\alpha+\delta) + \frac{C\omega V_o\sin(2\delta)}{2}}{\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha-\delta)\right]}}$$

and $C = C_1 = C_2$. Please refer Appendix A.1 for the derivation. Then, the boost inverter input current can be re-written as,

$$i_{in} = i_{in,DC} + i_{in,AC} \tag{7.5}$$

where $i_{in,DC}$ and $i_{in,AC}$ are the DC and AC components of the total input current respectively.

The output capacitor voltage references $v_{o1,ref}$ and $v_{o2,ref}$ given by (7.1) and (7.2) can be modified as (7.6) and (7.7), without affecting the inverter differential output voltage.

$$v_{o1,ref} = V_{DC,ref} + \frac{V_o}{2}\sin(\omega t + \delta) - Ki_{in.AC}$$
(7.6)

$$v_{o2,ref} = V_{DC,ref} - \frac{V_o}{2}\sin(\omega t + \delta) - Ki_{in.AC}$$
(7.7)

where K is a feedback gain value. Then,

$$i_1 = I_o \sin(\omega t + \alpha) + C\omega \frac{V_o}{2} \cos(\omega t + \delta) - CK \frac{di_{in,AC}}{dt}$$
(7.8)

and

$$i_2 = -I_o \sin(\omega t + \alpha) - C\omega \frac{V_o}{2} \cos(\omega t + \delta) - CK \frac{di_{in,AC}}{dt}$$
(7.9)

Using (7.6), (7.7), (7.8), and (7.9), a differential equation for the input current AC component $i_{in,AC}$ can be obtained as

$$\frac{di_{in,AC}}{dt} = \frac{v_{in}}{(2V_{DC}CK - 2CK^2 i_{in,AC})}(i_{in,AC0} - i_{in,AC})$$
(7.10)

where

$$i_{in,AC0} = \frac{V_o}{2v_{in}} \sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C \omega \sin(\alpha - \delta)\right]} \cos(2\omega t - \theta)$$
(7.11)

corresponds to the AC component of the input current i_{in} given in (7.4) for the unmodified output capacitor reference voltages (7.1) and (7.2). Please refer Appendix A.1 for the derivation. Before finding an analytical solution, first, a numerical solution of the differential equation (7.10) is found using Matlab for a sample case of inverter supplying 15W active power and 10VAr reactive power to the grid and considering experimental prototype parameters summarized in Table 7.1. Fig. 7.2(a) illustrates the second-order harmonic component amplitude $(m_{2\omega})$ and the fourth-order harmonic component amplitude $(m_{4\omega})$ of the obtained numerical solution for various values of the feedback gain K. From Fig. 7.2(a) it is evident that the higher the value of the feedback gain K the more significant is the reduction in the second-order as well as the fourth-order harmonic components of the input current. A change in $m_{2\omega}$ for a unit change in K, $(\Delta m_{2\omega})$ is shown in Fig. 7.2(b). As K becomes larger, the change in $m_{2\omega}$ becomes smaller. Using this result, the required minimum value for K can be found such that,

$$|\Delta m_{2\omega}| \le |\Delta m_{2\omega,\max}| \tag{7.12}$$

where $\Delta m_{2\omega,\text{max}}$ is a pre-determined value of the maximum allowed change in $m_{2\omega}$ for a unit change in K for all possible output power conditions. Then, for the selected maximum change in the second-order harmonic ripple amplitude for a unit change in K, $\Delta m_{2\omega,\text{max}}$, the bounds on K are given by,



$$0 < K \le [K|(\Delta m_{2\omega,\max} \ge |\Delta m_{2\omega}|)].$$
(7.13)

Fig. 7.2: Numerical solution of the variation of the input current with K, (a) the second and the fourth-order harmonic current ripple amplitude, (b) change in the second-order harmonic ripple amplitude for a unit change in K.

Now, to obtain an analytical solution of the second-order harmonic input current amplitude $m_{2\omega}$, it is assumed that the steady state input current AC component with the modified output capacitor voltage references is

$$\dot{h}_{in,AC} = m_{2\omega}\cos(2\omega t + \theta_{2\omega}) + m_{4\omega}\cos(4\omega t + \theta_{4\omega})$$
(7.14)

The input current AC component differential equation (7.10) can be re-written as

$$2V_{DC}CK\frac{di_{in,AC}}{dt} - 2CK^2i_{in,AC}\frac{di_{in,AC}}{dt} + v_{in}i_{in,AC} = v_{in}i_{in,AC0}$$
(7.15)

After substituting (7.14) into (7.15) and simplifying, (7.16) is obtained.

$$-4V_{DC}CK\omega m_{2\omega}\sin(2\omega t + \theta_{2\omega}) + 2CK^2\omega m_{2\omega}m_{4\omega}\sin(2\omega t + \theta_{4\omega} - \theta_{2\omega}) + v_{in}m_{2\omega}\cos(2\omega t + \theta_{2\omega}) = v_{in}i_{in,AC0}.$$
(7.16)

Considering that the product $m_{2\omega}m_{4\omega} \approx 0$, since $m_{2\omega}$ and $m_{4\omega}$ are very small when K is high, the magnitude of the second-order harmonic component in (7.16) can be found as

$$m_{2\omega} = \frac{v_{in}|i_{in,AC0}|}{\sqrt{(-4V_{DC}CK\omega)^2 + v_{in}^2}}$$
(7.17)

Fig. 7.3(a) compares the obtained analytical solution (7.17) with the numerical solution found using Matlab when the inverter delivers 15W, 10VAr to the grid. The error between the numerical and the analytical solution ($m_{2\omega,numerical} - m_{2\omega,analytical}$) is shown in Fig. 7.3(b) and proves the validity of (7.17) for higher values of K.

For further evaluation of the proposed method, a ripple reduction factor R is defined as the ratio of $m_{2\omega}$ before and after application the current feedback method. The ripple reduction factor R can be derived from (7.17) as

$$R = \frac{\sqrt{(4V_{DC}CK\omega)^2 + v_{in}^2}}{v_{in}}$$
(7.18)

To ensure proper operation of the boost inverter the DC shift of the output capacitor voltage V_{DC} has to meet condition

$$V_{DC} \ge v_{in} + \frac{V_o}{2} + \max(Ki_{in,AC})$$
 (7.19)

According to the numerical solution shown in Fig. 7.2(a), the second-order harmonic ripple component is the dominant component in the input current compared to the fourth-order harmonic component. Hence, (7.19) can be re-written as,

$$V_{DC} \ge v_{in} + \frac{V_o}{2} + \max(Km_{2\omega})$$
 (7.20)

Then, using (7.17) and assuming large K, (7.20) can be approximated as,

$$V_{DC} \ge v_{in} + \frac{V_o}{2} + \frac{v_{in} \max(|i_{in,AC0}|)}{4V_{DC}C\omega}$$
(7.21)

where

$$\max(|i_{in,AC0}|) = \frac{V_o}{2v_{in}} \sqrt{\left[I_{o,max}^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_{o,max} C\omega\right]}.$$
 (7.22)

Then, the minimum required DC voltage shift V_{DC} for the output capacitor reference voltages can be calculated from,

$$V_{DC} \ge V_{DC,\min} \tag{7.23}$$

where

$$V_{DC,\min} = \frac{2C\omega(V_o + 2v_{in}) + \sqrt{\left[2C\omega(V_o + 2v_{in})\right]^2 + 16C\omega v_{in}\max(|i_{in,AC0}|)}}{8C\omega}.$$
 (7.24)

Then, the maximum output capacitor voltage $V_{o1,\max}$ when $V_{DC} = V_{DC,\min}$ is,

$$V_{o1,\max} = V_{DC,\min} + \frac{V_o}{2} + \frac{v_{in}\max(|i_{in,AC0}|)}{4V_{DC,\min}C\omega}$$
(7.25)

The relationship between the $V_{DC,\min}$ and $V_{o1,\max}$ versus the output capacitor value Cfor the experimental prototype parameters given in Table 7.1 is illustrated in Fig. 7.4. The required output capacitor value $C_{req,i}$ can be found as explained in Chapter 6 [67]. First, the maximum allowable output capacitor voltage is obtained using the input voltage v_{in} and the maximum gain of the converter legs G_{\max} , as $V_{o1,\max} = v_{in}G_{\max}$. Then, the required output capacitor value $C_{req,i}$ and the minimum DC voltage shift corresponding to $V_{o1,\max}$ can be found from Fig. 7.4. A detailed comparison of the output capacitor



Fig. 7.3: Comparison of the numerical and approximated analytical solution, (a) the second-order harmonic current ripple amplitude, (b) error between the numerical and analytical solution.



Fig. 7.4: Illustration of the effect of (a) the output capacitor tolerance and (b) the input voltage variation on the ripple reduction method performance when P=15W and Q=10VAr

requirement for the boost inverter operation without and with an output capacitor voltage modified ripple reduction method can be found in Chapter 6 [67].

7.3 Boost Inverter Control System with the Current Feedback Method

The overall block diagram of the grid-connected boost inverter with the proposed ripple current reduction method is shown in Fig. 7.5. A set of band-pass filters are employed to obtain the AC component of the input current with minimum phase distortion and to block any DC component of the input current in the feedback path.

The maximum gain of the boost converter leg is taken as 6 and hence $V_{o1,\text{max}}$ equals to 76.8V. Then, the required output capacitor value for the operation of the boost inverter with the current feedback method is obtained from Fig. 7.4 as 54.6 μ F. The nearest commercially available value of 60 μ F was selected for the experimental setup.

The minimum required feedback gains K to satisfy (7.12) for $\Delta m_{2\omega,\text{max}} = 0.001$ A are calculated for each output power condition. The calculated feedback gains are illustrated in Fig. 7.6. Then, the value of K=100 is selected to satisfy (7.12) for all output power conditions. To verify the validity of the approximated analytical solution for the selected K, the error between the numerical solution and the analytical solution for all output power conditions and K=100 is illustrated in Fig. 7.7. It is evident that the error is negligible.

A double loop (DL) control method is used to control the boost converter legs as presented in Chapter 3. The power allocation is achieved using a high pass filter and a SCVC based power allocation method as explained in Chapter 3. The active and reactive power flow between the boost inverter and the grid is achieved by controlling the inverter output voltage amplitude and phase angle with respect to the grid voltage.



Fig. 7.5: Overall block diagram of the control system



Fig. 7.6: Illustration of the required feedback gains K to satisfy (7.12) for each output power condition.



Fig. 7.7: Error between the numerical solution and the analytical solution for all output power conditions and K=100.

7.4 Experimental Results

The proposed current ripple reduction method for a grid connected boost inverter was verified using an experimental prototype and the parameters of the experimental prototype are summarized in Table 7.1. A 6.4Ah, 12.8V LiFePO₄ battery from K2 Energy and a 16V, 58F Maxwell supercapacitor module are used in the prototype.

Fig. 7.8 illustrates operation of the single-phase grid connected boost inverter with and without the proposed second-order harmonic ripple reduction method when delivering 15W active and 10VAr reactive power to the grid. The output capacitor voltage waveforms and the inverter output voltage are shown in Figs. 7.8(a) and 7.8(b) without and with the proposed method respectively. The inverter output voltage remains unchanged despite the modification of the output capacitor voltages. The operation of the active and reactive power controller is illustrated in Figs. 7.8(c) and 7.8(d). Figs. 7.8(e) and 7.8(f) depict the

Parameter	Value	Parameter	Value
Power converter parameters			
V_o	40 V	f	$50~\mathrm{Hz}$
C_1, C_2	$60 \ \mu F$	L_1, L_2	$210~\mu\mathrm{H}$
$I_{o,max}$	1 A	T_F	0.001
$V_{DC,ref}$	$42 \mathrm{V}$	f_{sw}	$20 \mathrm{~kHz}$
R_{L1}, R_{L2}	$0.24 \ \Omega$		
Grid parameters			
V_g	40 V	L_g	$20 \mathrm{~mH}$
Voltage control loop parameters			
$K_{p,PR}$	0.1	$K_{i1,PR}$	4
$K_{i2,PR}$	6	ω_{c1},ω_{c2}	0.5
Current control loop parameters			
$K_{p,PI}$	2.615	$K_{i,PI}$	8.44×10^{-5}

Table 7.1: Experimental prototype parameters

inductor current waveforms of the boost converter legs and the total input current to the boost inverter without and with the proposed ripple current reduction method respectively. The inductor current waveforms modified due to the changes in the output capacitor voltage waveforms and a significant reduction of the input current ripple component can be observed.

The frequency spectrum characteristics of the boost inverter DC side current with and without the proposed ripple current reduction method are compared in Figs. 7.8(g) and 7.8(h). The second-order harmonic ripple amplitude is reduced from 1.482A to 0.0588A, achieving approximately 25 times second-order harmonic ripple current reduction. In addition, the proposed method reduced the first-order harmonic current component amplitude from 0.085A to 0.036A and the fourth-order harmonic current component amplitude from 0.037A to 0.004A. The frequency spectrum characteristics illustrate the ability of the proposed system to mitigate the second-order harmonic ripple component of the boost inverter input current without increasing other harmonic current components.

Dynamic performance of the proposed ripple current reduction method is shown in Fig. 7.9. At t=2s, 15W active power and 10VAr reactive power reference step change was applied. Without the proposed method, the second-order harmonic ripple amplitude reaches 1.482A due to output power step change. However, with the proposed system, the maximum second-order harmonic ripple amplitude during the transient time is 0.172A



Fig. 7.8: Experimental results illustrating operation of the battery storage system when delivering 15W active power and 10VAr reactive power. (a) Output capacitor voltage waveforms and inverter output voltage without the proposed controller, (b) Output capacitor voltage waveforms and inverter output voltage with the proposed controller, (c) Inverter output voltage, grid voltage and inverter output current without the proposed controller, (d) Inverter output voltage, grid voltage and inverter output current with the proposed controller, (e) Inductor currents and total input current of the boost inverter without the proposed controller, (f) Inductor currents and total input current of the boost inverter with the proposed controller, (g) Frequency characteristics of the input current with the proposed controller and (h) Frequency characteristics of the input current with the proposed controller.

and reduces to 0.0588A in the steady state. This validates the ability of the proposed method to reduce the second-order harmonic current component even during the active and reactive power transients.

The boost inverter input current ripple reduction achieved using the proposed current



Fig. 7.9: Experimental results illustrating transient operation of the proposed control system in response to 15W and 10VAr power step change, (a) Inverter output active power, (b) inverter output reactive power (c) second-order harmonic ripple amplitude with and without the proposed method.

feedback method is compared with the waveform control method [74] and the rule-based control method [67] for all the output power conditions in Fig. 7.10. In order to have a fair comparison, the same experimental setup and control parameters as in [67] were used. It can be observed that the proposed current feedback method is able to reduce the second-order harmonic ripple current more than twenty times for all output power conditions.

The change in the second-order harmonic ripple current amplitude $|\Delta m_{2\omega}|$ for a unit change in K is illustrated for each output power condition in Fig. 7.11. The result validates that the selection of the feedback gain K=100 results in the change in the second-order harmonic ripple current amplitude $|\Delta m_{2\omega}| < 0.001$ for all output power conditions.



Fig. 7.10: Comparison of the ripple current reduction using the waveform control method, the rule-based control method and the proposed current feedback method (K=100).



Fig. 7.11: Experimental results illustrating change in the second-order harmonic ripple current amplitude $\Delta m_{2\omega}$ for a unit change in K for all output power conditions.

7.5 Battery-Supercapacitor HESS with Reduced Low Frequency Input Current Ripple

The current feedback ripple reduction method is able to reduce the second-order harmonic ripple component at the DC side of the boost inverter. Compared to the existing ripple current reduction methods for the boost inverter, the current feedback method achieved significant ripple current reduction without increasing other harmonic components. Furthermore, the current feedback method is able to achieve a significant ripple current reduction even in the output power transients.

In this section, the current feedback ripple reduction method is applied to the boost inverter based battery-supercapacitor HESS presented in Chapter 3 to reduce the second-order harmonic ripple current component in the supercapacitor current [69].

7.5.1 Second-order Harmonic Ripple Component in the Supercapacitor Current

The configuration of the boost inverter based grid connected battery-supercapacitor HESS is illustrated in Fig. 7.12.

The instantaneous output power from the left hand side boost converter legs can be written as,

$$P_{o1} + P_{C1} = (i_{o1} + i_{C1})v_{o1}. (7.26)$$

Similarly,

$$P_{o2} + P_{C2} = (i_{o2} + i_{C2})v_{o2}. (7.27)$$

Then, assuming a lossless power conversion,

$$P_{in,HESS} = P_{o1} + P_{o2} + P_{C1} + P_{C2}.$$
(7.28)

where $P_{in,HESS}$ is the total input power to the HESS. Then,

$$P_{in,HESS} = \left[\frac{V_o I_o}{2}\cos(\alpha - \delta) + \frac{V_o}{2}\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha - \delta)\right]}\cos(2\omega t - \theta)\right]}$$
(7.29)

where

$$\cos\theta = \frac{-I_o\cos(\alpha+\delta) + \frac{C\omega V_o\sin(2\delta)}{2}}{\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha-\delta)\right]}}$$

From (7.29), the HESS total input power contains a DC component which corresponds to the HESS output power and a second-order harmonic power component. According to the power allocation strategy, the second-order harmonic power component is allocated to the supercapacitor.

7.5.2 Current Feedback Method for the HESS

The supercapacitor current of the battery-supercapacitor HESS can be written as,

$$i_{sc} = i_{sc,DC} + i_{sc,AC} \tag{7.30}$$

where $i_{sc,DC}$ and $i_{sc,AC}$ are the DC and AC components of the supercapacitor current. The output capacitor reference voltages of the HESS can be modified as,

$$v_{o1,ref} = V_{DC,ref} + \frac{V_{o,ref}}{2} - Ki_{sc,AC}$$
(7.31)

$$v_{o2,ref} = V_{DC,ref} - \frac{V_{o,ref}}{2} - Ki_{sc,AC}$$
(7.32)





where K is the feedback gain. Then, the total instantaneous input power supplied by the HESS can be obtained as,

$$P_{in,HESS} = v_{batt}i_{batt} + v_{sc}i_{sc,DC} + v_{sc}i_{sc,AC} = \frac{V_o I_o}{2}\cos(\alpha - \delta) + \frac{V_o}{2}\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha - \delta)\right]}\cos(2\omega t - \theta) - 2CKV_{DC}\frac{di_{sc,AC}}{dt} + 2CK^2 i_{sc,AC}\frac{di_{sc,AC}}{dt}$$
(7.33)

Considering the AC component of (7.33), a differential equation for the supercapacitor current AC component can be obtained as,

$$\frac{di_{sc,AC}}{dt} = \frac{v_{sc}}{(2V_{DC}CK - 2CK^2 i_{sc,AC})}(i_{sc,AC0} - i_{sc,AC})$$
(7.34)

where

$$i_{sc,AC0} = \frac{V_o}{2v_{sc}} \sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C \omega \sin(\alpha - \delta)\right]} \cos(2\omega t - \theta)$$
(7.35)

is the AC component of the supercapacitor current with the unmodified output capacitor reference voltages.

First, a numerical solution for (7.34) is obtained using Matlab differential equation solver when the HESS is supplying 15W active power and 10VAr reactive power to the grid as a sample case. For all the calculations, prototype parameters presented in Chapter 3 are used. The second-order harmonic amplitude $(m_{2\omega})$ and the fourth-order harmonic amplitude $(m_{4\omega})$ of the obtained numerical solution for various values of K at steady state $(v_{sc} = v_{sc,ref})$ are illustrated in Fig. 7.13(a). It can be observed that a higher feedback gain K leads to a significant reduction in both the second-order and the fourth-order harmonic components in the supercapacitor current. Fig. 7.13(b) illustrates the change in $m_{2\omega}$ for a unit change in K and it can be observed that as K increases, the change in $m_{2\omega}$ becomes smaller. Then the required minimum value for the feedback gain K can be calculated such that,

$$|\Delta m_{2\omega}| \le \Delta m_{2\omega,max} \tag{7.36}$$

where $\Delta m_{2\omega,max}$ is the maximum allowed change in $m_{2\omega}$ for a unit change in K for all HESS output power conditions.

As presented in Section 7.2, an analytical solution for the supercapacitor current second-order harmonic component can be obtained as,

$$m_{2\omega} = \frac{v_{sc}|i_{sc,AC0}|}{\sqrt{(-4V_{DC}CK\omega)^2 + v_{sc}^2}}$$
(7.37)



Fig. 7.13: Variation of the supercapacitor current numerical solution with K, (a) the second and the fourth-order harmonic current ripple amplitudes, (b) change in the second-order harmonic ripple amplitude for a unit change in K.

7.5.3 Experimental Results

A 6.4Ah, 12.8V, 81.92Wh LiFePO₄ battery from K2 Energy and a 16V, 58F, 7424J Maxwell supercapacitor module are used in the experimental prototype. The HESS control system is implemented on a DSpace DS1006 system using Matlab Simulink. Fig. 7.14 illustrates the overall block diagram of the single-phase grid connected HESS with the ripple current reduction method. The same double loop control system with the filter based power allocation method presented in Chapter 3 is used in the experiments.

To obtain the AC component of the supercapacitor current with minimum phase distortion and to block any DC component flowing in to the feedback path, a set of band-pass filters are used. For each output power condition, the minimum required feedback gain is calculated to satisfy (7.36) when $\Delta m_{2\omega,max} = 0.001$ A, and illustrated in Fig. 7.15. Then K is selected as 100 to satisfy (7.36) for all output power conditions.

A DL controller is used to control the boost inverter as explained in Chapter 3 and the same controller and converter parameters as in Chapter 3 are considered.

Fig. 7.16 depicts the steady state operation of the HESS without and with the current feedback ripple reduction method when delivering 15W and 10VAr power to the grid. Figs. 7.16(a) and 7.16(b) show the output capacitor voltage waveforms and the HESS output voltage waveforms without and with the ripple reduction method respectively.







Fig. 7.15: Illustration of the required feedback gains K to satisfy (7.36) for each output power condition.

The battery current and the supercapacitor current without the ripple reduction method are illustrated in Fig. 7.16(c), and it can be observed that the battery supplies the required average current while the supercapacitor supplies the second-order harmonic current component.

The battery and the supercapacitor current waveforms with the ripple reduction method are shown in Fig. 7.16(d) and a significant reduction in the supercapacitor current ripple component can be observed.

The frequency spectrums of the supercapacitor current without and with the ripple reduction method are shown in Figs. 7.16(e) and 7.16(f), respectively. The second-order harmonic ripple amplitude is reduced from 1.656A to 0.061A which corresponds to approximately 27 times ripple reduction. Moreover, the fourth-order harmonic ripple component of the supercapacitor is reduced from 0.102A to 0.043A.

Operation of the HESS with the ripple current reduction method for a step changing output power profile is illustrated in Fig. 7.17. It can be observed that the supercapacitor responds to sudden power changes and the battery current gradually increases to supply the required average power output. The battery current settles down to supply the required average current ones the supercapacitor is recharged to the reference voltage. Fig. 7.17 compares the second-order harmonic ripple amplitude without the ripple reduction method $(m_{2\omega,n})$ and with the ripple reduction method $(m_{2\omega,CFB})$. Significant supercapacitor current second-order harmonic ripple reduction can be observed in all output power conditions.



Fig. 7.16: Experimental waveforms of the HESS when delivering 15W active power and 10VAr reactive power. (a) Output capacitor voltage and inverter output voltage without the ripple reduction method, (b) output capacitor voltage and inverter output voltage with the ripple reduction method, (c) battery and supercapacitor currents without the ripple reduction method, (d) battery and supercapacitor currents with the ripple reduction method, (e) frequency characteristics of the supercapacitor current with the ripple reduction method and (f) frequency characteristics of the supercapacitor current with the ripple reduction method.

7.6 Conclusions

Due to the direct DC/AC power conversion, a second-order harmonic ripple current component appears in the DC side of the boost inverter based battery-supercapacitor HESS and the second-order harmonic current component was allocated to the supercapacitor as a way of extending the battery lifetime. However, the continuous ripple current component adversely affects the lifetime of the supercapacitor due to internal heating. Hence, it is vital to identify a suitable second-order harmonic ripple current reduction method, which can be applicable to the boost inverter based HESS.

The performance of the existing waveform control ripple reduction method was affected by the capacitor tolerances and the ESR of the boost inductors and a rule-based controller was proposed in Chapter 6 to overcome the issues associated with the existing waveform control ripple reduction method. Even though, the rule-based controller achieved better performances compared to the waveform control method, the rule-based controller required



Fig. 7.17: Operation of the HESS with the current feedback ripple reduction method for a step changing output power profile.

about 2 seconds to perform the P&O process. This limits how frequently the change in the reference active or reactive power can occur and hence it limits the applicability of the rule-based controller for the boost inverter based battery-supercapacitor HESS.

A novel current feedback ripple reduction method was proposed in this chapter, The compensation signals were generated using the AC component of the boost inverter input current and the method achieved superior ripple reduction performance over the other methods. Furthermore, the current feedback method was able to mitigate the second-order harmonic ripple component of the boost inverter input current even in the output power transients. The performance of the proposed current feedback method was validated using an experimental boost inverter prototype.

Later, the proposed current feedback ripple reduction method was applied to the boost inverter based battery-supercapacitor HESS. The boost inverter based HESS with the current feedback method was able to significantly reduce the second-order harmonic component of the supercapacitor current while operating as a conventional HESS where the fast power fluctuations were allocated to the supercapacitor. Operation of the proposed battery-supercapacitor HESS with reduced low frequency input current ripple was verified using experimental results.

The proposed boost inverter based HESS is the only single-phase direct grid connected battery-supercapacitor HESS where the second-order harmonic ripple current components as well as the switching frequency ripple current components of both the battery and the supercapacitor are mitigated. The ripple component reduction of the DC source currents reduces the internal heating and increases the lifetime of the DC sources.

Chapter 8

Conclusions and Future Work

8.1 Conclusions

The key objective of the research was to study and develop a single phase direct grid integrated battery-supercapacitor HESS. The boost inverter topology was identified as a suitable power converter configuration for the direct grid connected battery-supercapacitor HESS due to its ability to provide both the boosting and inversion functions in a single power processing stage.

Chapter 2 discussed the operation of the boost inverter topology which was used to design the single phase grid connected battery-supercapacitor HESS in this research. The switching frequency ripple component of the boost inverter DC side current was analyzed with the conventional in-phase PWM operation. Then, a phase shifted interleaved operation for the boost inverter topology was proposed and its effectiveness in reducing the switching frequency ripple component of the boost inverter DC side current was illustrated. Furthermore, a boost inverter component selection method for the interleaved boost inverter was also presented.

Chapter 3 presented the proposed interleaved boost inverter based single phase grid connected battery-supercapacitor HESS. A SCVC and a high-pass filter based power allocation method was employed to allocate the high frequency power variations and the second-order harmonic input current component to the supercapacitor. A DL control strategy was employed to control the power converter structure. A battery SOC controller with an extended Kalman filter based battery SOC estimator was developed. Experimental results were used to validate the operation of the proposed interleaved boost inverter based battery-supercapacitor HESS. The proposed boost inverter based HESS was able to allocate the fast power fluctuations and the second-order harmonic ripple current component to the supercapacitor while the battery supplied the steady power requirement. The supercapacitor voltage was maintained around a reference voltage level to reduce the risk of supercapacitor over-charging and over-discharging. The battery SOC obtained using an extended Kalman filter was maintained within a safe SOC operating region. The interleaved operation was able to significantly reduce the switching frequency ripple component of both the battery and the supercapacitor currents. The proposed HESS was the first experimentally verified single phase direct-grid integrated battery-supercapacitor HESS. Furthermore, the proposed HESS was the only HESS designed to reduce the switching frequency ripple component of both the battery of both the battery and the supercapacitor designed to reduce the switching frequency ripple component of both the battery and the battery and the supercapacitor currents.

Chapter 4 analyzed the power allocation between the battery and the supercapacitor for a supercapacitor SOC controlled filter based battery-supercapacitor HESS. A SCEC and a filter based power allocation method was proposed to circumvent the challenges associated with the non-linearity of the SCVC and a filter based HESS power allocation method discussed in Chapter 3. Furthermore, a power allocation parameter selection method, an energy storage element sizing method and the dynamic behavior of the HESS power allocation method for a supercapacitor energy controlled filter based HESS were analyzed for a given HESS application. Experimental results were presented to illustrate the operation of the HESS and the validity of the selected power allocation parameters. The selected supercapacitor size enabled the operation of the HESS within the maximum and the minimum supercapacitor voltage limits.

Chapter 5 presented a novel fixed frequency SM controller for the boost inverter based battery-supercapacitor HESS. The PWM based fixed frequency SM controller was employed to overcome high and variable frequency operation associated with the traditional SM control technique. Theoretical aspects of the SM controller design were discussed and a SCEC based power allocation method was proposed to allocate the fast power fluctuations to the supercapacitor without incorporating an additional filter. The power allocation parameter selection and the supercapacitor sizing were discussed for a given HESS application. The proposed control system for the boost inverter based HESS was experimentally verified using a laboratory prototype. The proposed controller was able to satisfy the HESS output power requirement while allocating the ripple current and fast power fluctuations to the supercapacitor. Moreover, the selected HESS power allocation parameters and the value of the supercapacitor enabled the operation of the HESS within the pre-determined supercapacitor voltage limits. Compared to the traditional DL controlled boost inverter based battery-supercapacitor HESS, the proposed SM controller was able to achieve better output capacitor reference tracking and hence reduce the risk of the DC current injection into the grid when the ESR values of the boost inductors become unequal due to the tolerances and temperature variations.

In the boost inverter based battery-supercapacitor HESS presented in Chapter 3 to Chapter 5, the second-order harmonic ripple current component at the DC side of the converter was allocated to the supercapacitor as a way of extending the lifetime of the battery. However, such continuous ripple current will adversely affect the lifetime of the supercapacitor due to internal heating.

Chapter 6 and Chapter 7 were focused on identifying a suitable second-order harmonic ripple reduction method for the boost inverter based battery-supercapacitor HESS.

Chapter 6 extensively studied the operation of the existing waveform control ripple reduction method for the boost inverter and illustrated that the performance of the waveform control method can be affected by the output capacitor tolerances and ESR values of the boost inductors. Hence, a novel rule-based controller was proposed to overcome the limitations of the waveform control method. The performance of the rule-based controller was experimentally verified and unlike the waveform control method, the performance of the rule-based controller was not affected by the output capacitor tolerances and ESR values of the boost inductors. However, the rule-based controller required about 2 seconds to reduce the ripple current component once a change in the active or reactive power occurred. Hence, it is not applicable if the boost inverter based battery-supercapacitor HESS is required to operate under variable output power conditions.

A novel current feedback ripple reduction method was proposed for the boost inverter topology in Chapter 7. The compensation signals for the output capacitor reference voltages were generated using an input current feedback signal. The experimental results illustrated a superior second-order harmonic ripple current reduction compared to the waveform control method as well as to the rule-based control method. Furthermore, the proposed current feedback ripple reduction method was able to achieve a significant ripple current reduction even during the output power transients and hence it is applicable to the boost inverter based battery-supercapacitor HESS operating under variable output power conditions. Then, the proposed current feedback ripple reduction method was applied to the boost inverter based HESS and the experimental results illustrated a significant second-order harmonic ripple component reduction in the supercapacitor current.

The proposed battery-supercapacitor HESS was the first experimentally verified single phase direct grid integrated HESS. Furthermore, the proposed HESS was the only single phase battery-supercapacitor HESS which is able to reduce both the second-order harmonic ripple component and the switching frequency ripple component of both the battery and the supercapacitor currents. The proposed SCEC and filter based power allocation method enabled precise sizing of the energy storage elements and also selection of the power allocation parameters. The HESS was able to allocate the fast power fluctuations to the supercapacitor while operating within the battery and the supercapacitor SOC limits. The proposed fixed frequency SM controller was able to achieve better output capacitor reference tracking and hence reduced the risk of the DC current injection into the grid. Furthermore, the proposed SM controlled HESS was the first experimentally verified SM controlled battery-supercapacitor HESS presented in the literature.

8.2 Future Work

Possible extensions to the research can be summarized as follows.

- 1. *Identify the applicability of advanced power allocation methods to the HESS*: The applicability of the advanced power allocation methods such as wavelet transform methods, fuzzy logic and neural network can be studied. A suitable power allocation parameter selection method and energy storage element sizing method have to be developed for the new power allocation methods.
- 2. Development of a three-phase direct grid connected battery-supercapacitor HESS: Three-phase boost inverter topology can be used to implement a direct three-phase grid integrated battery-supercapacitor HESS. The power allocation method design, power allocation parameter selection and energy storage element sizing can be studied for three-phase ESS applications.
- 3. Development of a control system to handle grid faults and abnormalities: The energy storage system has to be disconnected from the grid during grid faults and abnormalities while supporting critical loads. A control system can be studied to ensure a smooth transition between two operating modes.

A.1 Appendix 1

A.1.1 Derivation of equation (4.8)

Using Fig. A.1.1, following equation can be written.

$$P_{sc} = P_{tot} \times \left(\frac{as}{1+as}\right) - 2P_{charge} \tag{A.1.1}$$

Additionally,

$$2P_{charge} = E_{sc,ref} - \left(E_{sc,init} - \frac{P_{sc}}{s}\right) \times 2K_p, Esc$$
(A.1.2)

Using (A.1.1) and (A.1.2),

$$P_{sc} = P_{tot}\left(\frac{as}{1+as}\right) - 2K_{p,Esc}\left[E_{sc,ref} - \left(E_{sc},init - \frac{P_{s}c}{s}\right)\right]$$
(A.1.3)

The equation can be rearranged as,

$$P_{sc} = \frac{as^2}{as^2 + (1 + 2aK_{p,Esc})s + 2K_{p,Esc}}P_{tot} - \frac{2K_{p,Esc}s}{(s + 2K_{p,Esc})}(E_{sc,ref} - E_{sc,init}) \quad (A.1.4)$$

Since $P_{tot} = \frac{1}{\eta} P_{HESS}$,

$$P_{sc} = \frac{as^2}{as^2 + (1 + 2aK_{p,Esc})s + 2K_{p,Esc}} \frac{1}{\eta} P_{HESS} - \frac{2K_{p,Esc}s}{(s + 2K_{p,Esc})} (E_{sc,ref} - E_{sc,init})$$
(A.1.5)

A.1.2 Derivation of equation (5.15)

The sliding surface for the left hand side supercapacitor connected boost converter leg, S_{sc1} is given by,

$$S_{sc1} = K_{sc1,a}(v_{o1} - v_{o1,ref}) + K_{sc1,b} \int (v_{o1} - v_{o1,ref})dt + K_{sc1,c}i_{Lsc1}$$
(A.1.6)

The equivalent control input for the SM controller is obtained by solving,

$$\frac{dS_{sc1}}{dt} = 0 \tag{A.1.7}$$



Fig. A.1.1: Simplified block diagram of the HESS power allocation with the SCEC.

Using (A.1.6),

$$K_{sc1,a}\frac{dv_{o1}}{dt} + K_{sc1,b}\left(v_{o1} - v_{o1,ref}\right) + K_{sc1,c}\frac{di_{Lsc1}}{dt} = 0$$
(A.1.8)

Substituting (5.5) and (5.6) in to (A.1.8),

$$K_{sc1,a} \left(\frac{i_{obatt1} - i_{o1} + (1 - u_{sc1})i_{Lsc1}}{C_1} \right) + K_{sc1,b} \left(v_{o1} - v_{o1,ref} \right) + K_{sc1,c} \left(\frac{v_{sc} - i_{Lsc1}R_{Lsc1} - (1 - u_{sc1})v_{o1}}{L_{sc1}} \right) = 0 \quad (A.1.9)$$

The $(1 - u_{sc1})$ value which satisfies (A.1.9) is the equivalent control input $\bar{u}_{sc1,eq}$. Using (A.1.9), the equivalent control input can be obtained as,

$$\bar{u}_{sc1,eq} = \frac{k_{sc1,a}(i_{obatt1} - i_{o1}) + k_{sc1,b}(v_{o1} - v_{o1,ref}) + k_{sc1,c}(v_{sc} - i_{Lsc1}R_{Lsc1})}{k_{sc1,c}v_{o1} - k_{sc1,a}i_{Lsc1}}$$
(A.1.10)

where,

$$\begin{cases} k_{sc1,a} = K_{sc1,a} L_{sc1} \\ k_{sc1,b} = K_{sc1,b} L_{sc1} C_1 \\ k_{sc1,c} = K_{sc1,c} C_1 \end{cases}$$

A.1.3 Derivation of equation (5.26)

The sliding surface for the left hand side battery connected boost converter leg, S_{batt1} is given by,

$$S_{batt1} = (i_{Lbatt1} - i_{Lbatt1,ref}) + K_{batt1} \int (i_{Lbatt1} - i_{Lbatt1,ref}) dt$$
(A.1.11)

The equivalent control input for the SM controller is obtained by solving,

$$\frac{dS_{batt1}}{dt} = 0 \tag{A.1.12}$$

Using (A.1.11),

$$\frac{di_{Lbatt1}}{dt} + K_{batt1} \left(i_{Lbatt1} - i_{Lbatt1,ref} \right) = 0 \tag{A.1.13}$$

Substituting (5.20) in to (A.1.12),

$$\frac{v_{batt} - i_{Lbatt1}R_{Lbatt1} - (1 - u_{batt1})v_{o1}}{L_{b}att1} + K_{batt1}(i_{Lbatt1} - i_{Lbatt1,ref}) = 0$$
(A.1.14)

The $(1-u_{batt1})$ value which satisfies (A.1.14) is the equivalent control input $\bar{u}_{batt1,eq}$. Using (A.1.14), the equivalent control input can be obtained as,

$$\bar{u}_{batt,eq} = \frac{v_{batt} - i_{Lbatt1}R_{Lbatt1} + k_{batt1}(i_{Lbatt1} - i_{Lbatt1,ref})}{v_{o1}}$$
(A.1.15)

where

$$k_{batt1} = K_{batt1} L_{batt1}$$

A.1.4 Derivation of equation (5.39)

Using Fig. A.1.2, following equation can be written.

$$\left[-\left(-\frac{(P_{tot} - P_{batt})}{s} + E_{sc,init}\right) + E_{sc,ref}\right] \left(\frac{K_{p,Esc}s + K_{i,Esc}}{s}\right) 2v_{batt} = P_{batt} \quad (A.1.16)$$

Then, (A.1.16) can be re-arranged as

$$P_{batt} = \left(\frac{(2v_{batt}K_{p,Esc})s + (2v_{batt}K_{i,Esc})}{s^2 + (2v_{batt}K_{p,Esc})s + (2v_{batt}K_{i,Esc})}\right) P_{tot} + \left(\frac{(2v_{batt}K_{p,Esc})s^2 + (2v_{batt}K_{i,Esc})s}{s^2 + (2v_{batt}K_{p,Esc})s + (2v_{batt}K_{i,Esc})}\right) (E_{sc,ref} - E_{sc,init})$$
(A.1.17)

A.1.5 Derivation of equation (6.40)

Using the boost inverter model, the inductor currents i_{L1} can be written as

$$i_{L1} = \frac{i_1 v_{o1}}{v_{in} - i_{L1} R_{L1}} \tag{A.1.18}$$

Using (A.1.18) a quadratic equation for i_{L1} can be obtained as

$$R_{L1}i_{L1}^2 - v_{in}i_{L1} + v_{o1}i_1 = 0 (A.1.19)$$

Then, a solution for i_{L1} is given by

$$i_{L1} = \frac{v_{in} - \sqrt{\left(v_{in}^2 - 4R_L v_{o1} i_1\right)}}{2R_{L1}} \tag{A.1.20}$$

Similarly, i_{L2} can be found as

$$i_{L2} = \frac{v_{in} - \sqrt{\left(v_{in}^2 - 4R_L v_{o2} i_2\right)}}{2R_{L2}} \tag{A.1.21}$$

Since $i_{in} = i_{L1} + i_{L2}$, the total input current can be obtained as

$$i_{in} = \frac{v_{in} - \sqrt{\left(v_{in}^2 - 4R_L v_{o1} i_1\right)}}{2R_L} + \frac{v_{in} - \sqrt{\left(v_{in}^2 - 4R_L v_{o2} i_2\right)}}{2R_L}$$
(A.1.22)

where $R_L = R_{L1} = R_{L2}$.



Fig. A.1.2: Simplified block diagram of the HESS power allocation.

A.1.6 Derivation of equation (7.4)

The output capacitor current waveforms i_{C1} and i_{C2} can be written as

$$i_{C1} = C_1 \omega \frac{V_o}{2} \cos(\omega t + \delta) \tag{A.1.23}$$

$$i_{C2} = -C_2 \omega \frac{V_o}{2} \cos(\omega t + \delta) \tag{A.1.24}$$

Then,

$$i_1 = I_o \sin(\omega t + \alpha) + C_1 \omega \frac{V_o}{2} \cos(\omega t + \delta)$$
(A.1.25)

Similarly, i_2 can be written as,

$$i_2 = -I_o \sin(\omega t + \alpha) - C_2 \omega \frac{V_o}{2} \cos(\omega t + \delta)$$
(A.1.26)

Then, the total input current to the boost inverter can be obtained as,

$$i_{in} = i_1 \frac{v_{o1}}{v_{in}} + i_2 \frac{v_{o2}}{v_{in}} \tag{A.1.27}$$

Substituting (A.1.25), (A.1.26), (7.1), and (7.2) into (A.1.27),

$$i_{in} = \frac{1}{v_{in}} \left[\frac{V_o I_o}{2} \cos(\alpha - \delta) + \frac{V_o}{2} \sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C \omega \sin(\alpha - \delta) \right]} \cos(2\omega t - \theta) \right]$$
(A.1.28)

where

$$\cos\theta = \frac{-I_o\cos(\alpha+\delta) + \frac{C\omega V_o\sin(2\delta)}{2}}{\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha-\delta)\right]}}$$

and $C = C_1 = C_2$.

A.1.7 Derivation of equation (7.10) and (7.11)

The total input current of the boost inverter can be obtained by

$$i_{in} = i_1 \frac{v_{o1}}{v_{in}} + i_2 \frac{v_{o2}}{v_{in}} \tag{A.1.29}$$

Substituting (7.6), (7.7), (7.8), and (7.9) in to (A.1.29) and simplifying, the input current can be written as

$$i_{in} = \frac{1}{v_{in}} \left[\frac{V_o I_o}{2} \cos(\alpha - \delta) + \frac{V_o}{2} \sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C \omega \sin(\alpha - \delta) \right] \cos(2\omega t - \theta)} \right]} \frac{1}{v_{in}} \left[-2CK V_{DC} \frac{di_{in,AC}}{dt} + 2CK^2 i_{inAC} \frac{di_{in,AC}}{dt} \right]$$
(A.1.30)

where

$$\cos\theta = \frac{-I_o \cos(\alpha + \delta) + \frac{C\omega V_o \sin(2\delta)}{2}}{\sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C\omega \sin(\alpha - \delta)\right]}}$$

It can be observed that the input current waveform contains a DC component and an AC component. Hence $i_{in} = i_{in,AC} + i_{in,DC}$. Considering the AC component of (A.1.30) and simplifying

$$\frac{di_{in,AC}}{dt} = \frac{v_{in}}{(2V_{DC}CK - 2CK^2 i_{in,AC})}(i_{in,AC0} - i_{in,AC})$$
(A.1.31)

where

$$i_{in,AC0} = \frac{V_o}{2v_{in}} \sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C \omega \sin(\alpha - \delta)\right]} \cos(2\omega t - \theta)$$
(A.1.32)

A.1.8 Derivation of equation (7.24)

Form (7.21),

$$V_{DC} \ge v_{in} + \frac{V_o}{2} + \frac{v_{in} \max(|i_{in,AC0}|)}{4V_{DC}C\omega}$$
(A.1.33)

where

$$\max(|i_{in,AC0}|) = \frac{V_o}{2v_{in}} \sqrt{\left[I_{o,max}^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_{o,max} C\omega\right]}.$$
 (A.1.34)

Then,

$$4V_{DC}^2 C\omega \ge 4V_{DC}C\omega \left(v_{in} + \frac{V_o}{2}\right) + v_{in}\max(|i_{in,AC0}|) \tag{A.1.35}$$

$$4V_{DC}^2 C\omega - 2C\omega V_{DC} \left(V_o + 2v_{in} \right) - v_{in} \max(|i_{in,AC0}|) \ge 0$$
(A.1.36)

Then the minimum positive value which satisfies (A.1.36) is given by

$$V_{DC,\min} = \frac{2C\omega(V_o + 2v_{in}) + \sqrt{\left[2C\omega(V_o + 2v_{in})\right]^2 + 16C\omega v_{in}\max(|i_{in,AC0}|)}}{8C\omega}.$$
 (A.1.37)

A.1.9 Derivation of equation (7.29)

The output capacitor current waveforms i_{C1} and i_{C2} can be written as

$$i_{C1} = C_1 \omega \frac{V_o}{2} \cos(\omega t + \delta) \tag{A.1.38}$$

$$i_{C2} = -C_2 \omega \frac{V_o}{2} \cos(\omega t + \delta) \tag{A.1.39}$$

Then,

$$i_1 = I_o \sin(\omega t + \alpha) + C_1 \omega \frac{V_o}{2} \cos(\omega t + \delta)$$
(A.1.40)

Similarly, i_2 can be written as,

$$i_2 = -I_o \sin(\omega t + \alpha) - C_2 \omega \frac{V_o}{2} \cos(\omega t + \delta)$$
(A.1.41)

From (7.28)

$$P_{in,HESS} = P_{o1} + P_{o2} + P_{C1} + P_{C2}.$$
(A.1.42)

Hence

$$P_{in,HESS} = v_{o1}i_{o1} + v_{o2}i_{o2} + i_{C1}v_{o1} + i_{C2}v_{o2}.$$
(A.1.43)

$$P_{in,HESS} = v_{o1} \left(i_{o1} + i_{C1} \right) + v_{o2} \left(i_{o2} + i_{C2} \right) \tag{A.1.44}$$

Substituting (A.1.40), (A.1.41), (7.1), and (7.2) into (A.1.44),

$$P_{in,HESS} = \left[\frac{V_o I_o}{2}\cos(\alpha - \delta) + \frac{V_o}{2}\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha - \delta)\right]}\cos(2\omega t - \theta)\right]$$
(A.1.45)

where

$$\cos\theta = \frac{-I_o\cos(\alpha+\delta) + \frac{C\omega V_o\sin(2\delta)}{2}}{\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha-\delta)\right]}}$$

and $C = C_1 = C_2$.

A.1.10 Derivation of equation (7.33), (7.34), and (7.35)

The total input power of the HESS can be obtained by

$$P_{in,HESS} = v_{o1} \left(i_{o1} + i_{C1} \right) + v_{o2} \left(i_{o2} + i_{C2} \right)$$
(A.1.46)

With the ripple reduction method,

$$i_1 = I_o \sin(\omega t + \alpha) + C\omega \frac{V_o}{2} \cos(\omega t + \delta) - CK \frac{di_{sc,AC}}{dt}$$
(A.1.47)

and

$$i_2 = -I_o \sin(\omega t + \alpha) - C\omega \frac{V_o}{2} \cos(\omega t + \delta) - CK \frac{di_{sc,AC}}{dt}$$
(A.1.48)

Substituting (A.1.47), (A.1.48), (7.31), and (7.32) in to (A.1.46) and simplifying, the input current can be written as

$$P_{in,HESS} = \left[\frac{V_o I_o}{2}\cos(\alpha - \delta) + \frac{V_o}{2}\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha - \delta)\right]}\cos(2\omega t - \theta)\right]} \left[-2CKV_{DC}\frac{di_{sc,AC}}{dt} + 2CK^2 i_{sc_AC}\frac{di_{sc,AC}}{dt}\right] \quad (A.1.49)$$

where

$$\cos\theta = \frac{-I_o\cos(\alpha+\delta) + \frac{C\omega V_o\sin(2\delta)}{2}}{\sqrt{\left[I_o^2 + \frac{C^2\omega^2 V_o^2}{4} + V_o I_o C\omega\sin(\alpha-\delta)\right]}}$$

It can be observed that the input power contains a DC component and an AC component. Hence, $P_{in,HESS} = P_{in,HESS,AC} + P_{in,HESS,DC}$. Since the AC component of the input power is allocated to the supercapacitor, the AC component of the HESS input power corresponds to the AC component of the power supplied by the supercapacitor.

Considering the AC component of (A.1.49) and simplifying

$$\frac{di_{sc,AC}}{dt} = \frac{v_{in}}{(2V_{DC}CK - 2CK^2i_{in,AC})}(i_{in,AC0} - i_{in,AC})$$
(A.1.50)

where

$$i_{sc,AC0} = \frac{V_o}{2v_{in}} \sqrt{\left[I_o^2 + \frac{C^2 \omega^2 V_o^2}{4} + V_o I_o C \omega \sin(\alpha - \delta)\right]} \cos(2\omega t - \theta)$$
(A.1.51)

A.2 Appendix 2

A.2.1 Simulink Models used for DSpace controller implementation



Fig. A.2.1: Simulink blocks: Duty ratio signal output to DSpace system



Fig. A.2.2: Simulink blocks: Voltage and current inputs through DSpace analog to digital converter



Fig. A.2.3: Simulink blocks: Active and reactive power controller and double loop control system
Bibliography

- Q. Hao, Z. Jianhui, L. Jih-Sheng, and Y. Wensong, "A high-efficiency grid-tie battery energy storage system," *IEEE Transactions on Power Electronics*, vol. 26, no. 3, pp. 886–896, 2011.
- [2] C. A. Hill, M. C. Such, C. Dongmei, J. Gonzalez, and W. M. Grady, "Battery energy storage for enabling integration of distributed solar power generation," *IEEE Transactions on Smart Grid*, vol. 3, no. 2, pp. 850–857, 2012.
- [3] L. Maharjan, T. Yamagishi, and H. Akagi, "Active-power control of individual converter cells for a battery energy storage system based on a multilevel cascade pwm converter," *IEEE Transactions on Power Electronics*, vol. 27, no. 3, pp. 1099–1107, 2012.
- [4] V. Boicea, "Energy storage technologies: The past and the present," Proceedings of the IEEE, vol. 102, pp. 1777–1794, Nov 2014.
- [5] S. Teleke, M. E. Baran, S. Bhattacharya, and A. Q. Huang, "Rule-based control of battery energy storage for dispatching intermittent renewable sources," *IEEE Transactions on Sustainable Energy*, vol. 1, no. 3, pp. 117–124, 2010.
- [6] R. Carter, A. Cruden, and P. J. Hall, "Optimizing for efficiency or battery life in a battery/supercapacitor electric vehicle," *IEEE Transactions on Vehicular Technology*, vol. 61, no. 4, pp. 1526–1533, 2012.
- [7] N. Omar, J. V. Mierlo, B. Verbrugge, and P. V. den Bossche, "Power and life enhancement of battery-electrical double layer capacitor for hybrid electric and charge-depleting plug-in vehicle applications," *Electrochimica Acta*, vol. 55, no. 25, pp. 7524 – 7531, 2010.
- [8] M. Dubarry, V. Svoboda, R. Hwu, and B. Y. Liaw, "Capacity and power fading mechanism identification from a commercial cell evaluation," *Journal of Power Sources*, vol. 165, no. 2, pp. 566 – 572, 2007.

- [9] E. M. Krieger, J. Cannarella, and C. B. Arnold, "A comparison of lead-acid and lithium-based battery behavior and capacity fade in off-grid renewable charging applications," *Energy*, vol. 60, no. 0, pp. 492–500, 2013.
- [10] K. W. Wee, S. S. Choi, and D. M. Vilathgamuwa, "Design of a least-cost battery-supercapacitor energy storage system for realizing dispatchable wind power," *IEEE Transactions on Sustainable Energy*, vol. 4, no. 3, pp. 786–796, 2013.
- [11] A. J. Ruddell, A. G. Dutton, H. Wenzl, C. Ropeter, D. U. Sauer, J. Merten, C. Orfanogiannis, J. W. Twidell, and P. Vezin, "Analysis of battery current microcycles in autonomous renewable energy systems," *Journal of Power Sources*, vol. 112, no. 2, pp. 531–546, 2002.
- [12] A. M. Gee, F. V. P. Robinson, and R. W. Dunn, "Analysis of battery lifetime extension in a small-scale wind-energy system using supercapacitors," *IEEE Transactions on Energy Conversion*, vol. 28, no. 1, pp. 24–33, 2013.
- [13] S. Bala, T. Tengner, P. Rosenfeld, and F. Delince, "The effect of low frequency current ripple on the performance of a lithium iron phosphate (lfp) battery energy storage system," in 2012 IEEE Energy Conversion Congress and Exposition (ECCE), pp. 3485–3492.
- [14] T. Shimizu, Y. Jin, and G. Kimura, "Dc ripple current reduction on a single-phase pwm voltage-source rectifier," *IEEE Transactions on Industry Applications*, vol. 36, pp. 1419–1429, Sep 2000.
- [15] A. Kuperman, U. Levy, J. Goren, A. Zafransky, and A. Savernin, "Battery charger for electric vehicle traction battery switch station," *IEEE Transactions on Industrial Electronics*, vol. 60, no. 12, pp. 5391–5399, 2013.
- [16] Z. Haihua, T. Bhattacharya, T. Duong, T. S. T. Siew, and A. M. Khambadkone, "Composite energy storage system involving battery and ultracapacitor with dynamic energy management in microgrid applications," *IEEE Transactions on Power Electronics*, vol. 26, no. 3, pp. 923–930, 2011.
- [17] G. Lijun, R. A. Dougal, and S. Liu, "Power enhancement of an actively controlled battery/ultracapacitor hybrid," *IEEE Transactions on Power Electronics*, vol. 20, no. 1, pp. 236–243, 2005.
- [18] A. Kuperman and I. Aharon, "Battery-ultracapacitor hybrids for pulsed current loads: A review," *Renewable and Sustainable Energy Reviews*, vol. 15, no. 2, pp. 981–992, 2011.

- [19] J. Cao and A. Emadi, "A new battery/ultracapacitor hybrid energy storage system for electric, hybrid, and plug-in hybrid electric vehicles," *IEEE Transactions on Power Electronics*, vol. 27, no. 1, pp. 122–132, 2012.
- [20] J. Bauman and M. Kazerani, "A comparative study of fuel-cell-battery, fuel-cell-ultracapacitor, and fuel-cell-battery-ultracapacitor vehicles," *IEEE Transactions on Vehicular Technology*, vol. 57, no. 2, pp. 760–769, 2008.
- [21] A. Kuperman, I. Aharon, S. Malki, and A. Kara, "Design of a semiactive battery-ultracapacitor hybrid energy source," *IEEE Transactions on Power Electronics*, vol. 28, no. 2, pp. 806–815, 2013.
- [22] I. Aharon and A. Kuperman, "Topological overview of powertrains for battery-powered vehicles with range extenders," *IEEE Transactions on Power Electronics*, vol. 26, pp. 868–876, March 2011.
- [23] J. Moreno, M. E. Ortuzar, and J. W. Dixon, "Energy-management system for a hybrid electric vehicle, using ultracapacitors and neural networks," *IEEE Transactions on Industrial Electronics*, vol. 53, no. 2, pp. 614–623, 2006.
- [24] S. Junyi, S. Dusmez, and A. Khaligh, "Optimization of sizing and battery cycle life in battery/ultracapacitor hybrid energy storage systems for electric vehicle applications," *IEEE Transactions on Industrial Informatics*, vol. 10, no. 4, pp. 2112–2121, 2014.
- [25] M. Y. Ayad, M. Becherif, A. Djerdir, and A. Miraoui, "Sliding mode control for energy management of dc hybrid power sources using fuel cell, batteries and supercapacitors," in *International Conference on Clean Electrical Power*, 2007. *ICCEP* '07., pp. 500–505.
- [26] V. Yuhimenko, C. Lerman, and A. Kuperman, "Dc active power filter based hybrid energy source for pulsed power loads," *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. PP, no. 99, pp. 1–1, 2015.
- [27] B. Hredzak, V. G. Agelidis, and G. D. Demetriades, "A low complexity control system for a hybrid dc power source based on ultracapacitor-lead acid battery configuration," *IEEE Transactions on Power Electronics*, vol. 29, no. 6, pp. 2882–2891, 2014.
- [28] B. Hredzak, V. G. Agelidis, and J. Minsoo, "A model predictive control system for a hybrid battery-ultracapacitor power source," *IEEE Transactions on Power Electronics*, vol. 29, no. 3, pp. 1469–1479, 2014.

- [29] A. Lahyani, P. Venet, A. Guermazi, and A. Troudi, "Battery/supercapacitors combination in uninterruptible power supply (ups)," *IEEE Transactions on Power Electronics*, vol. 28, no. 4, pp. 1509–1522, 2013.
- [30] O. Laldin, M. Moshirvaziri, and O. Trescases, "Predictive algorithm for optimizing power flow in hybrid ultracapacitor/battery storage systems for light electric vehicles," *IEEE Transactions on Power Electronics*, vol. 28, no. 8, pp. 3882–3895, 2013.
- [31] F. Ongaro, S. Saggini, and P. Mattavelli, "Li-ion battery-supercapacitor hybrid storage system for a long lifetime, photovoltaic-based wireless sensor network," *IEEE Transactions on Power Electronics*, vol. 27, no. 9, pp. 3944–3952, 2012.
- [32] X. Qing, W. Yanzhi, K. Younghyun, M. Pedram, and C. Naehyuck, "Charge allocation in hybrid electrical energy storage systems," *IEEE Transactions on Computer-Aided Design of Integrated Circuits and Systems*, vol. 32, no. 7, pp. 1003–1016, 2013.
- [33] L. Xiong, W. Peng, L. Poh Chiang, G. Feng, and C. Fook Hoong, "Control of hybrid battery/ultra-capacitor energy storage for stand-alone photovoltaic system," in *IEEE Energy Conversion Congress and Exposition (ECCE)*, pp. 336–341, 2010.
- [34] R. Esteves Araujo, R. de Castro, C. Pinto, P. Melo, and D. Freitas, "Combined sizing and energy management in evs with batteries and supercapacitors," *IEEE Transactions on Vehicular Technology*, vol. 63, no. 7, pp. 3062–3076, 2014.
- [35] L. Baoquan, Z. Fang, and B. Xianwen, "Fuzzy control for hybrid energy storage system based on battery and ultra-capacitor in micro-grid," in 2012 7th International Power Electronics and Motion Control Conference (IPEMC), vol. 2, pp. 778–782.
- [36] C. Abbey, K. Strunz, and G. Joos, "A knowledge-based approach for control of two-level energy storage for wind energy systems," *IEEE Transactions on Energy Conversion*, vol. 24, no. 2, pp. 539–547, 2009.
- [37] M. A. Tankari, M. B. Camara, B. Dakyo, and G. Lefebvre, "Use of ultracapacitors and batteries for efficient energy management in wind diesel hybrid system," *IEEE Transactions on Sustainable Energy*, vol. 4, no. 2, pp. 414–424, 2013.
- [38] E. Schaltz, A. Khaligh, and P. O. Rasmussen, "Influence of battery/ultracapacitor energy-storage sizing on battery lifetime in a fuel cell hybrid electric vehicle," *IEEE Transactions on Vehicular Technology*, vol. 58, no. 8, pp. 3882–3891, 2009.

- [39] A. Etxeberria, I. Vechiu, H. Camblong, and J. M. Vinassa, "Comparison of sliding mode and pi control of a hybrid energy storage system in a microgrid application," *Energy Procedia*, vol. 12, pp. 966–974, 2011.
- [40] A. Hajizadeh, M. A. Golkar, and A. Feliachi, "Voltage control and active power management of hybrid fuel-cell/energy-storage power conversion system under unbalanced voltage sag conditions," *IEEE Transactions on Energy Conversion*, vol. 25, no. 4, pp. 1195–1208, 2010.
- [41] F. Liu, J. Liu, B. Zhang, H. Zhang, and U. Hasan Saad, "Energy management of hybrid energy storage system (hess) based on sliding mode control," in 2012 7th International Power Electronics and Motion Control Conference (IPEMC), vol. 1, pp. 406–410.
- [42] C. Mid-Eum, K. Seong-Woo, and S. Seung-Woo, "Energy management optimization in a battery/supercapacitor hybrid energy storage system," *IEEE Transactions on Smart Grid*, vol. 3, no. 1, pp. 463–472, 2012.
- [43] N. R. Tummuru, M. K. Mishra, and S. Srinivas, "Dynamic energy management of renewable grid integrated hybrid energy storage system," *IEEE Transactions on Industrial Electronics*, vol. PP, no. 99, pp. 1–1, 2015.
- [44] S. S. G. Jayasinghe, D. M. Vilathgamuwa, and U. K. Madawala, "Diode-clamped three-level inverter-based battery /supercapacitor direct integration scheme for renewable energy systems," *IEEE Transactions on Power Electronics*, vol. 26, no. 12, pp. 3720–3729, 2011.
- [45] I. Vechiu, A. Etxeberria, H. Camblong, and J. M. Vinassa, "Three-level neutral point clamped inverter interface for flow battery/supercapacitor energy storage system used for microgrids," in 2nd IEEE PES International Conference and Exhibition on Innovative Smart Grid Technologies (ISGT Europe), pp. 1–6, 2011.
- [46] N. Mendis, K. M. Muttaqi, and S. Perera, "Management of low- and high-frequency power components in demand-generation fluctuations of a dfig-based wind-dominated raps system using hybrid energy storage," *IEEE Transactions on Industry Applications*, vol. 50, no. 3, pp. 2258–2268, 2014.
- [47] A. M. Van Voorden, L. M. R. Elizondo, G. C. Paap, J. Verboomen, and L. Van der Sluis, "The application of super capacitors to relieve battery-storage systems in autonomous renewable energy systems," in *IEEE Lausanne Power Tech*, pp. 479–484, 2007.

- [48] Y. Yoshida, W. Guohong, and T. Minakawa, "Mitigating fluctuation of wind turbine power generation system by introduction of electrical energy storage with combined use of edlc and lead acid storage battery," in *International Conference onPower* System Technolog (POWERCON), pp. 1–7, 2010.
- [49] Y. Zhang, X. Tang, Z. Qi, and Z. Liu, "The ragone plots guided sizing of hybrid storage system for taming the wind power," *International Journal of Electrical Power and Energy Systems*, vol. 65, no. 0, pp. 246–253, 2015.
- [50] J. Quanyuan and H. Haisheng, "Wavelet-based capacity configuration and coordinated control of hybrid energy storage system for smoothing out wind power fluctuations," *IEEE Transactions on Power Systems*, vol. 28, no. 2, pp. 1363–1372, 2013.
- [51] R. A. Dougal, S. Liu, and R. E. White, "Power and life extension of battery-ultracapacitor hybrids," *IEEE Transactions on Components and Packaging Technologies*, vol. 25, no. 1, pp. 120–131, 2002.
- [52] A. Kuperman, I. Aharon, A. Kara, and S. Malki, "A frequency domain approach to analyzing passive batteryultracapacitor hybrids supplying periodic pulsed current loads," *Energy Conversion and Management*, vol. 52, no. 12, pp. 3433 – 3438, 2011.
- [53] S. Fiorenti, J. Guanetti, Y. Guezennec, and S. Onori, "Modeling and experimental validation of a hybridized energy storage system for automotive applications," *Journal of Power Sources*, vol. 241, pp. 112 – 120, 2013.
- [54] R. O. Caceres and I. Barbi, "A boost dc-ac converter: analysis, design, and experimentation," *IEEE Transactions on Power Electronics*, vol. 14, no. 1, pp. 134–141, 1999.
- [55] P. Sanchis, A. Ursaea, E. Gubia, and L. Marroyo, "Boost dc-ac inverter: a new control strategy," *IEEE Transactions on Power Electronics*, vol. 20, no. 2, pp. 343–353, 2005.
- [56] X. Liu, P. C. Loh, P. Wang, and F. Blaabjerg, "A direct power conversion topology for grid integration of hybrid ac/dc energy resources," *IEEE Transactions on Industrial Electronics*, vol. 60, pp. 5696–5707, Dec 2013.
- [57] J. Minsoo and V. G. Agelidis, "A minimum power-processing-stage fuel-cell energy system based on a boost-inverter with a bidirectional backup battery storage," *IEEE Transactions on Power Electronics*, vol. 26, no. 5, pp. 1568–1577, 2011.

- [58] W. Li, G. Joos, and J. Belanger, "Real-time simulation of a wind turbine generator coupled with a battery supercapacitor energy storage system," *IEEE Transactions* on *Industrial Electronics*, vol. 57, pp. 1137–1145, April 2010.
- [59] D. Shin, Y. Kim, Y. Wang, N. Chang, and M. Pedram, "Constant-current regulator-based battery-supercapacitor hybrid architecture for high-rate pulsed load applications," *Journal of Power Sources*, vol. 205, pp. 516 – 524, 2012.
- [60] M. Zandi, A. Payman, J. P. Martin, S. Pierfederici, B. Davat, and F. Meibody-Tabar, "Energy management of a fuel cell/supercapacitor/battery power source for electric vehicular applications," *IEEE Transactions on Vehicular Technology*, vol. 60, no. 2, pp. 433–443, 2011.
- [61] "Product guide maxwell technologies boostcap ultracapacitors," report, Maxwell Technologies, 2009 (Date last accessed: Nov 2016).
- [62] D. B. W. Abeywardana, B. Hredzak, and V. G. Agelidis, "An integration scheme for a direct ac line battery-supercapacitor hybrid energy storage system," in *Australasian Universities Power Engineering Conference (AUPEC)*, pp. 1–6, 2013.
- [63] D. B. W. Abeywardana, B. Hredzak, and V. G. Agelidis, "A single phase grid integration scheme for battery-supercapacitor ac line hybrid storage system," Feb. 2014.
- [64] D. B. W. Abeywardana, B. Hredzak, and V. G. Agelidis, "Single-phase grid-connected LiFePO₄ battery-supercapacitor hybrid energy storage system with interleaved boost inverter," *IEEE Transactions on Power Electronics*, vol. PP, no. 99, pp. 1–1, 2014.
- [65] D. B. W. Abeywardana, B. Hredzak, V. G. Agelidis, and G. Demetriades, "Supercapacitor sizing method for energy controlled filter based hybrid energy storage systems," *IEEE Transactions on Power Electronics*, vol. PP, no. 99, pp. 1–1, 2016 (Early access article).
- [66] D. B. W. Abeywardana, B. Hredzak, and V. G. Agelidis, "A fixed frequency sliding mode controller for a boost inverter based battery-supercapacitor hybrid energy storage system," *IEEE Transactions on Power Electronics*, vol. PP, no. 99, pp. 1–1, 2016 (Early access article).
- [67] D. B. W. Abeywardana, B. Hredzak, and V. G. Agelidis, "A rule-based controller to mitigate dc-side 2nd-order harmonic current in a single-phase boost inverter," *IEEE Transactions on Power Electronics*, vol. PP, no. 99, pp. 1–1, 2014.

- [68] D. B. W. Abeywardana, B. Hredzak, and V. G. Agelidis, "An input current feedback method to mitigate the dc-side low frequency ripple current in a single-phase boost inverter," *IEEE Transactions on Power Electronics*, vol. PP, no. 99, pp. 1–1, 2015.
- [69] D. B. W. Abeywardana, B. Hredzak, and V. G. Agelidis, "Battery-supercapacitor hybrid energy storage system with reduced low frequency input current ripple," in 2015 International Conference on Renewable Energy Research and Applications (ICRERA), pp. 328–332, Nov 2015.
- [70] R. Teodorescu, M. Liserre, and P. Rodrguez, Grid Converters for Photovoltaic and Wind Power Systems. John Wiley and Sons, Ltd, 2011.
- [71] D. Cortes, N. Vazquez, and J. Alvarez-Gallegos, "Dynamical sliding-mode control of the boost inverter," *IEEE Transactions on Industrial Electronics*, vol. 56, no. 9, pp. 3467–3476, 2009.
- [72] I. E. Colling and I. Barbi, "Reversible unity power factor step-up/step-down ac-dc converter controlled by sliding mode," *IEEE Transactions on Power Electronics*, vol. 16, no. 2, pp. 223–230, 2001.
- [73] Z. Wei, D. D. C. Lu, and V. G. Agelidis, "Current control of grid-connected boost inverter with zero steady-state error," *IEEE Transactions on Power Electronics*, vol. 26, no. 10, pp. 2825–2834, 2011.
- [74] Z. Guo-Rong, T. Siew-Chong, C. Yu, and C. K. Tse, "Mitigation of low-frequency current ripple in fuel-cell inverter systems through waveform control," *IEEE Transactions on Power Electronics*, vol. 28, no. 2, pp. 779–792, 2013.
- [75] Z. Guo-Rong, X. Cheng-Yuan, W. Hao-Ran, C. Wei, and T. Siew-Chong, "Dynamic characteristics of boost inverter with waveform control," in *Twenty-Ninth* Annual IEEE Applied Power Electronics Conference and Exposition (APEC),2014, pp. 1771–1775.
- [76] H. R. Wang, G. R. Zhu, X. B. Fu, W. X. Liu, C. Y. Xiao, B. Liang, and S. C. Tan, "Mitigating low frequency ripple in multiply micro sources inverter system with waveform control method," in *IEEE 23rd International Symposium on Industrial Electronics (ISIE)*, pp. 2390–2394, 2014.
- [77] G. R. Zhu, H. R. Wang, C. Y. Xiao, T. Siew-Chong, and Y. Kang, "Assessment of waveform control method for mitigation of low-frequency current ripple," in *Twenty-Eighth Annual IEEE Applied Power Electronics Conference and Exposition* (APEC), pp. 3101–3106, 2013.

- [78] J. Minsoo, M. Ciobotaru, and V. G. Agelidis, "A single-phase grid-connected fuel cell system based on a boost-inverter," *IEEE Transactions on Power Electronics*, vol. 28, no. 1, pp. 279–288, 2013.
- [79] S. Danyali, S. H. Hosseini, and G. B. Gharehpetian, "New extendable single-stage multi-input dc-dc/ac boost converter," *IEEE Transactions on Power Electronics*, vol. 29, no. 2, pp. 775–788, 2014.
- [80] C. Chin and M. A. Knights, "Interleaving technique in distributed power conversion systems," *IEEE Transactions on Circuits and Systems I: Fundamental Theory and Applications*, vol. 42, no. 5, pp. 245–251, 1995.
- [81] B. A. Miwa, D. Otten, and M. Schlecht, "High efficiency power factor correction using interleaving techniques," in *Conference Proceedings Seventh Annual Applied Power Electronics Conference and Exposition*, 1992. APEC '92., pp. 557–568, Feb 1992.
- [82] M. Ciobotaru, R. Teodorescu, and F. Blaabjerg, "A new single-phase pll structure based on second order generalized integrator," in 37th IEEE Power Electronics Specialists Conference, 2006. PESC '06, pp. 1–6.
- [83] C. Zheng, F. Yuhong, and C. C. Mi, "State of charge estimation of lithium-ion batteries in electric drive vehicles using extended kalman filtering," *IEEE Transactions on Vehicular Technology*, vol. 62, no. 3, pp. 1020–1030, 2013.
- [84] T. Xidong, M. Xiaofeng, L. Jian, and B. Koch, "Li-ion battery parameter estimation for state of charge," in *American Control Conference (ACC)*, pp. 941–946, 2011.
- [85] G. L. Plett, "Extended kalman filtering for battery management systems of lipb-based hev battery packs: Part 1. background," *Journal of Power Sources*, vol. 134, no. 2, pp. 252–261, 2004.
- [86] NREL, "National renewable energy laboratory-measurement and instrumentation data center."
- [87] F. Ciccarelli, A. Del Pizzo, and D. Iannuzzi, "Improvement of energy efficiency in light railway vehicles based on power management control of wayside lithium-ion capacitor storage," *IEEE Transactions on Power Electronics*, vol. 29, no. 1, pp. 275–286, 2014.

- [88] J. Pegueroles-Queralt, F. D. Bianchi, and O. Gomis-Bellmunt, "A power smoothing system based on supercapacitors for renewable distributed generation," *IEEE Transactions on Industrial Electronics*, vol. 62, no. 1, pp. 343–350, 2015.
- [89] A. Kuperman, M. Mellincovsky, C. Lerman, I. Aharon, N. Reichbach, G. Geula, and R. Nakash, "Supercapacitor sizing based on desired power and energy performance," *IEEE Transactions on Power Electronics*, vol. 29, no. 10, pp. 5399–5405, 2014.
- [90] J. Marcos, O. Storkl, L. Marroyo, M. Garcia, and E. Lorenzo, "Storage requirements for pv power ramp-rate control," *Solar Energy*, vol. 99, no. 0, pp. 28–35, 2014.
- [91] T. Siew-Chong, Y. M. Lai, and C. K. Tse, "General design issues of sliding-mode controllers in dc-dc converters," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 3, pp. 1160–1174, 2008.
- [92] S. Singh, D. Fulwani, and V. Kumar, "Robust sliding-mode control of dc/dc boost converter feeding a constant power load," *IET Power Electronics*, vol. 8, no. 7, pp. 1230–1237, 2015.
- [93] E. Vidal-Idiarte, A. Marcos-Pastor, G. Garcia, A. Cid-Pastor, and L. Martinez-Salamero, "Discrete-time sliding-mode-based digital pulse width modulation control of a boost converter," *IET Power Electronics*, vol. 8, no. 5, pp. 708–714, 2015.
- [94] T. Siew-Chong, Y. M. Lai, C. K. Tse, Marti, x, L. nez Salamero, and W. Chi-Kin, "A fast-response sliding-mode controller for boost-type converters with a wide range of operating conditions," *IEEE Transactions on Industrial Electronics*, vol. 54, no. 6, pp. 3276–3286, 2007.
- [95] "IEEE application guide for IEEE std 1547(tm), IEEE standard for interconnecting distributed resources with electric power systems," *IEEE Std 1547.2-2008*, pp. 1–217, April 2009.
- [96] Y. He and F. Luo, "Sliding-mode control for dc-dc converters with constant switching frequency," *IEE Proceedings Control Theory and Applications*, vol. 153, pp. 37–45, Jan 2006.
- [97] M. Lopez, L. de Vicuna, M. Castilla, P. Gaya, and O. Lopez, "Current distribution control design for paralleled dc/dc converters using sliding-mode control," *IEEE Transactions on Industrial Electronics*, vol. 51, pp. 419–428, April 2004.

- [98] M. Agostinelli, R. Priewasser, S. Marsili, and M. Huemer, "Constant switching frequency techniques for sliding mode control in dc-dc converters," in 16th Int'l Symposium on Theoretical Electrical Engineering (ISTET), 2011 Joint 3rd Int'l Workshop on Nonlinear Dynamics and Synchronization (INDS), pp. 1–5, July 2011.
- [99] K. Ramash Kumar and S. Jeevananthan, "Modelling and implementation of fixed switching frequency sliding mode controller for negative output elementary super lift luo-converter," *IET Power Electronics*, vol. 5, pp. 1593–1604, September 2012.
- [100] Z. Guo-Rong, T. Siew-Chong, W. Ke-Wei, and C. K. Tse, "Waveform control of fuel-cell inverter systems," in *IEEE Energy Conversion Congress and Exposition* (ECCE), pp. 4457–4463, 2012.
- [101] W. Yao, X. Wang, P. C. Loh, X. Zhang, and F. Blaabjerg, "Improved power decoupling scheme for a single-phase grid-connected differential inverter with realistic mismatch in storage capacitances," *IEEE Transactions on Power Electronics*, vol. 32, pp. 186–199, Jan 2017.
- [102] L. Hongbo, Z. Kai, Z. Hui, F. Shengfang, and X. Jian, "Active power decoupling for high-power single-phase pwm rectifiers," *IEEE Transactions on Power Electronics*, vol. 28, no. 3, pp. 1308–1319, 2013.
- [103] L. Fangrui, D. Shanxu, L. Fei, L. Bangyin, and K. Yong, "A variable step size inc mppt method for pv systems," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 7, pp. 2622–2628, 2008.
- [104] N. Femia, G. Petrone, G. Spagnuolo, and M. Vitelli, "Optimization of perturb and observe maximum power point tracking method," *IEEE Transactions on Power Electronics*, vol. 20, no. 4, pp. 963–973, 2005.
- [105] L. Kyoung-Jun, L. Jong-Pil, S. Dongsul, Y. Dong-Wook, and K. Hee-Je, "A novel grid synchronization pll method based on adaptive low-pass notch filter for grid-connected pcs," *IEEE Transactions on Industrial Electronics*, vol. 61, no. 1, pp. 292–301, 2014.
- [106] R. Teodorescu, F. Blaabjerg, M. Liserre, and P. C. Loh, "Proportional-resonant controllers and filters for grid-connected voltage-source converters," *IEE Proceedings Electric Power Applications*, vol. 153, no. 5, pp. 750–762, 2006.